Morningstar® Document Research®

# FORM 6-K

# AerCap Holdings N.V. - AER

## Filed: September 01, 2009 (period: September 01, 2009)

Report of foreign issuer rules 13a-16 and 15d-16 of the Securities Exchange Act

# **Table of Contents**

<u>6-K - 6-K</u>

SIGNATURE EXHIBIT INDEX EX-99.1 (EX-99.1)

EX-99.2 (EX-99.2)

# FORM 6 - K SECURITIES AND EXCHANGE COMMISSION

Washington, DC 20549

Report of Foreign Private Issuer Pursuant to Rule 13a - 16 or 15d - 16 under the Securities Exchange Act of 1934

For the Quarterly Period ended June 30, 2009

**Commission File Number 001-33159** 

## **AERCAP HOLDINGS N.V.**

(Translation of Registrant's Name into English)

Stationsplein 965, 1117 CE Schiphol Airport, The Netherlands, +31-20-655-9655 (Address of Principal Executive Office)

Indicate by check mark whether the registrant files or will file annual reports under cover of Form 20-F or Form 40-F.

Form 40-F

Form 20-F

Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(1):

Note: Regulation S-T Rule 101(b)(1) only permits the submission in paper of a Form 6-K if submitted solely to provide an attached annual report to security holders.

Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(7)):  $\Box$ 

Indicate by check mark whether the registrant by furnishing the information contained in this form is also thereby furnishing the information to the Commission pursuant to Rule 12g3-2(b) under the Securities Act of 1934.

Yes 🗆 No 🗵

If "Yes" is marked, indicated below the file number assigned to the registrant in connection with Rule 12g3-2(b): 82 -

N/A

## **Other Events**

On September 1, 2009, AerCap Holdings N.V. filed its interim financial report for the quarter ended June 30, 2009.

This form 6-K is incorporated by reference into the Company's form F-3 registration statement file no. 333-155194 and related prospectus, as such registration statement and prospectus may be amended from time to time, and is incorporated by reference into the Company's form S-8 registration statement file no. 333-154416, as such registration statement may be amended from time to time.

## Exhibits

- 99.1 AerCap Holdings N.V. interim financial report for the quarter ended June 30, 2009
- 99.2 Shareholders' Agreement, dated June 21, 2009, among AerCap AerVenture Holding B.V., Waha AV Participations B.V. and AerVenture Limited

## 2

## SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

## AERCAP HOLDINGS N.V.

By:/s/ Klaus HeinemannName:Klaus HeinemannTitle:Authorized Signatory

Date: September 1, 2009

3

## EXHIBIT INDEX

- 99.1 AerCap Holdings N.V. interim financial report for the quarter ended June 30, 2009
- 99.2 Shareholders' Agreement, dated June 21, 2009, among AerCap AerVenture Holding B.V., Waha AV Participations B.V. and AerVenture Limited

PART I	FINANCIAL INFORMATION	
Item 1.	Financial Statements (Unaudited)	2
Item 2.	Management's Discussion & Analysis of Financial Condition and Results of Operations	17
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	26
PART II	OTHER INFORMATION	29
Item 1.	Risk Factors	29
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	29
Item 3.	Defaults Upon Senior Securities	29
Item 4.	Submission of Matters to a Vote of Security Holders	29
Item 5	Other Information	29
Item 6	Exhibits	29
	1	

## PART I FINANCIAL INFORMATION

## Item 1. Financial Statements (Unaudited)

Unaudited Condensed Consolidated Balance Sheets as of June 30, 2008, December 31, 2008 and June 30, 20093Unaudited Condensed Consolidated Income Statements for the Three and Six months ended June 30, 2008 and June 30, 20094Unaudited Condensed Consolidated Cash Flow Statements for the Three and Six months ended June 30, 2008 and June 30, 20095Notes to the Unaudited Condensed Consolidated Financial Statements6

## **Unaudited Condensed Consolidated Balance Sheets**

## As of June 30, 2008, December 31, 2008 and June 30, 2009

	Note		June 30, 2008		December 31, 2008		June 30, 2009	
					ars in thousands exc nd per share amoun			
Assets						-~)		
Cash and cash equivalents		\$	175,870	\$	193,563	\$	218,423	
Restricted cash			183,808		113,397		128,184	
Trade receivables, net of provisions			40,642		43,649		39,244	
Flight equipment held for operating leases, net	5		3,765,378		3,989,629		4,493,793	
Flight equipment held for sale			48,390					
Net investment in direct finance leases					30,571		34,822	
Notes receivable, net of provisions	6		199,485		134,067		136,084	
Prepayments on flight equipment			328,172		448,945		576,754	
Investments			11,678		18,678		20,111	
Goodwill			6,776		6,776		6,776	
Intangibles			54,788		47,099		37,893	
Inventory			88,627		102,879		131,416	
Derivative assets			59,677		19,352		40,035	
Deferred income taxes			78,617		82,471		81,187	
Other assets	7		175,818		179,750		186,105	
Total Assets	14	\$	5,217,726	\$	5,410,826	\$	6,130,827	
Liabilities and Equity								
Accounts payable		\$	7,023	\$	7,510	\$	28,290	
Accrued expenses and other liabilities	8		107,865		104,750		76,559	
Accrued maintenance liability			248,517		202,834		206,873	
Lessee deposit liability			89,566		98,584		112,112	
Debt	9		3,580,002		3,790,487		4,336,966	
Accrual for onerous contracts			28,472		33,306		29,878	
Deferred revenue			40,042		34,922		36,805	
Derivative liabilities			2,584		12,378		8,285	
Deferred income taxes			7,779					
Commitments and contingencies	15							
Total Liabilities			4,111,850		4,284,771		4,835,768	
Ordinary share capital, €0.01 par value (200,000,000 ordinary shares authorized, 85,036,957 ordinary shares			699		699		699	
issued and outstanding)								
Additional paid-in capital			605,889		609,327		591,623	
Accumulated retained earnings	10		466,678		499,011		585,560	
Total AerCap Holdings N.V. Shareholders' Equity	10		1,073,266		1,109,037		1,177,882	
Non-controlling interest	10	_	32,610		17,018		117,177	
Total Equity	10	_	1,105,876		1,126,055		1,295,059	
Total Liabilities and Equity		\$	5,217,726	\$	5,410,826	\$	6,130,827	

The accompanying notes are an integral part of these condensed consolidated financial statements.

## **Unaudited Condensed Consolidated Income Statements**

## For the Three and Six Months Ended June 30, 2008 and 2009

		 Three months ended June 30,				Six montl June		
	Note	2008		2009		2008		2009
		(US dollars i	in tl	nousands, except	t sha	are and per sha	re a	mounts)
Revenues								
Lease revenue		\$ 144,346	\$	169,829	\$	288,202	\$	331,042
Sales revenue		180,725		111,635		323,188		153,352
Interest revenue		5,165		2,602		10,042		5,223
Management fee revenue		2,731		3,732		5,905		6,473
Other revenue		 386		6,897		549		7,107
Total Revenues	14	333,353		294,695		627,886		503,197
Expenses								
Depreciation	14	39,527		53,243		78,002		104,490
Asset impairment		7,689		13,733		7,689		20,950
Cost of goods sold		140,847		105,496		250,866		139,320
Interest on debt		19,628		5,989		69,224		35,475
Operating lease in costs		3,315		3,273		6,955		6,587
Leasing expenses		11,402		22,076		17,792		41,237
Provision for doubtful notes and accounts receivable		699		(879)		1,247		353
Selling, general and administrative expenses	11,12	 32,664	_	27,777		63,286		54,990
Total Expenses		255,771		230,708		495,061		403,402
Income from continuing operations before income taxes		77,582		63,987		132,825		99,795
Provision for income taxes		(6,955)		(827)		(11,525)		(2,687)
Net income		 70,627	-	63,160	_	121,300	-	97,108
Net (income) attributable to non-controlling interest		(2,031)		(6,564)		(1,828)		(10,558)
Net Income attributable to AerCap Holdings N.V.	13,14	\$ 68,596	\$		\$	119,472	\$	86,550
Basic and diluted earnings per share	13,14	\$ 0.81	\$	0.67	\$	1.40	\$	1.02
Weighted average shares outstanding, basic and diluted	13	35,036,957	\$	85,036,957		85,036,957		85,036,957

The accompanying notes are an integral part of these condensed consolidated financial statements.

4

## **Unaudited Condensed Consolidated Statements of Cash Flows**

## For the Three and Six Months Ended June 30, 2008 and 2009

		Three months ended June 30,				Six months ended June 30,			
		2008		2009		2008		2009	
				(US dollars i	n thou	isands)			
Net income	\$	70,627	\$	63,160	\$	121,300	\$	97,108	
Adjustments to reconcile net income to net cash provided by operating activities:									
Depreciation		39,528		53,243		78,002		104,490	
Asset impairment		7,689		13,733		7,689		20,950	
Amortization of debt issuance costs		3,365		4,054		6,757		7,888	
Amortization of intangibles		3,535		4,415		7,039		9,205	
Provision for doubtful notes and accounts receivable		699		(879)		1,247		353	
Capitalized interest on pre-delivery payments		(730)		(338)		(1,399)		(709)	
(Gain) loss on disposal of assets		(29,858)		570		(52,807)		1,018	
Mark-to-market of non-hedged derivatives		(13,943)		(18,502)		(11,076)		(19,504)	
Deferred taxes		6,556		(102)		10,990		1,139	
Share-based compensation		1,785		996		3,421		1,998	
Changes in assets and liabilities:									
Trade receivables and notes receivable, net		(723)		1,244		(20,963)		5,528	
Inventories		5,802		(17,205)		16,275		(2,721)	
Other assets and derivative assets		(23,944)		(6,543)		(25,490)		(10,730)	
Accounts payable and accrued expenses, including accrued maintenance liability and lessee deposits		11,534		13,977		7,813		2,047	
Deferred revenue		4,378		(3,328)		6,468		1,884	
Net cash provided by operating activities	_	86,300		108,495		155,266	_	219,944	
Purchase of flight equipment		(642,647)		(286,726)		(877,551)		(574,814)	
Proceeds from sale/disposal of assets		164,405		76,560		247,892		78,352	
Prepayments on flight equipment		(59,233)		(127,857)		(131,678)		(286,361)	
Purchase of intangibles		(12,895)				(21,522)			
Movement in restricted cash		(56,658)		16,770		(88,736)		(14,787)	
Net cash used in investing activities		(607,028)		(321,253)		(871,595)		(797,610)	
Issuance of debt		662,565		835,278		940,646		1,280,978	
Repayment of debt		(127,025)		(671,944)		(253,388)		(768,429)	
Debt issuance costs paid		(35,254)		(10,743)		(35,621)		(14,113)	
Capital contributions from non-controlling interests				104,200				104,200	
Net cash provided by financing activities		500,286		256,791		651,637		602,636	
Net (decrease) increase in cash and cash equivalents		(20,442)		44,033		(64,692)		24,970	
Effect of exchange rate changes		(858)		(691)		(1,174)		(110)	
Cash and cash equivalents at beginning of period		197,170		175,081		241,736		193,563	
Cash and cash equivalents at end of period	\$	175,870	\$	218,423	\$	175,870	\$	218,423	
Supplemental cash flow information									
Interest paid		28,390		23,365	\$	64,534		49,300	
Taxes paid (received)		128		(4,083)		185		(3,693)	

The accompanying notes are an integral part of these condensed consolidated financial statements.

#### Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

## 1. General

#### The Company

We are an integrated global aviation company, conducting aircraft and engine leasing and trading and parts sales. We also provide a wide range of aircraft management services to other owners of aircraft. We are headquartered in Amsterdam, The Netherlands, with principal offices in Shannon, Ireland, Ft. Lauderdale and Miami, Florida and Goodyear, Arizona.

These condensed consolidated financial statements include the accounts of AerCap Holdings N.V. and its subsidiaries. AerCap Holdings N.V. is a Netherlands public limited liability company ("*naamloze vennootschap*") formed on July 10, 2006 for the purpose of acquiring all of the assets and liabilities of AerCap Holdings C.V. AerCap Holdings C.V. is a limited partnership ("*commanditaire vennootschap*") formed under the laws of The Netherlands on June 27, 2005 for the purposes of acquiring the share capital, subordinated debt and senior debt of debis AirFinance B.V. ("AerCap B.V."), which occurred on June 30, 2005 (the "2005 Acquisition"). In anticipation of our initial public offering, we changed our corporate structure from a Netherlands partnership to a Netherlands public limited liability company. This change was effected through the acquisition of all of the assets and liabilities of AerCap Holdings C.V. by AerCap Holdings N.V. on October 27, 2006. In accordance with Statement of Financial Accounting Standards ("SFAS") 141, "*Business Combinations*", this acquisition was a transaction under common control and accordingly, AerCap Holdings N.V. recognized the acquisition of the assets and liabilities of AerCap Holdings C.V. at their carrying values and no goodwill or other intangible assets were recognized. Additionally in accordance with SFAS 141, these consolidated financial statements are presented as if AerCap Holdings N.V. had been the acquiring entity of AerCap B.V. on June 30, 2005. On November 27, 2006, we completed an initial public offering of 6,800,000 of our ordinary shares at \$23 per share generating net proceeds of \$143,017 which we used to repay debt.

#### Variable interest entities

In January 2006, we sold a 50% equity interest in AerVenture Ltd. ("AerVenture"), previously a wholly-owned entity, to LoadAir, a subsidiary of Al Fawares, an investment and construction company based in Kuwait. AerVenture had contracted with Airbus for the delivery of up to 70 A320 family aircraft, including five aircraft subject to reconfirmation rights, with the intent of leasing these aircraft to third parties. The joint venture agreement required us to make certain specified equity contributions and additional equity capital available to AerVenture depending on capital needs in the future. We have entered into agreements to provide management and marketing services to AerVenture in return for management fees. We determined that AerVenture was a variable interest entity for which we were the primary beneficiary. As such, we continued to consolidate AerVenture in our accounts since its inception date. In March 2009, LoadAir failed to make \$80.0 million in required capital contributions to AerVenture, and as a result, LoadAir lost its voting rights and economic rights in AerVenture. In addition, all of the directors appointed by LoadAir were automatically removed. In June 2009, the 50% investment of LoadAir was redeemed by AerVenture in conjunction with the sale of a 50% equity interest in AerVenture to a new joint venture partner, Waha Capital PJSC ("Waha Capital"). We have determined that AerVenture in our accounts.

## Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

## 2. Basis for presentation

Our financial statements are presented in accordance with accounting principles generally accepted in the United States of America.

We consolidate all companies in which we have a direct and indirect legal or effective control and all variable interest entities for which we are deemed the primary beneficiary under FIN 46R. All intercompany balances and transactions with consolidated subsidiaries have been eliminated. The results of consolidated entities are included from the effective date of control or, in the case of variable interest entities, from the date that we are or become the primary beneficiary. The results of subsidiaries sold or otherwise deconsolidated are excluded from the date that we cease to control the subsidiary or, in the case of variable interest entities, when we cease to be the primary beneficiary.

Other investments in which we have the ability to exercise significant influence and joint ventures are accounted for under the equity method of accounting.

The consolidated financial statements are stated in United States dollars, which is our functional currency.

Certain information and footnote disclosures required by U.S. GAAP for complete annual financial statements have been omitted and, therefore, it is suggested that these interim financial statements be read in conjunction with the Company's audited financial statements for the year ended December 31, 2008. In the opinion of management, these financial statements, which have been prepared pursuant to the rules of the SEC and U.S. GAAP for interim financial reporting, reflect all adjustments, which consisted only of normal recurring adjustments which were necessary to state fairly the results for the interim periods. The results of operations for the three and six months ended June 30, 2009 are not necessarily indicative of those for a full fiscal year.

The preparation of consolidated financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the amounts reported in the consolidated financial statements and accompanying notes. For us, the use of estimates is or could be a significant factor affecting the reported carrying values of flight equipment, inventory, intangibles, goodwill, investments, trade and notes receivable, deferred tax assets and accruals and reserves. Management utilizes professional appraisers and valuation experts, where possible, to support estimates, particularly with respect to flight equipment. Despite management's best efforts to accurately estimate such amounts, actual results could materially differ from those estimates.

## 3. Recent accounting pronouncements

## FSP No. 115-2 and FAS 124-2

In April 2009, the Financial Accounting Standards Board ("FASB") issued FSP No. 115-2/124-2, *Recognition and Presentation of Other-Than-Temporary Impairments*. FSP No. 115-2/124-2 is an amendment of SFAS No. 115, *Accounting for Certain Investments in Debt and Equity Securities* and SFAS No. 124, *Accounting for Certain Investments Held by Not-for-Profit Organizations*. FSP No. 115-2/124-2 amends the other-than-temporary impairment guidance for debt securities and expands the presentation and disclosure of other-than-temporary impairments on debt and equity securities in the financial statements. The effective date of FSP No. 115-2/124-2 is for interim and annual reporting periods ending after June 15, 2009, with early adoption permitted. The adoption of FSP No. 115-2/124-2 did not have an impact on our financial position or results of operation.



#### Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

#### FSP SFAS 157-4

In April 2009, the FASB issued FSP SFAS 157-4, "Determining Fair Value When the Volume and Level of Activity for the Asset or Liability Have Significantly Decreased and Identifying Transactions That Are Not Orderly" ("FSP SFAS 157-4"). FSP SFAS 157-4 provides guidance for estimating fair value in accordance with SFAS No. 157, "Fair Value Measurements" ("SFAS 157"), when the volume and level of activity for an asset or liability have significantly decreased and for identifying circumstances that indicate a transaction is not orderly. FSP SFAS 157-4 also requires extensive additional fair value disclosures. The adoption of FSP SFAS 157-4 did not have an impact on our financial position or results of operation.

## SFAS 165

In May 2009, the FASB issued Statement No. 165, *Subsequent Events* ("SFAS 165"). SFAS 165 addresses accounting and disclosure requirements related to subsequent events. SFAS 165 requires management to evaluate subsequent events through the date the financial statements are either issued or available to be issued, depending on the company's expectation of whether it will widely distribute its financial statements to its shareholders and other financial statement users. Companies are required to disclose the date through which subsequent events have been evaluated. SFAS 165 is effective for interim or annual financial periods ending after June 15, 2009 and should be applied prospectively. The adoption of SFAS 165 did not have a significant impact on our financial position or results of operation. The Company has performed an evaluation of subsequent events through September 1, 2009, which is the date the financial statements were issued."

#### SFAS 166

In June 2009, the FASB issued SFAS 166, "Accounting for Transfers of Financial Assets—an amendment of FASB Statement No. 140" ("SFAS 166"). SFAS 166 removes the concept of a qualifying special-purpose entity ("QSPE") from SFAS 140, "Accounting for Transfers and Servicing of Financial Assets and Extinguishments of Liabilities" and removes the exception from applying FIN 46R to QSPEs. SFAS 166 is effective for interim and annual periods beginning on January 1, 2010. Earlier application is prohibited. We believe the adoption will have no effect on our financial position or results of operation.

### SFAS 167

In June 2009, FASB issued SFAS 167, "Amendments to FASB Interpretation No. 46(R)" ("SFAS 167"). SFAS 167 amends the consolidation analysis with an approach focused on identifying which enterprise has the power to direct the activities of a VIE that most significantly affect the entity's economic performance and (i) the obligation to absorb losses of the entity or (ii) the right to receive benefits from the entity, and improves financial reporting by enterprises involved with VIEs. SFAS 167 is effective for interim and annual periods beginning on January 1, 2010 for us. Earlier application is prohibited. We are assessing the effect the adoption of SFAS 167 will have on our financial position or results of operation.

#### SFAS 168

In June 2009, the FASB issued Statement No. 168, *The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles* ("SFAS 168"). The FASB Accounting Standards Codification ("Codification") will become the source of authoritative U.S. GAAP recognized by the FASB to be applied by non-governmental entities. SFAS 168 will bring together in one place all authoritative GAAP previously held at different levels of GAAP. Rules and interpretive releases of the Securities and Exchange Commission (SEC) under authority of federal securities laws are also sources of authoritative GAAP for SEC registrants. On the effective date of this Statement, the Codification will supersede all then-existing non-SEC accounting and reporting standards. All other non-grandfathered non-SEC accounting literature not included in the Codification will become non-authoritative. This Statement is effective for financial statements issued for interim and annual periods ending after September 15, 2009. The adoption of SFAS 168 will not have a significant impact on our financial position or results of operation.

## Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

#### 4. Fair value measurements

In September 2006, the FASB issued SFAS 157, which is effective for fiscal years beginning after November 15, 2007. We adopted the standard on January 1, 2008.

Under SFAS 157, the Company determines fair value based on the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. It is the Company's policy to maximize the use of observable inputs and minimize the use of unobservable inputs when developing fair value measurements, in accordance with the fair value hierarchy as described below. Where limited or no observable market data exists, fair value measurements for assets and liabilities are based primarily on management's own estimates and are calculated based upon the Company's pricing policy, the economic and competitive environment, the characteristics of the asset or liability and other such factors. Therefore, the results may not be realized in actual sale or immediate settlement of the assets or liability.

The Company adopted SFAS 157 for all financial assets and liabilities required to be measured at fair value on a recurring basis prospectively from January 1, 2008. The application of SFAS 157 for financial instruments which are periodically measured at fair value did not have a material effect on the Company's results of operations or financial position.

Under SFAS 157, there is a hierarchal disclosure framework associated with the level of pricing observability utilized in measuring assets and liabilities at fair value.

The three broad levels defined by the SFAS 157 hierarchy are as follows:

Level 1 — Quoted prices available in active markets for identical assets or liabilities as of the reported date.

Level 2 — Observable market data. Inputs include quoted prices for similar assets, liabilities (risk adjusted) and market-corroborated inputs, such as market comparables, interest rates, yield curves and other items that allow value to be determined.

Level 3 — Unobservable inputs from the Company's own assumptions about market risk developed based on the best information available, subject to cost-benefit analysis. Inputs may include the Company's own data.

When there are no observable comparables, inputs used to determine value are derived through extrapolation and interpolation and other Company-specific inputs such as projected financial data and the Company's own views about the assumptions that market participants would use.

The following table summarizes our financial assets and liabilities as of June 30, 2009 that we measured at fair value on a recurring basis by level within the fair value hierarchy. As required by SFAS No. 157, assets and liabilities are classified in their entirety based on the lowest level of input that is significant to their fair value measurement.

	Ju	June 30, 2009		Level 1		Level 2	 Level 3
Cash and cash equivalents	\$	218,423	\$	218,423	\$		\$ 
Restricted cash		128,184		128,184			
Derivative assets		40,035		—		40,035	
Derivative liabilities		(8,285)				(8,285)	
	\$	378,357	\$	346,607	\$	31,750	\$ 

Our cash and cash equivalents, along with our restricted cash and cash equivalents balances, consists largely of money market securities that are considered to be highly liquid and easily tradable. These securities are valued using inputs observable in active markets for identical securities and are therefore classified as level 1 within our fair value hierarchy. Our derivative assets and liabilities included in level 2 consist of United States dollar denominated interest rate caps and foreign currency forward contracts swaps. Their fair values are determined by applying standard modeling techniques under the income approach to relevant market interest rates (cash rates, futures rates, swap rates) in effect at the period close to determine appropriate reset and discount rates. Changes in fair value are recognized immediately in income.

## Notes to the Unaudited Condensed Consolidated Financial Statements

#### (US dollars in thousands, except share and per share amounts)

We also measure the fair value of certain assets and liabilities on a non-recurring basis, when GAAP requires the application of fair value, including events or changes in circumstances that indicate that the carrying amounts of assets may not be recoverable. Assets subject to these measurements include aircraft. We record aircraft at fair value when we determine the carrying value may not be recoverable, in accordance with SFAS No. 144, "*Accounting for the Impairment or Disposal of Long-Lived Assets*" and other accounting pronouncements requiring remeasurements at fair value. Fair value measurements for aircraft in impairment tests are based on level 3 inputs, which include the Company's assumptions and appraisal data as to future cash proceeds from leasing and selling aircraft. In the six months ended June 30, 2009, we recognized an impairment of \$21.0 million. The impairment related to six older A320 aircraft which was triggered by the receipt of \$21.0 million of end-of-lease payments from the previous lessees. These end-of-lease payments were recorded as lease revenue during the six months ended June 30, 2009.

Our financial instruments consist principally of notes receivable, restricted cash, derivative assets and cash equivalents. The fair value of notes receivable, restricted cash and cash equivalents approximates the carrying value of these financial instruments because of their short term nature.

The fair values of our debt are estimated using a discounted cash flow analysis, based on our current incremental borrowing rates for similar types of borrowing arrangements.

The carrying amounts and fair values of our financial instruments at June 30, 2009 are as follows:

	 June 30, 2009				
	<b>Book value</b>		Fair value		
Assets					
Notes receivable	\$ 136,084	\$	136,084		
Restricted cash	128,184		128,184		
Derivative assets	40,035		40,035		
Cash and cash equivalents	218,423		218,423		
	\$ 522,726	\$	522,726		
Liabilities					
Debt	\$ 4,336,966	\$	4,029,049		
Derivative liabilities	8,285		8,285		
Guarantees	2,804		2,804		
	\$ 4,348,055	\$	4,040,138		



## Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

## 5. Flight equipment held for operating leases, net

At June 30, 2009 we owned 166 aircraft and 82 engines, which we leased under operating leases to 78 lessees in 39 countries. Movements in flight equipment held for operating leases during the periods presented were as follows:

	 months ended une 30, 2008	months ended une 30, 2009
Net book value at beginning of period	\$ 3,050,160	\$ 3,989,629
Additions	931,194	712,285
Depreciation	(75,711)	(101,712)
Impairment	(5,282)	(20,950)
Disposals	(156,080)	(77,181)
Transfers (to) from flight equipment held for sale	34,051	—
Transfer to inventory	(9,809)	(8,278)
Other (a)	 (3,145)	
Net book value at end of period	\$ 3,765,378	4,493,793
Accumulated depreciation/impairment at June 30, 2008 and 2009	(292,137)	(454,428)

(a) Onerous contract accruals were settled at a discount of \$3,145 in the six months ended June 30, 2008. This discount was applied to reduce the net book value of the related aircraft.

## 6. Notes receivable

Notes receivable consist of the following:

	December 31, June 30, 2008 2008				June 30, 2009	
Secured notes receivable	\$	6,700	\$	6,439	\$	7,555
Notes receivable in defeasance structures		190,785		126,301		127,410
Notes receivable from lessee restructurings		2,000		1,327		1,119
	\$	199,485	\$	134,067	\$	136,084

## 7. Other assets

Other assets consist of the following:

	June 30, 2008		December 31, 2008		Ju	ne 30, 2009
Debt issuance costs	\$	99,044	\$	99,486	\$	105,730
Other tangible fixed assets		16,259		16,313		14,456
Receivables from aircraft manufacturer		29,280		25,912		29,256
Prepaid expenses		5,696		7,428		6,124
Current tax receivable				5,356		—
Other receivables		25,539		25,255		30,539
	\$	175,818	\$	179,750	\$	186,105

## Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

## 8. Accrued expenses and other liabilities

Accrued expenses and other liabilities consist of the following:

	June 3	0, 2008	D	ecember 31, 2008	Ju	ne 30, 2009
Guarantee liability	\$	3,592	\$	3,219	\$	2,804
Accrued expenses		42,504		57,851		42,541
Accrued interest		8,811		13,608		10,971
Lease deficiency		14,154		12,574		10,439
Deposits under forward sale agreements		38,804		17,498		9,804
	\$	107,865	\$	104,750	\$	76,559

## 9. Debt

Debt consists of the following:

	Ju	June 30, 2008		December 31, 2008		une 30, 2009
ECA-guaranteed financings	\$	575,398	\$	636,813	\$	815,759
JOL financings		93,487		91,095		88,603
AerVenture pre-delivery payment facility-Calyon		120,802		96,432		21,461
AerVenture pre-delivery payment facility-HSH				68,109		178,538
A330- pre-delivery payment facilities		79,997		121,027		204,851
UBS revolving credit facility		303,330		477,277		259,810
AeroTurbine revolving credit facility		129,438		194,188		319,168
Calyon aircraft acquisition facility		127,076		211,346		137,385
TUI portfolio acquisition facility		425,730		407,804		389,238
Subordinated debt joint venture partner		62,800		61,921		62,704
Engine warehouse facility				53,300		46,164
Commercial bank debt		204,169		124,358		142,888
Aircraft Lease Securitisation debt		1,267,201		1,120,516		1,043,841
Aircraft Lease II Securitisation debt				—		499,146
Capital lease obligations under defeasance structures		190,574		126,301		127,410
	\$	3,580,002	\$	3,790,487	\$	4,336,966



## Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

## 10. Equity

Movements in equity during the periods presented were as follows:

	Six months ended June 30, 2008					
	AerCap Holdings N.V. Shareholders' Equity		Non- controlling interest		Т	otal Equity
Beginning of the period	\$	950,373	\$	30,782	\$	981,155
Net income for the period		119,472		1,828		121,300
Share-based compensation		3,421				3,421
End of the period	\$	1,073,266	\$	32,610	\$	1,105,876

		Twelve months ended December 31, 2008							
	AerCapHoldings N.V.Non-Shareholders'controllingEquityinterest				1	Fotal Equity			
Beginning of the period	\$	950,373	\$	30,782	\$	981,155			
Net income (loss) for the period		151,806		(10,883)		140,923			
Share-based compensation		6,858				6,858			
Capital contributions from non-controlling interests				5,000		5,000			
Purchase of non-controlling interests			_	(7,881)		(7,881)			
End of the period	\$	1,109,037	\$	17,018	\$	1,126,055			

	Six months ended June 30, 2009								
	AerCap Holdings N.V. Shareholders' Equity		Non- controlling interest		Total Equity				
Beginning of the period	\$ 1,109,037	\$	17,018	\$	1,126,055				
Net income for the period	86,550		10,558		97,108				
Share-based compensation	1,998				1,998				
Default AerVenture partner (a)	25,078		(25,078)		—				
Sale to new AerVenture partner (b)	 (44,781)		114,679		69,898				
End of the period	\$ 1,177,882	\$	117,177		1,295,059				

<sup>(</sup>a) In March 2009, LoadAir failed to make \$80.0 million in required capital contributions to AerVenture, and as a result, LoadAir lost its voting rights and economic rights in AerVenture with the exception of certain rights to limited residual payments upon liquidation of AerVenture. As of March 31, 2009 AerVenture was a wholly owned subsidiary. The default of LoadAir increased AerCap Holdings N.V Shareholders' Equity by \$25,078, through the elimination of the related non-controlling interest.

<sup>(</sup>b) In June 2009, we sold 50% of AerVenture to Waha Capital. The sale to Waha Capital decreased AerCap Holdings N.V Shareholders' Equity by \$44,781, through the establishment of the related non-controlling interest.

## Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

## 11. Share-based compensation

## **Bermuda Equity Grants**

There were no additional restricted shares or share options issued under the Bermuda Equity Plan during the six months ended June 30, 2009. The table below indicates the number of options outstanding under the Bermuda Equity Plan which are still subject to expense recognition under FAS 123R, stated in equivalent shares of AerCap Holdings N.V. into which such options are exercisable and exchangeable:

	Vested	Unvested	Per Share
	Options	Options	Strike Price
AerCap Holdings N.V. equivalent shares	561,476	319,459	\$ 7.00

Assuming that established performance criteria are met for 2009, we expect to recognize share-based compensation related to the share options above of \$587 during the remainder of 2009.

## AerCap Holdings N.V. Equity Grants

No additional stock options were issued under the NV Equity Plan during the six months ended June 30, 2009. At June 30, 2009, there were 2.4 million stock options outstanding at an exercise price of \$24.63 per share, 100,000 stock options outstanding at an exercise price of \$15.03 per share and 700,000 stock options outstanding at an exercise price of \$2.95 per share. At June 30, 2009, 300,000 outstanding options were vested, 312,500 options were subject to performance criteria which were not met and were therefore unexercisable and 2,587,500 options are met in the future, including performance criteria, and that no forfeitures occur, we expect to recognize share-based compensation charges related to NV Equity Grants of approximately \$1,811 during the remainder of 2009 and approximately \$3,610, \$2,660 and \$34 during the years 2010, 2011 and 2012, respectively.



## Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

## 12. Selling, general and administrative expenses

Selling, general and administrative expenses include the following expenses:

	Three months ended June 30, 2008	Three months ended June 30, 2009			Six months ended June 30, 2008	Six months ended June 30, 2009
Personnel expenses(a)	\$ 18,172(a)	\$	15,660(a)	\$	36,273(a)	\$ 30,089(a)
Travel expenses	2,616		1,792		4,347	3,493
Professional services	5,896		3,964		10,511	8,337
Office expenses	2,820		2,717		5,509	4,869
Directors expenses	866		871		1,704	1,622
Other expenses	 2,294		2,773		4,939	 6,580
	\$ 32,664		27,777	\$	63,286	\$ 54,990

(a) Includes share-based compensation of \$1,785, \$996, \$3,421 and \$1,998 in the three and six months ended June 30, 2008 and 2009, respectively.

## 13. Earnings per common share

Basic and diluted earnings per share is calculated by dividing net income by the weighted average of our common shares outstanding. As disclosed in Note 11, there are 3.2 million share options outstanding under the NV Equity Plan. These options could become dilutive in the future. The computations of basic and diluted earnings per common share for the periods indicated below are shown in the following table:

	Three months ended June 30, 2008		Three months ended June 30, 2009		Six months ended June 30, 2008		Six months ended June 30, 2009	
Net income for the computation of basic and diluted earnings per share	\$	68,596	\$	56,596	\$	119,472	\$	86,550
Weighted average common shares outstanding		85,036,957		85,036,957		85,036,957		85,036,957
Basic and diluted earnings per common share	\$	0.81	\$	0.67	\$	1.40	\$	1.02

15

## Notes to the Unaudited Condensed Consolidated Financial Statements

## (US dollars in thousands, except share and per share amounts)

## 14. Segment information

## Reportable Segments

We manage our business, analyze and report our results of operations on the basis of two business segments—leasing, financing, sales and management of commercial aircraft ("Aircraft") and leasing, financing and sales of engines and parts ("Engine and parts").

The following sets forth significant information from our reportable segments:

	 Six months ended June 30, 2008								
	Aircraft	Engin	es and parts	Total					
Revenues from external customers	\$ 535,374	\$	92,512	\$	627,886				
Segment profit	114,336		5,136		119,472				
Segment assets	4,773,603		444,123		5,217,726				
Depreciation	 71,411		6,591		78,002				

		Six months ended June 30, 2009							
	Aircraft Engines and parts		Total						
Revenues from external customers	\$	386,356	\$	116,841	\$	503,197			
Segment profit		83,534		3,016		86,550			
Segment assets		5,630,155		500,672		6,130,827			
Depreciation		96,888		7,602		104,490			

## 15. Commitments and contingencies

A detailed summary of our commitments and contingencies can be found in our 2008 Annual Report on Form 20-F filed with the SEC on April 1, 2009. There have been no material changes to our commitments and contingencies since the filing of those reports.

## 15. Subsequent events

No subsequent events have occurred as of September 1, 2009.



## Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read this discussion in conjunction with our unaudited condensed consolidated financial statements and the related notes included in this Interim Report. Our financial statements are presented in accordance with generally accepted accounting principles in the United States of America, or U.S. GAAP and are presented in U.S. dollars.

## **Special Note About Forward Looking Statements**

This report includes "forward looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. We have based these forward looking statements largely on our current beliefs and projections about future events and financial trends affecting our business. Many important factors, in addition to those discussed in this report, could cause our actual results to differ substantially from those anticipated in our forward looking statements, including, among other things:

- the availability of capital to us and to our customers and changes in interest rates,
- the ability of our lessees and potential lessees to make operating lease payments to us,
- our ability to successfully negotiate aircraft and engine purchases, sales and leases, to collect outstanding amounts due and to repossess aircraft and engines under defaulted leases, and to control costs and expenses,
- decreases in the overall demand for commercial aircraft and engine leasing and aircraft management services,
- the economic condition of the global airline and cargo industry,
- competitive pressures within the industry,
- the negotiation of aircraft management services contracts,
- regulatory changes affecting commercial aircraft operators, aircraft maintenance, engine standards, accounting standards and taxes, and
- the risks set forth in "Item 3. Key Information—Risk Factors" included in our Annual Report on Form 20-F, filed with the SEC on April 1, 2009.

The words "believe", "may", "will", "aim", "estimate", "continue", "anticipate", "intend", "expect" and similar words are intended to identify forward looking statements. Forward looking statements include information concerning our possible or assumed future results of operations, business strategies, financing plans, competitive position, industry environment, potential growth opportunities, the effects of future regulation and the effects of competition. Forward looking statements speak only as of the date they were made and we undertake no obligation to update publicly or to revise any forward looking statements because of new information, future events or other factors. In light of the risks and uncertainties described above, the forward looking events and circumstances described in this annual report might not occur and are not guarantees of future performance.

## Aircraft Portfolio

As of June 30, 2009, we owned and managed 221 aircraft. We owned 166 aircraft and managed 55 aircraft in our aircraft business. As of June 30, 2009, we leased these aircraft to 75 commercial airlines and cargo operator customers in 40 countries. In addition, as of June 30, 2009, we had 38 new Airbus A320 family narrow-body aircraft and 27 new Airbus A330 wide-body aircraft on order. We also entered into a purchase contract for three aircraft and had executed letters of intent for the purchase of two aircraft. Including all owned and managed aircraft, aircraft under contract or letter of intent and aircraft in our order book, our portfolio totals 291 aircraft as of June 30, 2009.

	Owned	portfolio	Managed portfolio		Number of aircraft under	
Aircraft type	Number of aircraft owned	Percentage of total net book value	Number of aircraft	Number of aircraft on order	purchase contract or letter of intent	Total owned, Managed and ordered aircraft
Airbus A300 Freighter	1	0.7%		_		1
Airbus A319	18	12.7%		6		24
Airbus A320	69	41.2%	16	29	1	115
Airbus A321	16	10.8%	1	3	—	20
Airbus A330	8	11.3%		27	_	35
Boeing 737Classics	14	2.8%	30	—		44
Boeing 737NGs	18	13.4%		—	3	21
Boeing 747 Freighter	—	—			1	1
Boeing 757	11	3.3%	3	—		14
Boeing 767	4	2.7%	2	—		6
MD-11 Freighter	1	0.7%	1		_	2
MD-82	2	0.1%	1			3
MD 83	4	0.3%	1	_		5
Total	166	100.0%	55	65	5	291

In July 2008, we entered into an agreement with Airbus Freighter Conversions GmbH ("AFC") whereby AFC would convert 30 of our older Airbus A320s and A321s from passenger to freighter aircraft. Delivery of the first converted aircraft is expected to take place in 2011, with the remaining 29 aircraft scheduled for conversion between 2012 and 2015. In the future we may choose to acquire additional freighter aircraft or continue to convert some of our older A320 and A321 aircraft to freighter aircraft.

## **Engine** Portfolio

We maintain a diverse inventory of high-demand, modern and fuel-efficient engines. As of June 30, 2009, we owned 82 engines and had one new engine on order through our wholly owned subsidiary AeroTurbine. Our engine portfolio consists primarily of CFM56 series engines, one of the most widely used engines in the commercial aviation market. As of June 30, 2009, 60 of our 82 engines were CFM56 series engines manufactured by CFM International.

## Inventory

Our inventory consists of aircraft and engine parts. The aircraft and engine parts sales allow us to increase value of our aircraft and engine assets by putting each sub-component (engines, airframes and related parts) to its most profitable use. We sell aircraft and engine parts primarily to parts distributors and maintenance, repair and overhaul ("MRO") service providers.

## **Critical Accounting Policies**

There have been no changes to our critical accounting policies from those disclosed in our 2008 Annual Report on Form 20-F filed with the SEC on April 1, 2009, except the following:

## Derivative financial instruments

We use derivative financial instruments to manage our exposure to interest rate risks and foreign currency risks. Derivatives are accounted for in accordance with SFAS 133, *Accounting for Derivative Instruments and Hedging Activities*.

All derivatives are recognized on the balance sheet at their fair value. Changes in fair values between periods are currently recognized as a reduction or increase of interest expense on the income statement, as we do not currently apply hedge accounting to our current derivatives. In the future we may enter into derivatives for which we may apply hedge accounting. Changes in fair values between periods for derivatives for which we would apply cash flow hedge accounting would be recognized as a reduction or increase of accumulated other comprehensive income. Net cash received or paid under derivative contracts where material in any reporting period is classified as operating cash flow in our consolidated cash flow statements.

## **Comparative Results of Operations**

	Six months ended June 30,				
		2008		2009	
	(US dollars in thousands, except shar and per share amounts)				
Revenues					
Lease revenue	\$	288,202	\$	331,042	
Sales revenue		323,188		153,352	
Interest revenue		10,042		5,223	
Management fee revenue		5,905		6,473	
Other revenue		549		7,107	
Total Revenues		627,886		503,197	
Expenses					
Depreciation		78,002		104,490	
Asset impairment		7,689		20,950	
Cost of goods sold		250,866		139,320	
Interest on debt		69,224		35,475	
Operating lease in costs		6,955		6,587	
Leasing expenses		17,792		41,237	
Provision for doubtful notes and accounts receivable		1,247		353	
Selling, general and administrative expenses		63,286		54,990	
Total Expenses		495,061		403,402	
Income from continuing operations before income taxes		132,825		99,795	
Provision for income taxes		(11,525)		(2,687)	
Net income		121,300		97,108	
Net (income) attributable to non-controlling interest		(1,828)		(10,558)	
Net Income attributable to AerCap Holdings N.V,	\$	119,472	\$	86,550	
Basic and diluted earnings per share	\$	1.40	\$	1.02	
Weighted average shares outstanding, basic and diluted		85,036,957		85,036,957	

## Six months ended June 30, 2009 compared to six months ended June 30, 2008

Revenues. The principal categories of our revenue and their variances were:

	 onths ended e 30, 2008	 x months ended June 30, 2009		Increase/ (decrease)	Percentage Difference
		(US dollars in	millio	ons)	
Lease revenue:					
Basic rents	\$ 253.1	\$ 282.8	\$	29.7	11.7 %
Maintenance rents and end-of-lease compensation	35.1	48.3		13.2	37.6 %
Sales revenue	323.2	153.4		(169.8)	(52.2)%
Interest revenue	10.0	5.2		(4.8)	(48.0)%
Management fee revenue	5.9	6.5		0.6	10.2 %
Other revenue	0.5	7.1		6.6	1320.0 %
Total	\$ 627.8	\$ 503.3	\$	(124.5)	(19.8)%

• Basic rents increased by \$29.7 million, or 11.7%, to \$282.8 million in the six months ended June 30, 2009 from \$253.1 million in the six months ended June 30, 2008. The increase in basic rents was attributable primarily to:

• the acquisition between January 1, 2008 and June 30, 2009 of 79 aircraft for lease with an aggregate net book value of \$2.2 billion at the date of acquisition, partially offset by the sale of 28 aircraft, during such period, with an aggregate net book value of \$0.4 billion at the date of sale. The net increase in our aircraft portfolio resulted in a \$51.1 million increase in basic rents;

## partially offset by

- a decrease in payments from leases with lease rates tied to floating interest rates in the six months ended June 30, 2009 due to decreases in market interest rates, which resulted in a \$9.7 million decrease in basic rents;
- a decrease of \$4.0 million in basic rents in the six months ended June 30, 2009, resulting from the decrease in our engine lease activities;
- a decrease in basic rents of \$3.1 million in the six months ended June 30, 2009 as a result of airline defaults which occurred in 2008.
- Maintenance rents and end-of-lease compensation increased by \$13.2 million, or 37.6%, to \$48.3 million in the six months ended June 30, 2009 from \$35.1 million in the six months ended June 30, 2008. The increase in maintenance rents is attributable to the termination of several leases, which resulted in the recording of \$21.0 million of incremental maintenance rents.
- Sales revenue decreased by \$169.8 million, or 52.2%, to \$153.4 million in the six months ended June 30, 2009 from \$323.2 million in the six months ended June 30, 2008. The decrease in sales revenue is mainly a result of reduced aircraft sales in the six months ended June 30, 2009, due, in large part, to a reduction in liquidity in the aircraft trading market. Sales revenue in the six months ended June 30, 2009 was generated from the sale of two aircraft, six engines and parts inventory. In the six months ended June 30, 2009, we sold two A321 aircraft, whereas in the six months ended June 30, 2009, we sold two A321 aircraft, whereas in the six months ended June 30, 2009, we sold two A321 aircraft, whereas in the six months ended June 30, 2009, we sold two A321 aircraft, whereas in the six months ended June 30, 2009, we sold two A321 aircraft, whereas in the six months ended June 30, 2009, we sold two A321 aircraft, two Boeing 737 aircraft, one MD 83 aircraft, two MD82 aircraft, one DC8 aircraft and two Fokker 100 aircraft.
- Interest revenue decreased by \$4.8 million, or 48.0%, to \$5.2 million in the six months ended June 30, 2009 from \$10.0 million in the six months ended June 30, 2008. The decrease was mainly caused by a decrease in deposit rates of interest.
- Management fee revenue did not materially change in the six months ended June 30, 2009 compared to the six months ended June 30, 2008.
- Other revenue increased by \$6.6 million to \$7.1 million in the six months ended June 30, 2009 from \$0.5 million in the six months ended June 30, 2008. The increase was caused by the sale of one A320 forward order position, the sale of an investment in shares and other items.

*Depreciation.* Depreciation increased by \$26.5 million, or 34.0%, to \$104.5 million in the six months ended June 30, 2009 from \$78.0 million in the six months ended June 30, 2008, due primarily to the acquisition of 79 new aircraft between January 1, 2008 and June 30, 2009 with a book value at the time of the acquisition of \$2.2 billion. The increase was partially offset by the sale of 28 aircraft with a book value at the time of \$0.4 billion.

*Asset Impairment.* In the six months ended June 30, 2009, we recognized an impairment of \$21.0 million. The impairment related to six older A320 aircraft which was triggered by the receipt of \$21.0 million of end-of-lease payments from the previous lessees. Asset impairment was \$7.7 million in the six months ended June 30, 2008 and related to four MD82 aircraft, which came off-lease and were subsequently sold, and six engines which came off-lease and were subsequently parted out..

*Cost of Goods Sold.* Cost of goods sold decreased by \$111.6 million, or 44.5%, to \$139.3 million in the six months ended June 30, 2009 from \$250.9 million in the six months ended June 30, 2008. The decrease in cost of goods sold is mainly a result of the significant decrease in aircraft sales.

*Interest on Debt.* Our interest on debt decreased by \$33.7 million, or 48.7%, to \$35.5 million in the six months ended June 30, 2009 from \$69.2 million in the six months ended June 30, 2008. The majority of the decrease in interest on debt was mainly caused by:

• a decrease in our average cost of debt of 1.9% to 2.8% in the six months ended June 30, 2009 from 4.7% in the six months ended June 30, 2008. The decrease in our average cost of debt results from the use of caps as part of

our hedging strategy in combination with a decrease in interest rates. This resulted in a \$30.2 million decrease in our interest on debt;

• a \$16.8 million increase in the non-cash recognition of mark-to-market gains on derivatives to a \$22.2 million gain in the six months ended June 30, 2009 from a \$5.4 million gain in the six months ended June 30, 2008;

## partially offset by

• an increase in the average outstanding debt balance to \$4.1 billion in the six months ended June 30, 2009 from \$3.1 billion in the six months ended June 30, 2008, resulting in a \$13.8 million increase in our interest on debt.

*Other Operating Expenses.* Our other operating expenses increased by \$22.2 million, or 85.7%, to \$48.1 million in the six months ended June 30, 2009 from \$25.9 million in the six months ended June 30, 2008. The principal categories of our other operating expenses and their variances were as follows:

	onths ended e 30, 2008		x months ended June 30, 2009	Increase/ (decrease)	Percentage Difference		
	(US dollars in millions)						
Operating lease in costs	\$ 6.9	\$	6.6	(0.3)	(4.3)%		
Leasing expenses	17.8		41.2	23.4	131.5 %		
Provision for doubtful notes and accounts							
receivable	 1.2		0.3	(0.9)	(75.0)%		
Total	\$ 25.9	\$	48.1	22.2	85.7 %		

Our operating lease in costs did not materially change in the six months ended June 30, 2009 compared to the six months ended June 30, 2008.

Our leasing expenses increased by \$23.4 million, or 131.5%, to \$41.2 million in the six months ended June 30, 2009 from \$17.8 million in the six months ended June 30, 2008. The increase is primarily due to expenses of \$11.3 million incurred in relation to airline defaults which occurred in 2008 plus an increase in lessor contributions and transition expenses.

Our provision for doubtful notes and accounts receivable decreased by \$0.9 million, or 75.0%, to \$0.3 million in the six months ended June 30, 2009 from \$1.2 million in the six months ended June 30, 2008. We did not have defaults that significantly affected the provision for doubtful notes and accounts receivable in the six months ended June 30, 2008 and 2009.

*Selling, General and Administrative Expenses.* Our selling, general and administrative expenses decreased by \$8.3 million, or 13.6%, to \$55.0 million in the six months ended June 30, 2009 from \$63.3 million in the six months ended June 30, 2008, due primarily to (i) the decrease in the USD/EUR exchange rate in the six months ended June 30, 2009 as compared to the six months ended June 30, 2008, and (ii) a reduction in the number of employees during the twelve month period ended June 30, 2009.

*Income From Continuing Operations Before Income Taxes.* For the reasons explained above, our income from continuing operations before income taxes decreased by \$33.0 million, or 24.9%, to \$99.8 million in the six months ended June 30, 2009 from \$132.8 million in the six months ended June 30, 2008.

*Provision for Income Taxes.* Our provision for income taxes decreased by \$8.8 million or 76.5% to \$2.7 million in the six months ended June 30, 2009 from \$11.5 million in the six months ended June 30, 2008. Our effective tax rate for the six months ended June 30, 2009 was 2.7% and was 8.7% for the six months ended June 30, 2008. Our effective tax rate in any period is impacted by the mix of operations among our different tax jurisdictions.

*Net Income.* For the reasons explained above, net income decreased by \$24.1 million, or 19.9%, to \$97.1 million in the six months ended June 30, 2009 from \$121.3 million in the six months ended June 30, 2008.

## Liquidity and Access to Capital

## Liquidity and Capital Resources

Our cash balance at June 30, 2009 was \$346.6 million including restricted cash of \$128.2 million and our operating cash flow was \$219.9 million for the six months ended June 30, 2009. Our unused lines of credit at June 30, 2009 were approximately \$3.6 billion. Our debt balance at June 30, 2009 was \$4.3 billion and the average interest rate on our debt, excluding the effect of mark-to-market movements on our interest rate caps during the six months ended June 30, 2009, was 2.8%. Our debt to equity ratio was 3.3 to 1 as of June 30, 2009.

We satisfy our liquidity requirements through several sources, including:

- lines of credit and other secured borrowings;
- aircraft and engine lease revenues;
- sales of aircraft, engines and parts;
- supplemental maintenance rent and security deposits provided by our lessees; and
- management fee revenue.

In order to access the required capital to meet our obligations under our forward purchase commitments, we have completed or have undertaken several initiatives as more fully described in our Annual Report on Form 20-F, filed with the SEC on April 1, 2009.

Since our 2008 20-F filing we have completed the following initiatives:

- On April 3, 2009, we sold two A321 aircraft which were owned by a non-restricted cash entity. The transaction generated unrestricted cash proceeds of \$10.9 million.
- In June 2009, we signed a joint-venture agreement and sold 50% of our ownership in AerVenture to Waha Capital.
- In June 2009, we signed a facility agreements with HSH Nordbank AG for \$221 million of pre-delivery financing on ten A330 aircraft.
- In July 2009, we signed a letter of intent for the sale of three A330-300 aircraft delivering in 2010. The cash and profit related to this sale will be recognized at the time of physical delivery of the respective aircraft.

## Cash Flows

		Six months ended June 30, 2008		x months ended June 30, 2009
		s)		
Net cash flow provided by operating activities	\$	155.3	\$	219.9
Net cash flow used in investing activities		(871.6)		(797.6)
Net cash flow provided by financing activities		651.6		602.6

Six months ended June 30, 2009 compared to Six months ended June 30, 2008.

*Cash Flows Provided by Operating Activities.* Our cash flows provided by operating activities increased by \$64.6 million, or 41.6%, to \$219.9 million for the six months ended June 30, 2009 from \$155.3 million for the six months ended June 30, 2008. The primary reasons for the increase are: (i) an increase in our aircraft portfolio and (ii) a decrease of our interest expenses.

*Cash Flows Used in Investing Activities.* Our cash flows used in investing activities decreased by \$74.0 million, or 8.5%, to \$797.6 million in the six months ended June 30, 2009 from \$871.6 million in the six months ended June 30, 2008, primarily due to (i) an decrease of \$73.9 million in the movement in restricted cash in the six months ended June 30, 2009 as

compared to the six months ended June 30, 2008, and (ii) a decrease of \$154.7 million in the net cash used in aircraft purchase and sale activity (including intangible lease premiums) in the six months ended June 30, 2009 as compared to the six months ended June 30, 2008 partially offset by an increase of \$154.7 million in the amount of pre-delivery payments made in the six months ended June 30, 2009 as compared to the six months ended June 30, 2008.

*Cash Flows Provided by Financing Activities.* Our cash flows provided by financing activities decreased by \$49.0 million, or 7.5%, to \$602.6 million in the six months ended June 30, 2009 from \$651.6 million in the six months ended June 30, 2008. This decrease is attributable to a decrease of \$153.2 million in new financing proceeds, net of repayments and debt issuance costs in the six months ended June 30, 2009 as compared to the six months ended June 30, 2008 partially offset by the receipt of \$104.2 million from our new AerVenture partner, Waha Capital.

## Indebtedness

As of June 30, 2009, our outstanding indebtedness totaled \$4.3 billion and primarily consisted of export credit facilities, commercial bank debt, revolving credit debt, securitization debt and capital lease structures.

The following table provides a summary of our indebtedness at June 30, 2009:

Debt Obligation	Collateral	C	Commitment Outstanding				Undrawn amounts	Final stated Maturity
			(US					
Export credit facilities—financings	23 aircraft	\$	2,817,776	\$	815,759	\$	2,002,017	2021
Japanese operating lease financings	3 aircraft		88,603		88,603			2015
AerVenture A320 Pre-delivery payment								
facilities			257,020		199,999		57,021	2011
Airbus A330 Pre-delivery payment								
facilities	_		487,917		204,851		283,066	2010
UBS revolving credit facility	8 aircraft		1,000,000		259,810		740,190	2014
AeroTurbine revolving credit facility	66 engines &							
	10 aircraft		328,000		319,168		8,832	2012
Calyon Aircraft Acquisition facility	20 aircraft		140,818		137,385		3,433	2014
TUI Portfolio Acquisition facility	19 aircraft		389,238		389,238			2015
TUI Portfolio Subordinated debt*	—		62,704		62,704		—	2015
Engine Acquisition facility	8 engines		92,684		46,164		46,520	2013
Commercial bank debt	9 aircraft		142,888		142,888			2019
Aircraft Lease Securitisation debt	62 aircraft		1,043,841		1,043,841			2032
Aircraft Lease Securitisation II Limited								
debt	16 aircraft		1,000,000		499,146		500,854	2038
Capital lease obligations under								
defeasance structures	3 aircraft		127,410		127,410			2010
Total		\$	7,978,899	\$	4,336,966	\$	3,641,933 **	

\* Subordinated debt issued to our joint venture partner relating to the TUI portfolio acquisition.

\*\* The undrawn amounts of our current debt facilities consist of collateralized term debt available to finance pre-delivery payments and the most significant portion of the purchase price of aircraft and engines.

## **Contractual Obligations**

Our contractual obligations consist of principal and interest payments on debt, executed purchase agreements to purchase aircraft, operating lease rentals on aircraft under lease in/lease out structures and rent payments pursuant to our office leases. We intend to fund our contractual obligations through our lines of credit and other borrowings as well as internally generated cash flows. We believe that our sources of liquidity will be sufficient to meet our contractual obligations.

The following table sets forth our contractual obligations and their maturity dates as of June 30, 2009:

	 2009 (07/01/2009- 12/31/2009)		2010		2011		2012		Thereafter
		(US dollars in thousands)							
Debt (1)	\$ 411,935	\$	907,765	\$	464,975	\$	867,686	\$	1,994,204
Purchase obligations (2)	879,764		1,512,637		331,574		257,892		152,370
Operating leases (3)	11,680		26,469		26,376		14,402		14,175
Derivative obligations	 2,525		2,306		(5,175)		(7,970)		(21,604)
Total	\$ 1,305,904	\$	2,449,177	\$	817,750	\$	1,132,010	\$	2,139,145

(1) Includes estimated interest payments based on one-month LIBOR and three-month LIBOR as of June 30, 2009, which were 0.309% and 0.595%, respectively.

(2) Includes 27 new A330 wide-body aircraft on order from Airbus, 38 new A320 family aircraft on order from Airbus by AerVenture and three Boeing 737 next generation aircraft.

(3) Represents contractual operating lease rentals on aircraft under lease in/out structures and contractual payments on our office and facility leases in Amsterdam, The Netherlands, Miami, Florida, Fort Lauderdale, Florida, Goodyear, Arizona and Shannon, Ireland.

The table below provides information as of June 30, 2009 regarding our debt and interest obligations per facility type:

	 2009 (07/01/2009- 12/31/2009)	 <u>2010</u> (US	dolla	2011 ars in thousand	ds)	2012	 Thereafter
Pre-delivery payment facilities (1)	\$ 156,563	\$ 251,875	\$	856	\$	—	\$ 
Debt facilities with non-scheduled	161 560	217 015		205 425		201 171	1 000 000
amortization (2)	161,569	317,815		305,435		284,474	1,099,888
Capital lease obligations under defeasance							
structures (3)	8,302	129,104				—	
Other facilities	 85,501	 208,971		158,684		583,212	 894,316
Total	\$ 411,935	\$ 907,765	\$	464,975	\$	867,686	\$ 1,994,204

(1) Repayment of debt owed on pre-delivery payment facilities is essentially offset by proceeds received from aircraft purchase debt facilities.

- (2) Debt amortization is due only to the extent that cash is available in these facilities.
- (3) Obligations are defeased through an offsetting notes receivable amount.

## **Capital Expenditures**

The table below sets forth our expected capital expenditures for future periods indicated based on contracted commitments as of June 30, 2009:

	2009								
	(07/01/2009- 12/31/2009)		2010		2011		2012		Thereafter
	 (US dollars in thousands)								
Capital expenditures	\$ 777,615	\$	1,357,416	\$	222,979	\$	190,919	\$	152,370
Pre-delivery payments	 102,149		155,221		108,595		66,973		
Total	\$ 879,764	\$	1,512,637	\$	331,574	\$	257,892	\$	152,370

As of June 30, 2009, we expect to make capital expenditures related to the 27 A330 aircraft, three A321 aircraft, 29 A320 aircraft, six A319 aircraft and three Boeing 737 aircraft on order between 2009 and 2011. As we implement our growth strategy and expand our aircraft and engine portfolio, we expect our capital expenditures to increase in the future. We anticipate that we will fund these capital expenditures through internally generated cash flows, draw downs on our committed revolving credit facilities and the incurrence of bank, and other debt and equity issuances.

## **Off-Balance Sheet Arrangements**

As of June 30, 2009, we were obligated to make sublease payments under four aircraft operating leases of aircraft with lease expiration dates between 2009 and 2013. We lease these four aircraft to aircraft operators. Since we are not fully exposed to the risks and rewards of ownership of these aircraft, we do not include these aircraft on our balance sheet. In addition, we do not recognize a financial liability for our operating lease obligations under these leases on our balance sheet. Due to the fact that sublease receipts related to these four aircraft are insufficient to cover our lease obligations, we have recognized an onerous contract accrual on our balance sheet which is equal to the difference between the present value of the lease expenses and the present value of the sublease income discounted at appropriate discount rates. This accounting treatment, however, does not result in the same presentation as if we accounted for these aircraft as owned assets and the related operating lease obligations as debt liabilities.

We continue to have an economic interest in AerCo. This interest is not assigned any value on our balance sheet because we do not expect to realize any value for our investment. We also have other investments in companies or ventures in the airline industry which we obtain primarily through restructurings in our leasing business. The value of these investments is immaterial to our financial position.

We have entered into a joint venture, AerDragon, which does not qualify for consolidated accounting treatment. The assets and liabilities of this joint venture are off our balance sheet and we only record our net investment under the equity method of accounting.

## INDEBTEDNESS

A detailed summary of the principal terms of our indebtedness can be found in our 2008 Annual Report on Form 20-F filed with the SEC on April 1, 2009. Following is a summary of changes to our indebtedness since our 20-F filing:

## AerCap Ireland A330 Pre-delivery Payment Facilities

*General.* In December 2006, we signed a purchase agreement to purchase up to 20 Airbus A330 aircraft. In May 2007, the purchase agreement was amended to add ten additional aircraft. As of June 30, 2009, five of the aircraft had been delivered and the remaining are scheduled to be delivered prior to the end of 2012.

Under the purchase agreement, we agreed to make scheduled pre-delivery payments to Airbus prior to the physical delivery of each aircraft. In connection with the scheduled delivery of one aircraft in September 2010, AerCap Ireland Limited entered into a facility in March 2009 with HSH Nordbank AG to finance up to 70% of the pre-delivery payments for this aircraft.

In connection with the scheduled delivery of another ten aircraft between October 2009 and May 2012, AerCap Ireland Limited entered into a facility in June 2009 with HSH Nordbank AG to finance up to the EUR equivalent of \$221.2 million of the pre-delivery payments to Airbus.

Prior to drawing on the facilities, we will pay, on average, 23% of the pre-delivery payment amount owed for each aircraft. In each case, part of the facility will be used to refinance each of the pre-delivery payments already made by AerCap Ireland Limited in respect of the relevant aircraft and the rest will be used to pay to Airbus the remaining pre-delivery payments in respect of the relevant aircraft as such fall due and payable. AerCap Ireland Limited must repay the lenders for the amounts drawn for the pre-delivery payment for each aircraft at the delivery date of that aircraft or, if the aircraft is not delivered on the scheduled delivery date, within three months of the scheduled delivery date.

*Interest Rate.* Borrowings under the first facility will bear interest at a floating interest rate of one-month LIBOR plus a margin of 4% per annum. Borrowings under the second facility will bear interest at a floating interest rate of one-month EURIBOR plus a margin of 5.5% per annum. Interest under both facilities is payable monthly in arrears after the initial drawing on the respective facility.

**Prepayment.** Borrowings under the facilities may be prepaid (subject to minimum amounts of \$1.0 million for the first facility and EUR1.0 million for the second facility or integral multiples thereof and subject to AerCap Ireland Limited giving the agent at least 10 Business Days' notice) without penalty, except for break funding costs if payment is made on a day other than an interest payment date.

*Maturity Date.* The maturity date of the facilities will be the earlier of (a) the delivery date for the final aircraft to be delivered and (b), in respect of the first facility, December 31, 2010, or, in respect of the second facility, November 1, 2012.

**Collateral.** Borrowings under the facilities are secured by, among other things, the partial assignment of the airframe and engine purchase agreements in respect of the 11 aircraft covered by the facilities, including the right to take delivery of the aircraft where the lenders have provided the pre-delivery payments and the aircraft remains undelivered.

*Guarantee.* AerCap Holdings N.V. guarantees the performance by AerCap Ireland Limited of its obligations under the facilities and related documentation.

*Certain Covenants.* The facilities contain customary covenants for secured financings. We are required to pay for all buyer furnished equipment, or BFEs, prior to the delivery date.

*Certain conditions.* A condition to the continued availability of the loan facility disbursed under the Facility Agreement dated 30 June 2009 between AerCap as Borrower and HSH as Lender, Agent and Security Trustee relating to pre-delivery payments in respect of ten (10) Airbus A330 aircraft (the "Facility") that the Borrower or an affiliate would acquire one Boeing 747-400 ERF aircraft or other assets acceptable to HSH on or before September 1, 2009. If this condition is not fulfilled the Final Repayment Date for the Facility will be 28 February 2010 and AerCap will be obliged to repay all amounts outstanding on that date. However, the delivery of the aircraft is now expected beyond September 1, 2009 and the date to meet this condition is expected to be modified in accordance with the aircraft delivery date.

## Item 3. Quantitative and Qualitative Disclosures About Market Risk

Our primary market risk exposure is interest rate risk associated with short and long-term borrowings bearing variable interest rates and lease payments under leases tied to floating interest rates. To manage this interest rate exposure, we enter into interest rate swap and cap agreements. We are also exposed to foreign currency risk, which can adversely affect our operating profits. To manage this risk, we enter into forward exchange derivatives.

The following discussion should be read in conjunction with our audited consolidated financial statements as filed with the SEC on April 1, 2009, which provide further information on our derivative instruments.

## Interest Rate Risk

The rentals we receive under our leases are based on fixed and variable interest rates. We fund our operations with a mixture of fixed and floating rate US dollar denominated debt and finance lease obligations. An interest rate exposure arises to the extent that the mix of these obligations is not matched with our assets. This exposure is primarily managed through the use of interest rate caps using a cash flow based risk management model. This model takes the expected cash flows generated by our assets and liabilities and then calculates how much the value of these cash flows will change by for a given movement in interest rates.

The table below provides information as of June 30, 2009 regarding our debt and finance lease obligation and their related interest rate exposure:

	2009 (07/01/2009- 12/31/2009)	2010 (US do	2011 llars in thousands)	2012	2013
Average fixed rate debt outstanding	383,217	304,704	228,386	205,012	183,852
Average floating rate debt outstanding	3,768,269	3,241,778	2,963,766	2,101,236	1,544,187
Fixed rate interest obligations	14,772	23,645	17,946	16,799	15,763
Floating rate interest obligations (1)	26,203	45,073	37,416	28,695	20,678

(1) Based on one-month LIBOR and three-month LIBOR as of June 30, 2009, which were 0.309% and 0.595%, respectively.

Under our interest rate caps, we will receive the excess, if any, of LIBOR, reset monthly or quarterly on an actual/360 adjusted basis, over the strike rate of the relevant cap. The caps amortize based on a number of factors, including

the expiration dates of the leases under which our lessees are contracted to make fixed rate rental payments and the six- or six-month LIBOR reset dates under our floating rate leases. Under our interest rate floors, we pay for the difference when the LIBOR rate, reset monthly or quarterly on an actual/360 adjusted basis, falls below the strike rate of the relevant floor.

The table below provides information as of June 30, 2009 regarding our derivative financial instruments that are sensitive to changes in interest rates on our borrowing, including our interest rate caps and floors. The table presents the average notional amounts and weighted average strike rates relating the interest rate caps and floors for the specified year. Notional amounts are used to calculate the contractual payments to be exchanged under the contract.

	 2009		2010		2011	_	2012 (US Dollar	_	2013 nillions)	_	2014		Thereafter	]	Fair value
Interest rate caps							(1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.								
Average notional amounts	\$ 2,900	\$	2,248	\$	1,890	\$	1,381	\$	986	\$	708	5	5 1,008	\$	37.9
Weighted average strike rate	3.94 %	6	4.05 %	6	4.10%	ó	4.68 %	6	4.90	%	4.99	%	5.28	%	
	 2009		2010		2011		2012 (US Dollar	_	013 nillions)	2	014	_1	`hereafter	Fa	iir value
Interest rate floors															
Notional amounts	\$ 183	\$	166	\$	141	\$	107	\$	70	\$	45	\$	27	\$	(8.1)
Weighted average strike rate															

As of June 30, 2009, the interest rate caps and floors had notional amounts of \$2.98 billion and a fair value of \$29.8 million. The variable benchmark interest rates associated with these instruments ranged from one- to six-month LIBOR.

Our Board of Directors is responsible for reviewing and approving our overall interest rate management policies and transaction authority limits. Specific hedging contracts are approved by the Treasury Committee acting within the overall policies and limits. Our counterparty risk is monitored on an ongoing basis, but is mitigated by the fact that the majority of our interest rate derivative counterparties are required to cash collateralize in the event of their downgrade by the rating agencies below a certain level. Our counterparties are subject to the prior approval of the Treasury Committee.

# Foreign Currency Risk and Foreign Operations

Our functional currency is the US dollar. As of June 30, 2009, all of our aircraft leases and all of our engine leases were payable in US dollars. We incur Euro-denominated expenses in connection with our offices in The Netherlands and Ireland. For the six months ended June 30, 2009, our aggregate expenses denominated in currencies other than the US dollar, such as payroll and office costs and professional advisory costs, were \$23.2 million in US dollar equivalents and represented 42.2% of total selling, general and administrative expenses. We enter into foreign exchange derivatives based on our projected exposure to foreign currency risks in order to protect ourselves from the effect of period over period exchange rate fluctuations. Mark-to-market gains or losses on such derivatives are recorded as part of selling, general and administrative expenses. We do not believe that a change in foreign exchange rates will have a material impact on our results of operations. However, the portion of our business conducted in foreign currencies could increase in the future, which could increase our exposure to losses arising from currency fluctuations.

# PART II OTHER INFORMATION

# **Legal Proceedings**

There have been no material changes to legal proceedings described in our Annual Report on Form 20-F, filed with the SEC on April 1, 2009.

# **Item 1. Risk Factors**

There have been no material changes to the disclosure related to the risk factors described in our Annual Report on Form 20-F, filed with the SEC on April 1, 2009.

# Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

# Item 3. Defaults upon Senior Securities

None.

# Item 4. Submission of Matters to a Vote of Security Holders

None.

# Item 5. Other Information

None.

# Item 6. Exhibits

None

Dated 21 June 2009

AERCAP AERVENTURE HOLDING B.V. (as a shareholder)

WAHA AV PARTICIPATIONS B.V. (as a shareholder)

AERVENTURE LIMITED (as the company)

SHAREHOLDERS' AGREEMENT relating to shares in

# **AERVENTURE LIMITED**

# McCann FitzGerald

Solicitors Riverside One Sir John Rogerson's Quay Dublin 2 RTM\1959497.18

# CONTENTS

CLA	USE	PAGE		
1.	INTERPRETATION	1		
2.	PURPOSE OF THE COMPANY, WARRANTIES AND CONDITIONALITY	9		
3.	COMPLETION	11		
4.	CAPITAL; FURTHER FINANCE; DIVIDEND POLICY	20		
5.	DIRECTORS AND MANAGEMENT	25		
6.	RESERVED MATTERS	27		
7.	INFORMATION AND REPORTING	31		
8.	CONFIDENTIALITY	32		
9.	REGULATORY MATTERS	34		
10.	RELATIONSHIP OF THE COMPANY WITH SHAREHOLDERS	35		
11.	TAX MATTERS	35		
12.	TRANSFER OF SHARES	35		
13.	TERMINATION EVENTS	42		
14.	DEFAULT	43		
15.	TRANSFER TERMS	43		
16.	ASSURANCES	44		
17.	NON-ASSIGNMENT	44		
18.	WAIVER OF RIGHTS	44		
19.	AMENDMENTS	44		
20.	INVALIDITY	45		
21.	NO PARTNERSHIP OR AGENCY	45		
22.	ANNOUNCEMENTS	45		
23.	ENTIRE AGREEMENT	45		
24.	CONFLICT WITH MEMORANDUM AND ARTICLES	46		
25.	DURATION	46		
26.	NOTICES	46		
27.	DEADLOCK	48		
28.	COUNTERPARTS	51		
29.	EXPENSES	51		
30.	RIGHTS OF THIRD PARTIES ACT	51		
31.	GOVERNING LAW; SUBMISSION TO JURISDICTION	51		
	IEDULE 1 - FORM OF DEED OF ADHERENCE	53		
	IEDULE 2 - LOAN NOTE INSTRUMENT	55		
SCH	IEDULE 3 - 15% LOAN NOTE INSTRUMENT	72		

# THIS AGREEMENT is made on 21 June 2009

# **BETWEEN:**

- (1) **AERCAP AERVENTURE HOLDING B.V.**, a company organised and existing under the laws of the Netherlands, whose principal place of business is Stationsplein 965, 1117 CE Schiphol Airport, Amsterdam, The Netherlands ("AerCap");
- (2) **WAHA AV PARTICIPATIONS B.V.**, a company organised and existing under the laws of the Netherlands, whose principal place of business is Parkstraat 107, 2514 JH, The Hague, The Netherlands, ("**Waha**"); and
- (3) **AERVENTURE LIMITED**, a company incorporated in Ireland (registered no. 410443) whose registered office is at AerCap House, Shannon, Co. Clare (the "**Company**").

# WHEREAS:

- (A) The Company was incorporated on 7 November 2005 under the Companies Acts 1963 to 2005 and is a private company limited by shares.
- (B) The Company has an authorised share capital of US\$400,000,000 divided into 318,871,144 ordinary shares of US\$1.00 and 81,128,856 redeemable ordinary shares of US\$1.00 each of which 59,128,856 ordinary shares and 51,128,856 redeemable ordinary shares have been issued and are outstanding, all of which are held by AerCap.
- (C) Waha has agreed to subscribe for 59,128,856 Shares at an aggregate subscription price of US\$59,128,856 and to make a capital contribution of US\$4,971,144 to the Company. The proceeds of the subscription and capital contribution shall be used by the Company to redeem all of the redeemable ordinary shares of US\$1.00 each in the capital of the Company held by AerCap for an aggregate redemption price of US\$56,100,000 and to repay certain loans made by AerCap to the Company which are outstanding. In addition, both AerCap and Waha have agreed to subscribe for further Shares at Completion and to make the Required Equity Subscriptions, in each case on the terms set out in this Agreement. Following Completion, the Company shall be a joint venture vehicle between AerCap and Waha for the purpose of the acquisition, leasing and trading of a fleet of new Airbus aircraft as described in the Airbus Purchase Agreement.
- (D) Waha and AerCap are entering into this Agreement in order to achieve the purpose of the Company specified in Recital C and to set out the terms upon which AerCap and Waha have agreed to invest in the Company and provisions governing the operation of the Company.

#### IT IS AGREED as follows:

# 1. **INTERPRETATION**

# 1.1 **Definitions**

In this Agreement:

"15% Loan Note Instrument" means an instrument of the Company constituting redeemable loan notes bearing interest at a rate of 15% per annum,



in the form contained in Schedule 3, subject to any amendments that may be agreed by the Board;

"2010 First Equity Subscription" means the subscription by AerCap and Waha for Shares pursuant to Clause 3.5(b);

"2010 Second Equity Subscription" means the subscription by AerCap and Waha for Shares pursuant to Clause 3.5(c);

"Adjusted Fair Value" means, in respect of any Shares, the Fair Value of such Shares discounted by 10%;

"Administrative Agent" means AerCap Administrative Services Limited;

"Administrative Agency Agreement" means the administrative agency agreement entered into on 13 January 2006 (as amended from time to time) between the Company and the Administrative Agent, pursuant to which the Administrative Agent agreed to provide certain administrative services to the Company and other Group Members which have acceded thereto;

"AerCap Directors" means the Company's directors appointed by the AerCap Shareholder from time to time in accordance with the terms of this Agreement;

"AerCap Group" means AerCap Holdings N.V. and its Related Companies;

"AerCap Share Charge" means the share charge in respect of the Shares held by AerCap in favour of Waha Capital PJSC dated on the date of this Agreement;

"AerCap Shareholder" means, collectively, AerCap (for so long as it holds Shares) and such other member of the AerCap Group (if any) which may subscribe for shares or to which AerCap shall transfer Shares from time to time, in each case in accordance with the terms of this Agreement, provided that such other member of the AerCap Group shall have executed a Deed of Adherence at the time of such subscription or transfer;

"AerCap Shares" means those Shares beneficially owned by the AerCap Shareholder from time to time;

"Aeroflot Aircraft Acquisition Agreement" means the share purchase agreement between Waha Capital PJSC and Flotlease MSN 3699 Limited in respect of the Aeroflot Aircraft dated on the date of this Agreement;

"Aeroflot Aircraft" means the Airbus A320-200 aircraft with manufacturer's serial number 3699 together with two CFM56-5B4/P engines having manufacturer's serial numbers 699117 and 699118 including all buyer furnished equipment, galley equipment, emergency equipment, manuals, technical records and aircraft documents relating to such aircraft;

"Agreed Proportion" means in respect of a particular Shareholder, the percentage which the nominal value of the Shares beneficially owned by that Shareholder at the relevant time bears to the combined nominal value of all issued Shares at that time;

"Airbus Purchase Agreement" means the aircraft purchase agreement entered into between the Company and Airbus S.A.S. on 30 December 2005;

"Aircraft" means the aircraft already delivered and to be delivered to the Company or any other Group Member pursuant to the terms of the Airbus Purchase Agreement;

"Approved Budget" means the operating budget of the Company in respect of each Financial Year;

"Articles" means the Articles of Association of the Company;

"Board" means the Company's board of directors or any duly appointed committee of it;

"Business" means the business intended to be carried on by the Company, as described in Clause 2.1;

"**Business Day**" means a day (other than a Saturday) on which banks generally are open in the Netherlands, Ireland and Abu Dhabi for a full range of business;

"Cash Management Agreement" means the cash management agreement entered on 13 January 2006 (as amended from time to time) between the Company and the Cash Manager pursuant to which the Cash Manager agreed to provide certain cash management services to the Company and the other Group Members which have acceded thereto;

"Cash Manager" means AerCap Cash Manager II Limited;

"company" includes any body corporate, wherever incorporated;

"Completion" means completion of the matters referred to in Clause 3.4;

"**connected with**" in relation to two or more persons, means two or more persons who are connected with each other for the purposes of section 10 of the Taxes Consolidation Act 1997 and a "**Connected Person**" of any person means a person who is connected with that first-mentioned person;

"Deed of Adherence" means a deed of adherence to this Agreement in substantially the form set out in Schedule 1;

"**Deed of Warranty**" means the deed of warranty between Waha, Waha Capital PJSC, AerCap Holdings N.V. and the Company dated on the date of this Agreement;

"Default Termination Notice" has the meaning given to it in Clause 14.2(a);

"Defaulting Party" has the meaning given in Clause 3.6(a);

"Delivered Waha A320-200 Aircraft" has the meaning given in Clause 3.2(b);

"Directors" means the Company's directors;

"**Encumbrance**" means any mortgage, charge (fixed or floating), pledge, lien, option, right to acquire, right of pre-emption, assignment by way of security or trust arrangement for the purpose of providing security or other security interest of any kind (including any retention arrangement), or any agreement to create any of the foregoing;

"**Fair Value**" means, in respect of one or more Shares, the fair market value of such Shares as determined by the Valuer, calculated as a pro rata proportion of the market value of all of the issued Shares of the Company taken as a whole, between a willing seller and a willing third party buyer at the date of the Transfer Notice without any premium or discount by reference to the number of the Shares being sold or transferred;

"Financial Year" means a financial period of the Company commencing on 1 January and ending on 31 December;

"Funding Support Certificate" means a certificate from the Cash Manager to the Company which:

- (a) states that, in the reasonable opinion of the Cash Manager and on the basis of the figures specified therein, the Company will not have sufficient available cash to meet its obligations for the next succeeding three month period; and
- (b) contains reasonably sufficient detail to evidence the matters of fact contained therein;

"Governmental Entity" means any governmental, judicial, legislative, executed, administrative or regulatory authority of the United Kingdom (or any part thereof), Ireland, the Netherlands, the European Union or any other federal, supranational, national, state, provincial or local government or body;

"Group" means the Company and its Subsidiaries and "Group Member" means any of them;

"holding company" shall have the meaning given to it in section 155 of the Companies Act 1963 (as amended);

"Loan Note Instrument" means the instrument of the Company constituting convertible redeemable loan notes with an aggregate nominal value of US\$36,500,000 and which loan notes shall bear interest at a rate of 20% per annum, in the form contained in Schedule 2;

"Loan Notes" means loan notes issued by the Company to a Non Defaulting Party pursuant to the Loan Note Instrument;

"Material Default" has the same meaning given to that expression in Clause 14.2;

"Memorandum and Articles" means the Company's Memorandum and Articles of Association in the agreed form, as amended from time to time;

"Non Defaulting Party" has the meaning given in Clause 3.6(a);

"**Parties**" means the Company, Waha and AerCap and any other person who enters into a Deed of Adherence in accordance with the terms of this Agreement (and "**Party**" shall be construed accordingly);

"Regulatory Action" means:

(a) any order of a court of competent jurisdiction; or

- (b) any order, decision or conclusive view made, given or expressed by a competent Governmental Entity; or
- (c) an enactment of a legislative body which:
  - (i) materially prohibits or restricts the carrying on of the Business; or
  - (ii) in consequence of which, any of the Parties would incur fines or a liability in damages were this Agreement to be performed in accordance with its terms;

"**Related Company**" means, in relation to any company, any other company which is for the time being its Subsidiary or holding company or a Subsidiary of its holding company;

"**Remarketing Agent**" means, in respect of certain Aircraft, AerCap Ireland Limited and, in respect of other Aircraft, AerCap Group Services B.V.;

"**Required Equity Subscriptions**" means the 2010 First Equity Subscription and the 2010 Second Equity Subscription and "**Required Equity Subscription**" means any one of them;

"Reserved Shareholders Matters" means those matters listed in Clause 6.2 (Reserved Shareholder Matters);

"Sale Shares" has the meaning give in Clause 12.3;

"Servicing Agreements" is the collective term for the Services and Remarketing Agreement, the Administrative Agency Agreement and the Cash Management Agreement and "Servicing Agreement" means any of them;

"Servicing Agreement Amendment Agreement" means the Servicing Agreement Amendment Agreement dated on the date of this Agreement between each Remarketing Agent (in its capacity as primary servicer), the Cash Manager (in its separate capacities as insurance servicer and cash manager) the Administrative Agent (in its capacity as administrative agent) and the Company, pursuant to which the Services and Remarketing Agreement is to be amended;

"Services and Remarketing Agreement" means the Amended and Restated Servicing Agreement dated 2 November 2007 between each Remarketing Agent (in its capacity as primary servicer), the Cash Manager (in its separate capacities as insurance servicer and cash manager) the Administrative Agent (in its capacity as administrative agent) and the Company, under which the Remarketing Agent agreed to provide certain services relating to the Aircraft to the Company and the other Group Members which have acceded thereto, as amended by the Servicing Agreement Amendment Agreement;

"Services Delegation and Co-operation Agreement" means the services delegation and co-operation agreement dated on the date of this Agreement between inter alia the Servicer and Waha Capital PJSC pursuant to which the Servicer will sub-delegate to Waha Capital PJSC the performance of certain of the services which the Remarketing Agent provides to the Group, Aircraft Lease Securitisation II Limited and its Subsidiaries and AerFunding I Limited and its Subsidiaries;

"Servicer" means (i) the Remarketing Agent, (ii) the Administrative Agent and (iii) the Cash Manager;

"Shareholders" means all of the beneficial owners of Shares excluding any person who has acquired any beneficial ownership otherwise than in accordance with this Agreement (and "Shareholder" shall be construed accordingly);

"Shareholder Interest" means, at any time and in respect of any Shareholder the aggregate nominal value of the Shares in issue at such time which are beneficially owned by such Shareholder;

"Shares" means the ordinary shares in the Company's capital;

"Significant Shareholder" means, at any time, a Shareholder which is at such time the beneficial owner of 25% or more of the issued Shares;

"Subsidiary" means in relation to any company or corporation, a company or corporation:

- (a) which is controlled, directly or indirectly by the first mentioned company or corporation;
- (b) more than half the issued share capital of which is beneficially owned, directly or indirectly by the first mentioned company or corporation;
- (c) which is a Subsidiary of another Subsidiary of the first mentioned company or corporation;

"Supplementary Disclosure Letter" has the meaning given in the Deed of Warranty;

"Tag-Along Offer" has the meaning given in Clause 12.9;

"**Tax**" means all forms of taxation including income, withholding, corporation, capital gains, inheritance, value added, employment, property, sales, goods, wealth, customs and other import or export duties, excise duties, stamp duties, transfer taxes, social security or other similar contributions and all charges, duties, imposts and levies of a similar nature, and any interest, penalty, surcharge or fine relating to such taxation;

"Tax Authority" means any authority competent to impose, administer or collect any Tax;

"**Termination Event**" means in respect of a Shareholder the occurrence of any of the following events in respect of such Shareholder or any of its holding companies (including, for the avoidance of doubt, in the case of Waha, Waha Capital PJSC and in the case of AerCap, AerCap Holdings N.V.):

- (a) a court of competent jurisdiction makes an order or a resolution is passed, for the dissolution, examination or administration of such Shareholder or any holding company of it (otherwise than in the course of a solvent reorganisation or restructuring previously approved by each Significant Shareholder); or
- (b) any person takes any step (which is not withdrawn or discharged within thirty (30) days) to appoint a liquidator, manager, receiver, administrator, examiner, administrative receiver or other similar officer in respect of it or any of its assets, which include either:
  - (i) the Shares held by it; or

- (ii) shares in it or any holding company of it; or
- (c) it (or any holding company of it) convenes a meeting of its creditors or makes or proposes any arrangement or composition with, or any assignment for the benefit of, its creditors or any of them; or
- (d) it commits a Material Default which any Significant Shareholder treats as a Termination Event in accordance with Clause 14.2;

"Third Party Purchaser" has the meaning given in Clause 12.3;

"UBS Facility" means the Amended and Restated Credit Agreement dated as of 8 May 2007 between AerFunding 1 Limited as borrower, AerCap Ireland Limited individually and as primary servicer, AerCap Administrative Services Limited individually and as primary administrative agent, AerCap Cash Manager II Limited individually and as financial administrative agent, cash manager and insurance servicer, the lenders named therein as Class A Lenders, Class B Lenders, UBS Securities LLC as Administrative Agent and UBS Funding Agent, the other funding agents named therein as Other Funding Agents and Deutsche Bank Trust Company Americas as Collateral Agent and Account Bank;

"Undelivered Waha A320-200 Aircraft" has the meaning given in Clause 3.2(b);

"US\$" and "dollar" means the lawful currency of the United States of America;

"Valuer" means:

- (a) Lazard & Co Limited; or
- (b) if Lazard & Co Limited are unwilling or unable to act as Valuer for the purposes of this Agreement, NM Rothschild & Sons Limited; or
- (c) if NM Rothschild & Sons Limited are unwilling or unable to act as Valuer for the purposes of this Agreement, Goldman, Sachs & Co; or
- (d) if Goldman, Sachs & Co. are unwilling or unable to act as Valuer for the purposes of this Agreement, such other independent investment bank or corporate finance firm that may be agreed between the Significant Shareholders (acting reasonably);

"Waha Capital PJSC" means Waha Capital PJSC, a company organised and existing under the laws of the United Arab Emirates, whose principal place of business is Aseel Building, Six Towers, 4th Floor, Al Bateen, Abu Dhabi, United Arab Emirates;

"Waha Directors" means the Company's directors appointed by the Waha Shareholder from time to time in accordance with the terms of this Agreement;

"Waha Group" means Waha and its Related Companies;

"Waha Loan Agreement" means the Loan Agreement in relation to four Airbus A320-200 Aircraft between AerCap Holdings N.V. (as Borrower) Waha Capital PJSC (as Lender) dated on the date of this Agreement;

"Waha Shareholder" means, collectively, Waha (for so long as it holds Shares) and such other member of the Waha Group (if any) which may subscribe for

Shares or to which Waha may transfer shares from time to time, in each case in accordance with the terms of this Agreement, provided that such other member of the Waha Group shall have executed a Deed of Adherence at the time of such subscription or transfer; and

"Waha Shares" means those Shares beneficially owned by the Waha Shareholder from time to time.

### 1.2 Currency

Any reference in this Agreement to an amount in dollars includes the equivalent amount at the relevant time in any other currency or combination of currencies.

# 1.3 Agreed Form

A reference to a document in this Agreement in the agreed form is to a document agreed between the parties and initialled by them or on their behalf for identification purposes.

# 1.4 Exercise of powers of control

Where any obligation in this Agreement is expressed to be undertaken or assumed by any party, that obligation is to be construed as requiring the party concerned to exercise all rights and powers of control over the affairs of any other person which that party is able to exercise (whether directly or indirectly) in order to secure performance of the obligation.

# 1.5 Interpretation

In this Agreement, unless the context requires otherwise:

- (a) all references to Clauses or a Schedule refer to a Clause of, or the Schedule to, this Agreement and such Schedule, is deemed incorporated herein by reference;
- (b) the headings in this Agreement do not affect its interpretation;
- (c) references to any English legal term for any action, remedy, method of judicial proceeding, legal document, legal status, court, official or any other legal concept shall, in respect of any jurisdiction other than England, be deemed to include the legal concept which most nearly approximates in that jurisdiction to the English legal term;
- (d) words of the masculine, feminine or neuter gender shall mean and include the correlative words of other genders, and words in the singular shall include the plural, and vice versa;
- (e) references in this Agreement to an agreement or other document (including this Agreement) include references to such agreement or document as amended, replaced, restated or otherwise modified (without, however, limiting the effect of the provisions of this Agreement with regard to any such amendment, replacement, restatement or modification), and the provisions of this Agreement apply to successive events and transactions;
- (f) references in this Agreement to any statute or other legislative provision shall include any statutory or legislative modification or re-enactment



thereof, or any substitution therefor, and references to any Governmental Entity shall include reference to any Governmental Entity succeeding to the relevant functions of such person;

- (g) the rule of interpretation known as the *ejusdem generis* rule shall not apply to the construction of this Agreement. The terms "**include**", "**including**" and similar terms shall be construed as if followed by the phrase "**without limitation**";
- (h) any reference to a "**person**" shall be construed as a reference any individual, firm, company, corporation, undertaking, government, Governmental Entity, state or agency of a state, or any association or partnership (whether or not have separate legal personality);
- (i) a reference to a person (including a party to this Agreement) includes a reference to that person's legal personal representatives, successors and permitted assigns; and
- (j) a reference to "**other Shareholders**" or "**other Significant Shareholders**" or any of them shall include a reference to "**the Other Shareholder**" and "**the other Significant Shareholder**" respectively, if there shall be only two Shareholders or two Significant Shareholders (as the case may be) at the relevant time.

# 2. PURPOSE OF THE COMPANY, WARRANTIES AND CONDITIONALITY

#### 2.1 Business

- (a) The primary object of the Company shall be to carry on the business of acquiring, leasing, selling or otherwise disposing of the Aircraft and activities ancillary thereto in accordance with the Approved Budget (the "**Business**").
- (b) The proceeds of the subscriptions provided for in Clauses 3.4(f) and 3.4(h) and the Required Equity Subscriptions shall be used for the purpose of funding the requirements to purchase Aircraft that have yet to be delivered pursuant to the Airbus Purchase Agreement and providing working capital for the Group.

# 2.2 Commercial principles and Tax Residency

- (a) The Business shall be conducted in the best interests of the Company in accordance with the general principles of the then current Approved Budget (as such term is defined in the Services and Remarketing Agreement) and on sound commercial profit-making principles with the aim of generating the maximum achievable maintainable profits available for distribution.
- (b) The central management and control of the Company shall be exercised in the Netherlands and each of the Shareholders shall use its best endeavours to ensure that the Company is treated by all relevant authorities as being resident for taxation and other purposes in the Netherlands. Meetings of the Board will take place and be conducted in accordance with Clause 5 (*Directors and Management*).

# 2.3 Warranties

Each Party warrants to the other Party at Completion that:

- (a) it is validly incorporated with limited liability and is duly incorporated or organised and validly existing under the applicable laws of its jurisdiction of incorporation or organisation and has the power and all necessary other consents, approvals, licences and authorisations from all relevant Governmental Entities in any applicable jurisdiction to own its material assets and carry on its business substantially as it is conducted on the date of this Agreement;
- (b) it has full power and authority to enter into and perform this Agreement and any other agreements referred to in this Agreement to which it is a party and no limits on its powers will be exceeded as a result of the taking of any action contemplated by any such agreement;
- (c) all actions, conditions and things required to be taken, fulfilled and done (including the obtaining of any necessary authorisations, consents, orders, licences and approvals imposed by its constitutional documents, any Governmental Entity or otherwise), in order to enable it lawfully to enter into, exercise its rights and perform and comply with its obligations contained in this Agreement and any other agreements referred to in this Agreement to which it is a party have been so taken, fulfilled, obtained or done and the requisite resolutions of its board of directors (and, if necessary, its shareholder(s)) have been duly and properly passed at a duly convened and constituted meeting at which all statutory and other relevant formalities were observed to authorise its execution and performance of this Agreement and any other agreements referred to in this Agreement to which it is a party and such resolutions are in full force and effect and have not been varied or rescinded;
- (d) when executed, this Agreement and any other agreements referred to in this Agreement to which it is a party, will constitute legal, valid and binding obligations on it in accordance with their terms;
- (e) neither the execution nor the delivery of this Agreement and any other agreements referred to in this Agreement to which it is a party, nor the carrying out of any transaction or the exercise of any rights or the performance of any obligations contemplated by this Agreement and any other agreements referred to in this Agreement to which it is a party will result in:
  - (i) violation of any law to which it is subject;
  - (ii) any breach of any of its constitutional documents;
  - (iii) any breach of any deed, agreement, instrument or obligation made with or owed to any other person; or
  - (iv) any breach of any order, judgment or decree of any Court or Governmental Entity to which it is a party or by which it is bound; and
- (f) it is not involved in or engaged in any litigation, arbitration or other legal proceedings of a litigious nature (whether as plaintiff, claimant or defendant and whether civil, criminal or administrative) which is likely to

be adversely determined and, if adversely determine, would have an adverse effect on its ability to perform its obligations under this Agreement and any other agreements referred to in this Agreement to which it is a party.

# 2.4 Conditionality

Except for Clauses 1, 2.4, 3, 8, 9, 17 to 23 (inclusive) and 28 to 31 (inclusive) this Agreement is conditional upon all of the conditions to Completion as set out in Clause 3.2 (the "**Conditions**") having been satisfied or waived by the Party or Parties entitled to do so pursuant to Clause 3.1(b).

# 3. COMPLETION

# 3.1 **Completion Date**

- (a) Completion shall take place at the offices of AerCap House, Stationsplein 965, 1117 CE Schiphol Airport, The Netherlands at 3 p.m. (Amsterdam time) on 25 June 2009 (or at such other place or date as the Shareholders agree in writing) (the "Target Completion Date") or, if all of the Conditions have not been fulfilled or waived by the Party or Parties entitled to do so by the Target Completion Date, subject to Clause 3.1(d), at the same time and place on the first date on which the Conditions are satisfied or so waived;
- (b) the Conditions shall be capable of being waived as follows:
  - (i) each of AerCap and Waha shall be entitled to waive compliance with the Condition in Clause
    3.2(a) insofar as it requires that the warranties given by the other such Party are true and correct as of the date of Completion; and
  - (ii) Waha shall be entitled to waive compliance with the Conditions in Clauses 3.2(b) and (c);
- (c) each of the Parties shall use its respective reasonable endeavours, insofar as it is within its capacity to do so, to ensure that the Conditions are satisfied as soon as is reasonably practicable but in any event not later than the Long Stop Date;
- (d) if all of the Conditions have not been satisfied or waived by the Party or Parties entitled to do so pursuant to Clause 3.1(b) on or before 30 June 2009 (the "Long Stop Date") then this Agreement shall terminate immediately; and
- (e) if this Agreement terminates pursuant to Clause 3.1(d) each Party shall be released from its obligations under this Agreement (other than Clauses 8 and 22) without any further action of the parties and neither party shall have any claim or liability to the other, other than in respect of any breach of this Clause 3.1 and Clauses 8 and 22.

#### 3.2 Conditions Precedent to Completion

The following are the Conditions:

(a) the warranties of each Party shall be true and correct as of the date of Completion;

- (b) the relevant member of the AerCap Group having executed a counterpart of the relevant Waha A320-200 Acquisition Agreement in respect of:
  - (i) each Delivered Waha A320-200 Aircraft; and
  - (ii) each Undelivered Waha A320-200 Aircraft;

For the purposes of this Clause 3.2(b):

"Delivered Waha A320-200 Aircraft" means each of the Waha A320-200 Aircraft (if any) which, within the month of June 2009, has been delivered to, and situated at, Goodyear, Arizona for not less than 72 hours;

"**Undelivered Waha A320-200 Aircraft**" means each of the Waha A320-200 Aircraft (if any) that is not a Delivered Waha A320-200 Aircraft;

"Waha A320-200 Acquisition Agreement" means the four aircraft sale agreements in the agreed form to be entered into between AerCap Holdings N.V. and the relevant member(s) of the Waha Group in respect of the Waha A320-200 Aircraft;

"Waha A320-200 Aircraft" means, collectively:

- (i) the Airbus A320-200 aircraft with manufacturer's serial number 313 together with two CFM56-5A3 engines bearing manufacturer's serial numbers 731645 and 731674;
- (ii) the Airbus A320-200 aircraft with manufacturer's serial number 363 together with two V2500-A1 engines bearing manufacturer's serial numbers V0254 and V0256;
- (iii) the Airbus A320-200 aircraft with manufacturer's serial number 438 together with two CFM56-5A3 engines bearing manufacturer's serial numbers 731796 and 731789; and
- (iv) the Airbus A320-200 aircraft with manufacturer's serial number 445 together with two CFM56-5A3 engines bearing manufacturer's serial numbers 731791 and 731793,

in each case including all buyer furnished equipments, galley equipment (excluding loose galley equipment), emergency equipment, manuals, technical records and aircraft documents relating thereto; and

- (c) the Supplementary Disclosure Letter not disclosing:
  - (i) any matter which constitutes a breach of any of the Warranties in paragraph 1.1, 1.2 or 1.4 of schedule 2 to the Deed of Warranty; or
  - (ii) any matter which has had, or is reasonably likely to have, a material adverse effect on the Business or the Company. For this purpose, a matter shall be deemed to have a material adverse affect if and only if it would (if not disclosed in the Supplementary Disclosure Letter and if Completion takes place), entitle the Beneficiaries to bring a Relevant Claim in which they would be entitled to recover not less than US\$2,000,000.

12

Source: AerCap Holdings N.V., 6-K, September 01, 2009

For the purposes of this Clause 3.2(c), "Beneficiaries" and "Relevant Claim" having the meaning ascribed to them in the Deed of Warranty.

# 3.3 **Pre-Completion Covenants and Undertakings**

- (a) AerCap will use its best endeavours to procure that between the date of this Agreement and Completion:
  - (i) the Company will, upon reasonable notice, give to Waha and its advisers such access to its books of account and records as Waha may reasonably request; and
  - (ii) the Company will carry on business in the ordinary course, provided that the Company may make capital or other expenditures which are not in the ordinary course of business if they are of an emergency nature and are required to avoid imminent material damage to, or shutdown of, the business of the Company or if they are necessary for safety reasons and the Company may take any other actions which are required by law.
- (b) Without prejudice to the generality of Clause 3.3(a), AerCap will procure that, between the date of this Agreement and Completion, save as expressly contemplated by this Agreement, no member of the Group shall, without the prior consent of Waha (such consent not to be unreasonably withheld or delayed):
  - (i) create, extend, grant or issue or agree to create, extend, grant or issue any Encumbrance over its assets;
  - (ii) convert, redeem or cancel any of its own shares, acquire any of its own shares or convert any of its own shares into redeemable or treasury shares;
  - (iii) create or issue or agree to create or issue any share or subordinated loan capital or loan stock or give or agree to give any warrants, rights of pre-emption or options in respect of any share or subordinated loan capital or loan stock;
  - (iv) purchase or otherwise acquire or agree to purchase or acquire any share or subordinated loan capital or loan stock of any company;
  - (v) borrow or incur a liability in excess of US\$5,000,000;
  - (vi) propose, pay, declare or make any dividends or any other distribution to the Company's shareholders;
  - (vii) permit or cause to be proposed any alteration to its share capital (including any increase or reduction thereof) or the rights attaching to its shares;
  - (viii) cease or propose to cease to carry on its business or be wound up or enter into receivership, or any form of management or administration over its assets;
  - (ix) authorise or incur any capital expenditure in excess of US\$5,000,000 per item or US\$5,000,000 in aggregate or agree to do so;



- (x) sell, lease, assign or otherwise transfer or dispose of any asset or any interest therein or agree to do so, involving a consideration in excess of US\$5,000,000 per item or US\$5,000,000 in aggregate;
- (xi) amend, terminate or enter into or agree to amend, terminate or enter into:
  - (A) any aircraft lease agreement to which it is a party; or
  - (B) any contract with a value of US\$5,000,000 or over to which it is a party;
- (xii) amend or terminate or agree to amend or terminate the Servicing Agreements or any of them; or
- (xiii) employ or agree to employ or offer to employ any person.
- (c) Nothing in paragraph (a) or (b) of this Clause 3.3 shall operate so as to restrict the completion or performance of any obligation undertaken pursuant to any contract or arrangement entered into by the Company or any member of the AerCap Group prior to the date of this Agreement.
- (d) Without prejudice to Clause 2.4 and Clause 3.1(d) Waha irrevocably and unconditionally undertakes to AerCap and the Company that it will procure that at or prior to Completion, the relevant member of the Waha Group will execute and deliver to AerCap a counterpart of each Waha A320-200 Acquisition Agreement.

# 3.4 **Completion**

At Completion all, but not some only, of the following shall take place (to the extent that they have not taken place prior to Completion):

- (a) the passing of the special resolutions of the Company to approve the adoption of the Memorandum and Articles;
- (b) a meeting of the Board at which the decisions and actions of the Company as described in this Clause 3.4 are approved and any other agreements contemplated by this Agreement to which the Company is a party are approved;
- (c) Waha shall:
  - (i) apply for the issue to it of 59,128,856 Shares and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as Waha shall have been notified by or on behalf of the Company prior to Completion, an aggregate subscription price of US\$59,128,856 in respect of the Shares specified in this Clause 3.4(c); and
  - (ii) make a capital contribution of US\$4,971,144 to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as Waha shall have been notified by or on behalf of the Company prior to Completion;
- (d) the Company shall allot and issue the number of Shares specified in Clause 3.4(c) to Waha, shall enter the name of Waha in the register of



members of the Company as registered holder of such shares and shall issue and deliver to Waha a share certificates duly executed by the Company for the shares subscribed by it;

- (e) the Company shall, using the proceeds of the issue of Shares and capital contribution provided for in Clause 3.4(c) in accordance with the written request from AerCap, redeem 51,128,856 redeemable ordinary shares of US\$1.00 each held by AerCap received by the Company prior to Completion for an aggregate redemption price of US\$56,100,000;
- (f) Waha shall apply for the issue to it of 40,100,000 Shares and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as Waha shall have been notified by or on behalf of the Company prior to Completion, an aggregate subscription price of US\$40,100,000 in respect of the Shares specified in this Clause 3.4(f);
- (g) the Company shall pay US\$66,000,000 to AerCap, by way of telegraphic transfer of immediately available funds to such account or in such other manner as the Company shall have been notified by or on behalf of the AerCap prior to Completion, in full and final settlement (including as to any obligation to pay interest) of certain loans which were made to the Company by AerCap pursuant to the terms of a master loan agreement between the Company and AerCap dated 27 March 2009;
- (h) AerCap shall apply for the issue to it of 40,100,000 Shares and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as AerCap shall have been notified by or on behalf of the Company prior to Completion, an aggregate subscription price of US\$40,100,000 in respect of the Shares specified in this Clause 3.4(h);
- the Company shall allot and issue the number of Shares specified in (i) Clause 3.4(f) to Waha and (ii) Clause 3.4(h) to AerCap, shall enter the names of AerCap and Waha in the register of members of the Company as registered holder of such shares and shall issue and deliver to each of AerCap and Waha a share certificate duly executed by the Company for the shares subscribed by them at which time Waha and AerCap shall each own 50% of the Shares in issue;
- (j) Salem Rashid Al Noaimi, Simon McLean, David Edwards and Wael Aburida shall be appointed to the Board as Waha Directors;
- (k) Cole Reese, Paul Rofe, Soeren Eric Ferre and Edward O'Byrne shall continue as AerCap Directors. AerCap Administrative Service Limited shall continue as the secretary of the Company;
- (l) each Group Member (other than the Company) shall take all necessary actions (including a meeting of the board of directors of such Group Member) necessary or desirable in connection with Completion;
- (m) the acquisition by the relevant member of the AerCap Group of each of the Delivered Waha A320-200 Aircraft shall be complete in accordance with the terms of the relevant Waha A320-200 Acquisition Agreement;
- (n) Waha shall procure that there is delivered to AerCap a legal opinion in respect of Waha from BarentsKrans N.V. which opinion shall be in form

and substance reasonably satisfactory to AerCap and which shall address, inter alia, the due execution by and enforceability against Waha of this Agreement and each other agreement entered or to be entered into by Waha in connection with Completion; and

- (o) AerCap shall procure that there is delivered to Waha a legal opinion from:
  - (i) Nauta Dutilh in respect of AerCap and each relevant member of the AerCap Group, which opinion shall be in form and substance reasonably satisfactory to Waha and which shall address, *inter alia*, the due execution by and enforceability against AerCap of this Agreement, the Deed of Warranty and each other agreement entered or to be entered into by AerCap or any relevant member of the AerCap Group in connection with Completion; and
  - (ii) McCann FitzGerald in respect of each of the Remarketing Agent, the Administrative Agent, the Cash Manager and the Company, which opinion shall be in form and substance reasonably satisfactory to Waha and which shall address, inter alia, the due execution by and enforceability against the Company and relevant members of the AerCap Group of this Agreement and each agreement entered or to be entered into by them (or any of them) in connection with Completion.

# 3.5 Required Equity Subscriptions

- (a) No later than 15 Business Days prior to the date of completion of each Required Equity Subscription, each of AerCap and Waha shall confirm in writing to the Cash Manager that it shall comply with its subscription obligations under this Clause 3.5 in respect of the relevant Required Equity Subscription.
- (b) Completion of the 2010 First Equity Subscription shall take place at AerCap House, Stationsplein 965, 1117 CE Schiphol Airport, The Netherlands on 5 January 2010 (or at such other place or date as AerCap and Waha agree in writing) (the "First 2010 Completion Date"). At the First 2010 Completion Date all but not some only of the following shall take place (to the extent that they have not taken place prior to the First 2010 Completion Date):
  - (i) AerCap shall apply for the issue to it of 22,000,000 Shares and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as AerCap shall have been notified by or on behalf of the Company prior to Completion, an aggregate subscription price of US\$22,000,000 in respect of the Shares specified in this Clause 3.5(b)(i);
  - (ii) Waha shall apply for the issue to it of 22,000,000 Shares and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as Waha shall have been notified by or on behalf of the Company prior to Completion, an aggregate subscription price of US\$22,000,000 in respect of the Shares specified in this Clause 3.5(b)(ii); and
  - (iii) the Company shall allot and issue the number of Shares specified in (i) Clause 3.5(b)(i) to AerCap and
    (ii) Clause 3.5(b)(ii) to Waha,

shall enter the names of AerCap and Waha in the register of members of the Company as registered holder of such shares and shall issue and deliver to AerCap and Waha a share certificates duly executed by the Company for the shares subscribed by them.

- (c) Completion of the 2010 Second Equity Subscription shall take place at AerCap House, Stationsplein 965, 1117 CE Schiphol Airport, The Netherlands on 15 March 2010 (or at such other place or date as AerCap and Waha agree in writing) (the "Second 2010 Completion Date"). At the Second 2010 Completion Date all but not some only of the following shall take place (to the extent that they have not taken place prior to the Second 2010 Completion Date):
  - (i) AerCap shall apply for the issue to it of 14,500,000 Shares and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as AerCap shall have been notified by or on behalf of the Company prior to Completion, an aggregate subscription price of US\$14,500,000 in respect of the Shares specified in this Clause 3.5(c)(i);
  - Waha shall apply for the issue to it of 14,500,000 Shares and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as Waha shall have been notified by or on behalf of the Company prior to Completion, an aggregate subscription price of US\$14,500,000 in respect of the Shares specified in this Clause 3.5(c)(ii);
  - (iii) the Company shall allot and issue the number of Shares specified in (i) Clause 3.5(c)(i) to AerCap and (ii) Clause 3.5(c)(ii) to Waha, shall enter the names of AerCap and Waha in the register of members of the Company as registered holder of such shares and shall issue and deliver to AerCap and Waha a share certificates duly executed by the Company for the shares subscribed by them.

# 3.6 **Default on Required Equity Subscriptions**

- (a) If either Waha or AerCap is in default of any of its obligations to complete a Required Equity Subscription (such Party being the "Defaulting Party"), then the Company shall notify each of the other Parties other than the Company (each a "Non Defaulting Party") in writing as soon as practicable thereafter. Upon receipt of such notification from the Company, a Non Defaulting Party may, in its absolute discretion and within a period of 15 Business Days, elect to:
  - (i) complete the subscription for some or all of the Shares, on the same terms, that the Defaulting Party failed to subscribe for in respect of the relevant Required Equity Subscription (such Shares in respect of which the Non Defaulting Party elects to subscribe for pursuant to this Clause 3.6(a)(i) being the "**Default Shares**"); and/or
  - (ii) subscribe (or procure that one of its Related Companies subscribes) for some or all of the Loan Notes with an aggregate nominal value equal to:

- (A) the aggregate subscription price that would have been payable by the Defaulting Party for the Shares the Defaulting Party failed to subscribe for in respect of the relevant Required Equity Subscription; less
- (B) the subscription price payable by the Non Defaulting Party in respect of the Default Shares (if any),

(such Loan Notes being the "Default Loan Notes").

- (b) If a Non Defaulting Party elects to complete the subscription for the Default Shares in accordance with Clause 3.6(a)(i):
  - (i) the Non Defaulting Party shall apply for the issue to it of the Default Shares and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as the Non Defaulting Party shall have been notified by or on behalf of the Company prior to completion, an aggregate subscription price equal to the subscription price that would have been payable in respect of the Default Shares if the Defaulting Shareholder had complied with its obligations in respect of portion of the relevant Required Equity Subscription represented by the Default Shares; and
  - (ii) the Company shall allot and issue the Default Shares to the Non Defaulting Party, shall enter the names of the Non Defaulting Party, in the register of members of the Company as registered holder of such shares and shall issue and deliver to the Non Defaulting Party a share certificate duly executed by the Company for the shares subscribed by it.
- (c) If a Non Defaulting Party elects to subscribe for Loan Notes in accordance with Clause 3.6(a)(ii):
  - the Non Defaulting Party shall apply (or procure that one of its Related Companies applies) for the issue to it of the Default Loan Notes and shall pay to the Company, by way of telegraphic transfer of immediately available funds to such account or in such other manner as the Non Defaulting Party shall have been notified by or on behalf of the Company prior to completion, an aggregate subscription price equal to the subscription price that would have been payable in respect of the Default Loan Notes if the Defaulting Shareholder had complied with its obligations in respect of the portion of the relevant Required Equity Subscription represented by the Default Loan Notes; and
  - (ii) the Company shall allot and issue the Default Loan Notes to the Non Defaulting Shareholder (or its relevant Related Company), enter the name of the Non Defaulting Shareholder (or its relevant Related Company), in the register of noteholders of the Company as registered holders of the Default Loan Notes (to which entry the Non Defaulting Shareholder hereby consents) and issue and deliver to the Non Defaulting Party a loan note certificate duly executed by the Company for the Default Loan Notes subscribed by it (or its relevant Related Company).

- (d) Where more than one Non-Defaulting Party elects to subscribe for Default Shares or Default Loan Notes, the aggregate amount of Default Shares and Default Loan Notes which each shall be entitled to subscribe for shall be scaled-back pro rata to their holdings of Shares as at the date of the relevant Required Equity Subscription. The split of subscriptions between Default Shares and Default Loan Notes for each such Non-Defaulting Shareholder shall be determined on the basis of the same percentage reflected in their original election to subscribe.
- (e) Any decisions or actions required to be taken by the Board in order to authorise the allotment and issue of the Default Loan Notes and/or the Default Shares shall be delegated to the Directors appointed by the Non Defaulting Party, which Directors (acting reasonably and in good faith in the best interests of the Company having regard to their fiduciary duties as Directors) shall have full authority on behalf of the Company to take and shall take, all such decisions and actions in order to authorise the allotment and issue of the Default Loan Notes and/or the Default Shares (as the case may be). The quorum requirement for any meeting of the Board shall be satisfied if two Directors appointed by the Non Defaulting Party are present and any Board resolution necessary to implement or authorise the allotment and issue of the Default Shares shall be deemed to have validly passed if approved by such Directors. The provisions of this Clause 3.6(e) apply only to resolutions of the Board necessary to implement or authorise the allotment and issue of the Default Loan Notes and/or Default Shares and not to any other resolutions of the Board and no meeting of the Board which is quorate solely because of this Clause 3.6(e) shall consider any other matters. For the avoidance of doubt, nothing in this Clause 3.6(e) shall prejudice the right of any Director to receive notice of and to attend Board meetings.
- (f) For the avoidance of doubt, the failure by a Defaulting Shareholder to comply with its obligations in respect of the 2010 First Equity Subscription shall in no way limit its obligations in respect of the 2010 Second Equity Subscription.

# 3.7 Allotment of Shares

- (a) Waha consents to its name being entered in the register of members of the Company in respect of the Shares to be subscribed for by it pursuant to Clauses 3.4(c), 3.4(f), 3.5(b)(ii), 3.5(c)(ii) and, if any, 3.6 and agrees that it will take such Shares with the benefit of the rights and subject to the restrictions set out in the Articles.
- (b) AerCap consents to its name being entered in the register of members of the Company in respect of the Shares to be subscribed for by it pursuant to Clauses 3.4(h), 3.5(b)(i), 3.5(b)(ii) and, if any, 3.6 and agrees that it will take such Shares with the benefit of the rights and subject to the restrictions set out in the Articles.
- (c) The Company, AerCap and Waha (to the extent that Waha is a Shareholder at the time of subscription for any Shares pursuant to this Agreement) consent to the subscriptions provided for in this Agreement and waive or agree to procure the waiver of any rights or restrictions which may exist in the Memorandum and Articles or otherwise which might prevent any such subscriptions.

# 4. CAPITAL; FURTHER FINANCE; DIVIDEND POLICY

# 4.1 Issue of Shares

- (a) Save for any Required Equity Subscription, the Directors may not allot any new Shares without the prior written approval of each of the Significant Shareholders in accordance with this Clause 4.1 and Clause 6.2.
- (b) Subject to Clause 4.1(a) and Clause 6.2 and except as otherwise determined by the Board with the prior approval in writing of all Significant Shareholders, all new Shares (the "**Offer Shares**") shall, before issue, be offered, on the same terms, to the Shareholders in accordance with the provisions of this Clause 4.1. Such offer shall be made by notice in writing:
  - (i) inviting each Shareholder to irrevocably offer to subscribe for up to its Agreed Proportion of the Offer Shares (such number of Shares being the Shareholder's "Equity Allocation");
  - (ii) specifying the subscription price per Share (the "Offer Price"); and
  - (iii) limiting the date (being not less than ten (10) Business Days after the date of the notice) within which the offer if not accepted will be deemed to be declined (the "Expiry Date").
- (c) If, following the Expiry Date in respect of an offer of Shares pursuant to Clause 4.1(b), one or more Shareholders (a "Declining Shareholder") has not irrevocably offered to subscribe for its full Equity Allocation, the Company shall serve notice in writing (a "Top-Up Offer Notice") of this fact on each Shareholder (if any) that has irrevocably offered to subscribe for its full Equity Allocation (an "Accepting Shareholder"). A Top-Up Offer Notice shall:
  - (i) specify:
    - (A) the number of Shares comprised in the Equity Allocation of the Declining Shareholder for which the Declining Shareholder has not irrevocably offered to subscribe (the "**Declined Shares**");
    - (B) the amount equal to the number of Declined Shares multiplied by the Offer Price (such amount being the "Declined Equity Finance"); and
  - (ii) invite each Accepting Shareholder to provide (as contemplated by Clause 4.1(d) below) to the Company, in addition to the amount equal to its full Equity Allocation multiplied by the Offer Price (such amount being its "Agreed Equity Finance"), some or all of the Declined Equity Finance to the Company in the manner specified below.
- (d) Within 5 Business Days of service of a Top-Up Offer Notice, each Accepting Shareholder shall notify the Company in writing (a "Top-Up Response") whether or not it wishes to provide any of the Declined Equity Finance and, if it does so wish, the amount of the Declined Equity Finance that it wishes to provide (or procure be provided) to the Company (the "Top-Up Finance"). Each Top-Up Response shall also specify the proportion of the Agreed Equity Finance and (if any) the Top-Up Finance that:

- (i) the Accepting Shareholder is to provide by means of a subscription for Shares at the Offer Price; and/or
- (ii) the Accepting Shareholder, or (in the case of AerCap) any member of the AerCap Group, or (in the case of Waha), any member of the Waha Group, is to provide by means of a subscription for loan notes in the Company, which loan notes shall be issued pursuant to a loan note instrument in the form of the 15% Loan Note Instrument.

If an Accepting Shareholder fails to serve a Top-Up Response within 5 Business Days of service of a Top-Up Offer Notice, it shall be deemed to have irrevocably agreed to provide its Agreed Equity Finance by way of subscription for Shares at the Offer Price.

- (e) Where more than one Accepting Shareholder indicates in its Top-Up Response that it wishes to provide Top-Up Finance and the aggregate amount of Top-Up Finance that the Accepting Shareholders wish to provide is in excess of the Declined Equity Finance, then unless the Board agrees otherwise, the amount of Top-Up Finance to be provided by such Accepting Shareholders shall be reduced so that:
  - (i) the aggregate amount of Top-Up Equity Finance is equal to the Declined Equity Finance; and
  - (ii) the amount of Top-Up Equity Finance to be provided by each such Accepting Shareholder is pro rata to the amount of their respective offers to provide Top-Up Equity Finance.
- (f) Any decisions or actions required to be taken by the Board in order to authorise the allotment and issue of Shares to an Accepting Shareholder or loan notes to an Accepting Shareholder (or a Related Company of an Accepting Shareholder) pursuant to this Clause 4.1 shall be delegated to the Directors appointed by the Accepting Shareholder, which Directors (acting reasonably and in good faith in the best interests of the Company having regard to their fiduciary duties as Directors) shall have full authority on behalf of the Company to take and shall take, all such decisions and actions in order to authorise the allotment and issue of such Shares and/or loan notes (as the case may be). The quorum requirement for any meeting of the Board shall be satisfied if two Directors appointed by the Accepting Shareholder are present and any Board resolution necessary to implement or authorise the allotment and issue of such Shares and/or loan notes (as the case may be) shall be deemed to have validly passed if approved by such Directors. The provisions of this Clause 4.1(f) apply only to resolutions of the Board necessary to implement or authorise the allotment and issue of Shares to an Accepting Shareholder and/or loan notes to an Accepting Shareholder (or a Related Company of an Accepting Shareholder) pursuant to this Clause 4.1 and not to any other resolutions of the Board and no meeting of the Board which is quorate solely because of this Clause 4.1(f) shall consider any other matters. For the avoidance of doubt, nothing in this Clause 4.1(f) shall prejudice the right of any Director to receive notice of and to attend Board meetings.
- (g) Subject to Clause 4.1(h), if after exhaustion of the preceding provisions of this Clause 4.1, any Offer Shares have not be accepted (or substituted by means of any loan contemplated by Clause 4.1(d)(ii)), such Shares may

be issued to such persons and upon such terms and conditions as the Directors deem most beneficial to the Company.

- (h) Except as may otherwise be agreed with the prior written consent of each Significant Shareholder, the Company shall not allot any Shares to any person that is not an existing Shareholder or a Related Company of an existing Shareholder (a "Third Party Allottee"):
  - (i) unless the Shares are allotted to the Third Party Allottee upon the condition that such allottee agrees to execute and executes a Deed of Adherence and agrees to be bound by the terms of and assume the rights and obligations of the Shareholders under this Agreement;
  - (ii) if, following any such allotment, the Third Party Allottee (together with any person connected with it) would beneficially own more than fifty per centum (50%) but less than one hundred per centum (100%) of the issued Shares;
  - (iii) if the Third Party Allottee (or any Related Company or person connected with it) is:
    - (A) directly or indirectly a substantial competitor of any member of the AerCap Group (other than any member of the Waha Group) or the Waha Group (other than any member of the AerCap Group); or
    - (B) is a person who might reasonably be regarded as an aircraft operating lessor in competition with any member of the AerCap Group (other than any member of the Waha Group) or the Waha Group (other than any member of the AerCap Group); and
  - (iv) if the allotment of any Shares to a Third Party Allottee would result in the Company being in breach of any financing arrangements to which it is party or which would otherwise constitute an event triggering the right for a third party to call for the Company to pay a sum of money.

# 4.2 **Funding support by the parties**

- (a) Save as set out in Clauses 3 (*Completion*), 4.3, 4.4 and this Clause 4.2, the Parties intend that the Group should be self-financing and should obtain additional funds from third parties without recourse to the Shareholders. The Parties anticipate that, following completion of the Required Equity Subscriptions, the Group will have sufficient capital to fund its future obligations. However, the Parties acknowledge their intention to support the Company and the Group in accordance with the Approved Budget in place from time to time and will in good faith consider providing such funding as may reasonably be required to ensure that the Company and the Group is adequately funded.
- (b) Without prejudice to the generality of Clause 4.2(a), if at any time the Cash Manager delivers a Funding Support Certificate to the Board and at the next succeeding meeting of the Board:

- (i) no Director, acting reasonably, alleges that the Funding Support Certificate contains a manifest error; or
- a Director, acting reasonably, alleges that the Funding Support Certificate contains a manifest error but the auditors of the Company (who shall be instructed for this purpose by the Company as soon as possible following such board meeting) confirm in writing that the Funding Support Certificate does not contain a manifest error,

the Board will consider whether or not to approach the Company's bankers or other financial institutions or, in appropriate circumstances, to seek such further finance from the Shareholders. Where the Board resolves to seek such further finance from the Shareholders, provided that the Shareholders agree on the amount and method of providing the finance, the Shareholders will be requested, in each Shareholder's absolute discretion, to contribute amounts to the Company pro rata to the aggregate nominal value of the Shares in issue at such time which are beneficially owned by such Shareholder expressed as a percentage of the issued Shares at such time (whether by subscribing for Shares, stock or debentures or by way of loan or otherwise), at the same time and on the same terms.

#### 4.3 Failure to obtain funding support from bankers or Shareholders

If, at any time the Board determines that the Business requires further finance (in addition to that contemplated by Clause 4.2) or:

- (a) the Cash Manager has delivered a Funding Support Certificate to the Board;
- (b) either sub-paragraph (i) or (ii) of Clause 4.2(b) applies in respect of such Funding Support Certificate; and
- (c) by the date 30 days after the date of such Funding Support Certificate (or 15 Business Days after delivery of the auditor's confirmation referred to in Clause 4.2(b)(ii) (whichever is later)) either:
  - (i) the Board has not agreed to approach the Company's bankers or other financial institutions or the Shareholders to obtain such further finance;
  - (ii) the Board has agreed to approach the Company's bankers or other financial institutions to obtain such further finance but none of these approaches have resulted in the provision of the further finance required (as specified by the Cash Manager or the Board, as the case may be); or
  - (iii) the Board has resolved to seek such further finance from the Shareholders but the Shareholders have been unable to agree on the amount, method or terms of providing the finance,

any Shareholder may elect by notice in writing to the Company that it or (in the case of AerCap) any member of the AerCap Group, or (in the case of Waha), any member of the Waha Group, shall subscribe for loan notes in an aggregate nominal amount equal to the required amount, such loan notes to the issued pursuant to a loan note instrument in the form of the

15% Loan Note Instrument. Following the making of any such election, the Company will take, and each Shareholder will instruct the Directors appointed by it to take, all action necessary to implement the issue of the relevant loan notes pursuant to a loan note instrument in the form of the 15% Loan Note Instrument. If more than one Shareholder makes an election pursuant to this Clause 4.3, the amount of such loan notes that each such Shareholder shall be entitled to make to the Company shall be an amount equal to it its Agreed Proportion of such loan notes.

### 4.4 Guarantees to third parties

Notwithstanding Clause 4.3, where the Remarketing Agent confirms to the Company that in the context of the lease or sale of an Aircraft the lessee or buyer thereof (as the case may be) requires a guarantee of the obligations of the relevant Group Member under the lease or sale agreement (as the case may be), the Shareholders shall, if they so elect, provide such guarantee to such person on a several basis only and in the form recommended by the Remarketing Agent.

# 4.5 **Dividend Policy**

- (a) Subject to Clause 4.5(b), unless otherwise agreed by the Significant Shareholders in writing, the profits of the Company available for distribution shall be dealt with as the Board may determine from time to time.
- (b) Unless otherwise agreed by the Significant Shareholders in writing, for so long as there is any amount of principal or accrued interest outstanding in respect of any loan notes issued by the Company to a Party pursuant to Clauses 3.6 or 4.3 or any Loan Notes:
  - (i) no distribution or other return of capital by the Company shall be made; and
  - (ii) the Company shall apply all available cash flow, and otherwise use its best endeavours, to procure repayment of all such loan notes as soon as practicable and in the following order of priority:
    - (A) first, in repaying all Loan Notes (with the first issued Loan Notes being repaid first and any subsequently issued Loan Notes being repaid secondly); and
    - (B) then, in repaying all loan notes issued pursuant to Clause 3.6 and 4.3, the aggregate amount so repaid to be paid to the holders of all such loan notes in proportion to their holdings of such loan notes.

### 4.6 **Restriction on reduction in AerCap interest in the Company**

AerCap undertakes to Waha that, save with Waha' consent:

(a) it will not transfer any Shares if and to the extent that such transfer would result in a breach of, or give rise to a mandatory pre-payment event under, any of the "facilities" (as defined in paragraph 4.3 of Schedule 2 to the Deed of Warranty) disclosed in the Data Room (as defined in the Deed of Warranty) or any future financing facilities made available to the Company; and

(b) it will subscribe for any new Shares offered to it in accordance with the terms of this Agreement if and to the extent that a failure to do so would result in a breach of, or give rise to a mandatory pre-payment event under, any of such facilities or future financing facilities.

# 5. DIRECTORS AND MANAGEMENT

# 5.1 **Supervision by the Board**

The Board shall be responsible for the overall direction, supervision and management of the Company. The Board shall not, however, take any decision in relation to any of the Reserved Shareholder Matters without the prior approval of the Shareholders.

# 5.2 Entitlement to appoint Directors

For so long as each Shareholder (or, in the case of AerCap, any member of the AerCap Group or, in the case of Waha, any member of the Waha Group) beneficially owns the percentage of the issued Shares set out in column (1) below, it shall be entitled to appoint up to the number of persons set out in column (2) below as Directors and to remove from office any person so appointed and to appoint another person in his place.

(1) Percentage of issued Shares	(2) Number of Directors
Equal to or greater than 50%	4
Greater than 25% but less than 50%	2
Equal to or greater than 15% but not greater than 25%	1
less than 15%	None

# 5.3 Shareholder's Obligation to reduce Directors

Each Shareholder agrees with the other Shareholders that if at any time the percentage of the issued Shares which it beneficially owns is reduced (by whatever means) such that the number of Directors which it is entitled to appoint under Clause 5.2 is thereby reduced, it shall forthwith upon such reduction procure the removal of such number of Directors appointed by it as is necessary to reflect this reduction. If any Shareholder fails immediately to procure the removal of such number Directors as required under this Clause 5.3, then the office of such Director(s) shall be automatically vacated and where the number of Directors to be removed by such Shareholder is less than the number appointed by it immediately prior to the reduction contemplated by this Clause 5.3, the other Shareholders shall be entitled to determine which of the Directors appointed by such Shareholder shall resign.

# 5.4 Voting powers of Directors

Where not all of the Directors appointed by a Shareholder are present at a meeting of the Board (whether in person or through an alternate), then the Director or Directors appointed by a Shareholder who are present at such



meeting (whether in person or through an alternate) shall be deemed to exercise the number of votes as is equal to the number of Directors which such Shareholder is entitled, as of the date of such meeting, to appoint.

# 5.5 **Right to appoint Chairman**

Each Significant Shareholder shall have the right exercisable alternately for a period of one year of nominating one of the Directors to be the Chairman of meetings of the Board and Shareholders and a Chairman so appointed shall hold office as such until the termination of the next annual general meeting following his appointment or (if earlier) the first day after such appointment on which the Shareholder who has nominated such Chairman ceases to be a Significant Shareholder.

### 5.6 **Rotation of Chairman**

Notwithstanding the generality of Clause 5.5, the first Chairman shall be nominated by Waha, and the second Chairman shall be nominated by AerCap.

# 5.7 Substitute Chairman

If the Chairman is unable to attend any meeting of the Board, then the Shareholder who nominated him shall be entitled to appoint another Director to act as chairman in his place at such meeting.

#### 5.8 No casting vote for Chairman

Questions arising at any meeting of the Board shall be decided by a majority of votes. In the case of an equality of votes at any meeting of the Board the Chairman shall not be entitled to a second or casting vote and the Chairman shall not have a second or casting vote at any meeting of the Shareholders of the Company.

# 5.9 **Procedure for appointment/removal of Director**

Any appointment or removal pursuant to this Clause shall be made by notice in writing served on the Company and the Company agrees to procure that such appointment and/or removal shall be effected as soon as possible following receipt of such notice.

### 5.10 Appointment of Alternate Director

Notwithstanding any provision of the Articles, each Director and each person appointed to the board of directors of any subsidiary undertaking of the Company shall be entitled to appoint any person to be an alternate director, shall not be entitled to be paid any remuneration by any member of the Group, shall not be required to hold any share qualification, shall not be subject to retirement by rotation and shall not be removed except by the Shareholder which appointed him or pursuant to Clause 5.3.

#### 5.11 **Committees of the Board**

Each Director shall have the right to be appointed to any committee or sub-committee of or established by the Board provided that this right may be waived by that Director or any other Director appointed by the same Shareholder on his behalf including by approving the establishment of such committee or sub-committee.



# 5.12 Shareholder indemnity regarding removed Director

Each Shareholder agrees with each of the other Shareholders that if it removes a Director appointed by it in accordance with this Clause 5 or if any such Director is removed pursuant to Clause 5.3 it shall be responsible for, and shall indemnify the Company and the other Shareholders against, any claims by such Director arising out of the Director's removal or loss of office. Each Shareholder acknowledges that the Company shall not be obliged to procure any insurance in respect of its Directors and officers.

# 5.13 Quorum

Subject to Clause 3.6(e) and any other provisions of this Agreement which expressly provide for a different quorum, the quorum for any meeting of the Board (including any adjourned meeting) shall comprise one Director appointed by each Significant Shareholder or their duly appointed alternates present in person. If a quorum is not present the meeting shall be adjourned to the same time and place five Business Days later (or if that day is not a Business Day, the next succeeding Business Day).

# 5.14 Notice of meetings of Board

A meeting of the Board shall, unless otherwise agreed by all of the Directors, be called by notice in writing to all Directors of no less than five (5) Business Days (inclusive of the date of service or deemed service and the date of the meeting) or such lesser period as may be agreed to enable the Company to give any instructions, directions, consent or response to each Servicer in accordance with the terms of the Servicing Agreements and shall always take place on a Business Day unless the Significant Shareholders agree otherwise. Any such notice shall specify the place, the day and the hour of the meeting, and the nature of the business to be discussed thereat.

# 5.15 Location for Board Meetings

Meetings of the Board shall be held in the Netherlands at such location as may be nominated by the Secretary **provided that** any such location shall meet the requirements of Clause 2.2(b).

# 5.16 Frequency of Board Meetings

The Board shall meet not less frequently that four (4) times per annum and, in any case, no more than four (4) months shall elapse between a meeting of the Board and the next subsequent meeting of the Board.

#### 5.17 Application of Board provisions to other Group Members

Save for Clause 5.15, this Clause 5 shall apply to every other Group Member *mutatis mutandis* **provided that** for such purposes the term "**Shareholders**" shall continue to have the meaning given to it in this Agreement.

# 6. **RESERVED MATTERS**

### 6.1 Use of powers

The Shareholders shall use their respective powers to ensure, so far as they are legally able, that no action or decision relating to any of the matters specified in Clause 6.2 (*Reserved Shareholder Matters*) is taken (whether by the Board, the

Company or managers within the Company) unless each Significant Shareholder (provided such Significant Shareholder is not, at the relevant time, a Defaulting Party) has given its prior approval in writing to proceed.

#### 6.2 **Reserved Shareholder Matters**

The Reserved Shareholder Matters are:

(a) *issue of Shares* 

any issuance of Shares or securities or other instruments convertible into shares (including but not limited to Loan Notes) other than those issued at or in connection with Completion or otherwise contemplated by this Agreement;

(b) *material litigation* 

decisions relating to the conduct (including the settlement) of material legal proceedings to which any Group Member is a party;

(c) change in nature of Business

materially changing the scope or nature of the Business (as defined in Clause 2.1(a)) of the Company or the scope or nature of the business of any other Group Member;

(d) asset purchase agreements

any Group Member entering into any material aircraft or asset purchase agreements;

(e) *winding-up* 

any proposal to wind up any Group Member or other voluntary proceeding seeking liquidation, administration (whether out of court of otherwise), reorganisation, examination, readjustment or other relief under any bankruptcy, insolvency or similar law or the consent by any Group Member to a decree or order for relief or any filing of a petition, application or document under such law or to the appointment of a trustee, receiver, administrator (whether out of court or otherwise), examiner or liquidator or any other voluntary action by any Group Member in furtherance of its bankruptcy, reorganisation, administration (whether out of court or otherwise), examiner or termination of its corporate status;

(f) *acquisitions and disposals* 

any major asset or business acquisitions or disposals by any Group Member;

(g) *partnerships and joint ventures* 

any Group Member entering into (or terminating) any material partnership, joint venture, profit-sharing agreement, technology licence or collaboration;



### (h) material dealings

any material dealings between any Group Member and a Significant Shareholder (or any Related Company of a Significant Shareholder) or a Director;

(i) *financing arrangements* 

any Group Member entering into any financing arrangements of a material nature with any third party;

(j) dividends

any change to the dividend policy of any Group Member;

(k) *capital expenditure* 

any Group Member incurring any expenditure in connection with the acquisition of aircraft (other than (i) pursuant to the Airbus Purchase Agreement or (ii) pursuant to any option agreement (and any documents related thereto) which any Group Member may enter into in connection with any aircraft financed or to be financed under the UBS Facility) in excess of that contemplated in the then current Approved Budget;

(l) borrowings

any Group Member borrowing any money from persons other than the Shareholders under arrangements expressly contemplated by this Agreement;

(m) charges

creating any mortgage, charge, encumbrance or other security interest of any nature in respect of all or any material part of any Group Member's undertaking, property or assets; and

(n) Approved Budget

approval of any Approved Budget and any modification thereto.

# 6.3 Servicing Agreements Reserved Matters

Each instruction or direction given by the Company or action taken by the Company pursuant to the Servicing Agreements shall require the approval of the Board subject as provided below and subject further to Clause 6.4,:

- (a) Each AerCap Director shall abstain from voting on a resolution:
  - (i) to approve:
    - (A) an instruction of the Board to the Servicer to withdraw from acting as Servicer pursuant to clause 3.2(c) of the Servicing and Remarketing Agreement;
    - (B) the giving of the express prior written approval of the Company for the purposes set out in clause 7.4(a)(vi) of the Servicing and Remarketing Agreement;



- (C) the giving of consent by the Company pursuant to clause 12 of the Servicing and Remarketing Agreement;
- (D) a consent of the Company given pursuant to clause 8.1 of the Administrative Agency Agreement;
- (E) a consent of the Company given pursuant to clause 11.1(a) of the Cash Management Agreement;
- (ii) a Special Board Resolution pursuant to clause 2.3(k) of the Administrative Agency Agreement;
- (iii) a decision of the Board for the purposes of clause 3.3(b) of the Administrative Agency Agreement;
- (iv) a decision of the Board for the purposes of clause 9.2(b) of the Cash Management Agreement;
- (b) any resolution passed pursuant to clause 10.2(b) or 10.2(c) of the Servicing and Remarketing Agreement shall require the approval of at least one Director appointed by each Significant Shareholder (excluding AerCap) and each AerCap Director shall abstain from voting on that resolution;
- (c) a resolution to terminate the Administrative Agency Agreement pursuant to clause 7.2(b) or (c) of the Administrative Agency Agreement shall require the approval of at least one Director appointed by each Significant Shareholder (excluding AerCap) and each AerCap Director shall abstain from voting on that resolution; and
- (d) a decision by the Company to exercise its rights in respect of a default by a Service Provider under any of the Services Agreements shall require the prior written consent of each Significant Shareholder (other than AerCap).

# 6.4 Servicing Agreements Reserved Matters — Waha Delegation

If and to the extent that any matter which is the subject of an approval, decision, resolution or consent pursuant to any of sub-paragraphs (a) to (d) of Clause 6.3 arises out of or relates to any default by any member of the Waha Group under the Services Delegation and Co-operation Agreement:

- (a) Clause 6.3 shall not apply so as:
  - (i) to require any AerCap Director to abstain from voting on any such resolution; or
  - (ii) restrict in any manner whatsoever AerCap or any AerCap Director from giving or participating in the giving of any such approval, decision or consent; and
- (b) each Waha Director shall abstain from voting on any such resolution and Waha and each Waha Director shall be restricted from participating in the giving of any such approval, decision or consent as if Clause 6.3 applied to such resolution, approval decision or consent with references therein to "AerCap" replaced by references to "Waha".

# 6.5 Method of approval by Shareholders

The approval of a Significant Shareholder (provided such Significant Shareholder is not, at the relevant time, a Defaulting Party) to any of the Reserved Shareholder Matters may be given:

- (a) in writing;
- (b) by its authorised representative for this purpose;
- (c) by its representative at a general meeting of the Company; or
- (d) by a director appointed by that Significant Shareholder at a meeting of the Board.

#### 6.6 Meetings of Shareholders

General meetings of Shareholders shall take place in accordance with the applicable provisions of the Articles. Any matters requiring a general meeting of or approval by the Shareholders but not covered by the Reserved Shareholder Matters, shall be dealt with in accordance with the Memorandum and Articles.

# 7. INFORMATION AND REPORTING

#### 7.1 **Inspection and information**

- (a) Each Shareholder may examine the separate books, records and accounts to be kept by the Company and each Group Member. Each Shareholder shall be entitled to receive all information, including monthly management accounts of the Group and operating statistics and other trading and financial information relating to the Group as it may reasonably request, in such form as a Shareholder reasonably requests to keep it properly informed about the business and affairs of the Company and generally to protect its interests as a Shareholder.
- (b) Each Director appointed by a Significant Shareholder shall be entitled to exercise all rights of the Company under the Servicing Agreements to make enquiries of and receive information from the Servicers.
- (c) Each Shareholder and Director shall be entitled to have at all reasonable times the facility of remote electronic access to the contract management and other appropriate systems of the AerCap Group relating to the Aircraft and the Business but only to the extent that those systems give access to information relating solely to the Aircraft and the Business.

#### 7.2 Accounts and Budgets

Without prejudice to the generality of Clause 7.1, the Company shall supply each Shareholder with copies of:

(a) annual audited accounts for the Company complying with all relevant legal requirements and the Group annual audited consolidated accounts (which, for the avoidance of doubt, shall include, notwithstanding that such companies are not members of the Group, Aircraft Lease Securitisation II Limited (a company incorporated in Bermuda with registered number 41915) and results of beneficially owned aircraft via AerFunding I Limited (a company incorporated in Ireland with registered number 905464));

<sup>31</sup> 

- (b) the Approved Budget;
- (c) a quarterly up-date on the financial position of the Business and any other information relevant to the Business or the Group, including any material contract of the Group;
- (d) quarterly management accounts of the Group (which, for the avoidance of doubt, shall include quarterly management accounts for June 2009); these shall include a consolidated profit and loss account, balance sheet and cash flow statement with accompanying notes together with a statement of progress against the relevant Approved Budget; and
- (e) monthly management accounts of the Group (which, for the avoidance of doubt, shall include monthly management accounts for June 2009); these shall include a consolidated profit and loss account and balance sheet.

# 7.3 May Management Accounts

Each of the Parties undertakes to each of the other parties to use their best endeavours to ensure that the Management Accounts (as defined in the Deed of Warranty) are audited to the standard required for the reporting purposes of the Waha Group by no later than 8 July 2009.

# 8. **CONFIDENTIALITY**

# 8.1 **Confidentiality obligation**

Each Shareholder shall use all reasonable endeavours to keep confidential (and to ensure that its officers, employees, agents and professional and other advisers keep confidential) any information:

- (a) which it may have or acquire before or after the date of this Agreement in relation to the Group's customers, Business, assets or affairs; this includes, without limitation, any information provided pursuant to Clause 7 (*Information and Reporting*);
- (b) which it may have or acquire before or after the date of this Agreement in relation to the customers, Business, assets or affairs of Waha (if the Party is AerCap) or of AerCap (if the Party is Waha) resulting from:
  - (i) negotiating this Agreement;
  - (ii) being a Shareholder;
  - (iii) having appointees on the Board; or
  - (iv) exercising its rights or performing its obligations under this Agreement; or
- (c) which relates to the contents of this Agreement (or any agreement or arrangement entered into pursuant to this Agreement).

No Shareholder shall use for its own business purposes or disclose to any third party any such information (collectively, "**Confidential Information**") without the consent of each other Shareholder. In performing its obligations under this Clause 8, each Shareholder shall apply the confidentiality standards and procedures it applies generally in relation to its own confidential information.

# 8.2 Exceptions from confidentiality obligation

The obligation of confidentiality under Clause 8.1 does not apply to:

- (a) information which is independently developed by the relevant party or acquired from a third party to the extent that it is acquired with the right to disclose the same;
- (b) the disclosure of information to the extent required to be disclosed by law, any stock exchange regulation (including any SEC form documents), in connection with any financial statement of any Shareholder (or member of any group of companies of which a Shareholder is a member) or the disclosures or explanatory notes to any such financial statement, or any binding judgment, order or requirement of any court or other competent authority;
- (c) the disclosure of information to any tax authority to the extent reasonably required for the purposes of the tax affairs of the party concerned;
- (d) the disclosure (subject to Clause 8.3) in confidence to a bona fide potential purchaser of its Shareholder Interest or to a Shareholder's financiers or to its professional advisers, auditors, directors, officers or agents of information reasonably required to be disclosed for a purpose reasonably incidental to this Agreement;
- (e) information which becomes within the public domain (otherwise than as a result of a breach by the disclosing Shareholder of this Clause 8); or
- (f) any announcement made in accordance with the terms of Clause 22 (Announcements).

#### 8.3 Employees, agents and advisers

Each Shareholder shall inform any officer, employee or agent or any financiers or professional or other adviser advising it in relation to the matters referred to in this Agreement, or to whom it provides Confidential Information, that such information is confidential and shall instruct them:

- (a) to keep it confidential; and
- (b) not to disclose it to any third party (other than those persons to whom it has already been disclosed in accordance with the terms of this Agreement).

The disclosing Shareholder is responsible for any breach of this Clause 8 by the person to whom it is disclosed.

# 8.4 **Return of Confidential Information**

If this Agreement terminates or a person ceases to be a Shareholder (such person being the "**Departing Shareholder**"), any Shareholder may by notice require:

- (a) each other Shareholder (in the case of a termination of this Agreement); or
- (b) the Departing Shareholder

to return the Confidential Information of the Shareholder which gives such notice. If so, each such other Shareholder or the Departing Shareholders (as the case may be) shall:

- (i) return all documents containing Confidential Information which have been provided by or on behalf of the Shareholder demanding the return of Confidential Information; and
- (ii) save to the extent that:
  - (A) there is a legal obligation or requirement to retain them; or
  - (B) such destruction would prejudice the ability of such Shareholder to enforce its rights hereunder,

destroy any copies of such documents and any document or other record reproducing, containing or made from or with reference to the Confidential Information (save, in each case, for any submission to or filings with governmental, tax or regulatory authorities). Subject to the foregoing, each relevant Shareholder shall return or destroy the Confidential Information as soon as practicable after receiving notice.

#### 8.5 **Survival after termination**

The provisions of this Clause 8 continue to apply if this Agreement ceases or is terminated.

#### 9. **REGULATORY MATTERS**

#### 9.1 **Co-operation**

The Shareholders shall co-operate with each other to ensure that all information necessary or desirable for making (or responding to any requests for further information following) any notification or filing made in respect of this Agreement, or the transactions contemplated by it, is supplied to the party dealing with such notification and filing and that they are properly, accurately and promptly made.

#### 9.2 **Regulatory Action**

If any material Regulatory Action is taken or threatened, the Shareholders shall promptly meet to discuss:

- (a) the situation and the action to be taken as a result; and
- (b) whether any modification to the terms of this Agreement (or any agreement entered into pursuant to this Agreement) should be made in order that any requirement (whether as a condition of giving any approval, clearance or consent or otherwise) of any other regulatory authority may be reconciled with, and within the intended scope of, the business arrangement contemplated by this Agreement. The Shareholders shall co-operate to give effect to any agreed modifications.

# 10. RELATIONSHIP OF THE COMPANY WITH SHAREHOLDERS

## 10.1 **Claims by Company**

- (a) If any Group Member has or may have any claim against a Shareholder (the "**Respondent**") or a Related Company of such Respondent, the quorum requirement for any meeting of the Board at which such claim against the Respondent or the Respondent's Related Company is considered shall be satisfied if two Directors appointed by the other Shareholder are present and any Board resolution regarding such claim shall be deemed to have validly passed if approved by such other Shareholder's appointed Directors. The provisions of this Clause 10.1(a) apply only to resolutions of the Board and no meeting of the Board which is quorate solely because of this Clause 10.1(a) shall consider any other matters. For the avoidance of doubt, nothing in this Clause 10.1(a) shall prejudice the right of any Director to receive notice of and to attend Board meetings or any right of the defendant Shareholder itself to dispute the claim.
- (b) If the Company is entitled to terminate a Servicing Agreement in accordance with its terms, the AerCap Directors (if any) shall not withhold their consent to the Company resolving to proceed with such termination. A decision to terminate a Servicing Agreement shall not be a Reserved Shareholder Matter for the purposes of Clause 6.2(h) (*Reserved Shareholder Matters material dealings*).

#### 10.2 **Promotion of the Company's interest**

The Shareholders confirm their intention to promote the best interests of the Company and to consult fully on all matters materially affecting the development of the business of the Company. Each Shareholder shall act in good faith towards the other in order to promote the success of the Company.

# 11. TAX MATTERS

Each Shareholder agrees to co-operate to such extent as may be reasonably requested in connection with the making of any returns, claims or elections for taxation purposes:

- (a) by each other Shareholder in relation to their taxation affairs for any period ending before the date of this Agreement; or
- (b) by any Group Member in relation to the taxation affairs of the Group Member.

# 12. TRANSFER OF SHARES

#### 12.1 General

- (a) The provisions of this Clause 12 apply in relation to any transfer, or proposed transfer, of Shares or any interest in Shares.
- (b) For the purposes of this Agreement the following shall be deemed (but without limitation) to be a transfer of Shares:



- (i) any direction (by way of renunciation or otherwise) by a holder entitled to an allotment or transfer of Shares that Shares be allotted or issued or transferred to some person other than himself; and
- (ii) any sale or any other disposition (including by way of charge or other security interest,) of any legal or equitable interest in Shares (including any voting right attached to it), (i) whether or not by the relevant holder, (ii) whether or not for consideration, and (iii) whether or not effected by an instrument in writing but excluding:
  - (A) any charge or security interest created in favour of any person to which Shares could be transferred in compliance with paragraphs (b) to (f) of Clause 12.3; and
  - (B) for the avoidance of doubt, the creation of the AerCap Share Charge.

#### 12.2 General Restriction on transfer of Shares

Except with the prior written consent of each Significant Shareholder, no Shareholder shall:

- (a) transfer any Shares, except in accordance with this Clause 12;
- (b) grant, declare, create or dispose of any right or interest in any Shares save in accordance with this Clause 12; or
- (c) create or permit to exist any pledge, lien, fixed or floating charge or other encumbrance over any Shares (excluding, for the avoidance of doubt, the creation of the AerCap Share Charge).

# 12.3 **Restrictions on Transfer of Shares**

Except (i) as provided in Clause 12.4, (ii) as may otherwise be agreed with the prior written consent of each Significant Shareholder or (iii) as may occur pursuant to the AerCap Share Charge, no Shareholder (the "Seller") may transfer an amount of Shares (the "Sale Shares") to any person that is not an existing Shareholder (a "Third Party Purchaser"):

- (a) unless the Sale Shares are transferred to the Third Party Purchaser upon the condition that such transferee agrees to execute and executes a Deed of Adherence and agrees to be bound by the terms of and assume the rights and obligations of such Seller under this Agreement;
- (b) unless the Significant Shareholders (other than the Seller) are satisfied that the Third Party Purchaser has a net worth of not less than US\$100,000,000;
- (c) if, following any such transfer (or any other simultaneous transfer of Shares made to it), the Third Party Purchaser (together with any person connected with it or Related Company of it) would beneficially own more than fifty per centum (50%) but less than one hundred per centum (100%) of the issued Shares at such time; or
- (d) if, where the Seller is not a member of the AerCap Group, the Third Party Purchaser (or nominee of it or any person connected with it) is:

- (i) directly or indirectly a substantial competitor of any member of the AerCap Group; or
- (ii) is a person who might reasonably be regarded as an aircraft operating lessor in competition with any member of the AerCap Group; and
- (e) if, where the Seller is not a member of the Waha Group, the Third Party Purchaser (or nominee of it or any person connected with it) is:
  - (i) directly or indirectly a substantial competitor of any member of the Waha Group; or
  - (ii) is a person who might reasonably be regarded as an aircraft operating lessor in competition with any member of the Waha Group; and
- (f) if the transfer to the Third Party Purchaser would result in the Company being in breach of any financing arrangements to which it is party or which would otherwise constitute an event triggering the right for a third party to call for the Company to pay a sum of money; or
- (g) if such transfer obliges the Seller to procure the making of a Tag-Along Offer, unless such Tag-Along Offer has been made and (if accepted) completed.

For the avoidance of doubt, nothing in this Agreement or the Articles shall prevent a transfer of Shares pursuant to the AerCap Share Charge.

# 12.4 **Permitted Transfers**

Notwithstanding the provisions of this Clause 12:

- (a) any Shareholder which is a company may at any time transfer all or any of its Shares to a Related Company **provided that**:
  - (i) save where the transferee is an AerCap Shareholder or a Waha Shareholder, prior to such transfer, the transferring Shareholder shall procure that such Related Company (or such other Related Company of the transferring Shareholder that is acceptable to the Significant Shareholders (other than any Significant Shareholder in respect of which the transferre is a Related Company)) provide a parent company guarantee to the Company and the non transferring Shareholders in respect of its obligations under this Agreement, which guarantee shall be in a form acceptable to the Significant Shareholders (other than any Significant Shareholder in respect of which the transferre is a Related Company);
  - (ii) if at any time following such transfer, such transfere ceases to be a Related Company of such company then it shall, within twenty one (21) days of so ceasing, transfer any Shares held by it to such company or any Related Company of such company and failing such transfer within such period, the relevant holder(s) shall be deemed to have given a Transfer Notice pursuant to this Clause 12; and

- (iii) no transfer shall be permitted pursuant to this Clause 12.4 if, in the reasonable opinion of the tax advisers to the Company, the acquisition of Shares by the proposed transferee would materially adversely affect the taxation status or future liability to taxation of the Group; and
- (b) any Shares may be transferred pursuant to the acceptance of a Tag-Along Offer.

## 12.5 Transfer Notice

- (a) Before the Seller makes any transfer of the Sale Shares, the Seller shall first give the Company and each other Shareholder (each a "Continuing Party" and together the "Continuing Parties") a notice (a "Transfer Notice") which notice:
  - (i) shall identify the proposed transferee and where such transferee is a Third Party Purchaser it shall provide details of such Third Party Purchaser; and
  - (ii) shall specify the number of Shares which comprise the Sale Shares; and
  - (iii) shall contain an offer to sell the Sale Shares to each Continuing Party on the same terms (including as to price) as may have been agreed between the Seller and the proposed transferee (the "Transfer Price").
- (b) Involuntary Transfer Deemed delivery of Transfer Notice

Where any Transfer Notice is deemed to have been given in accordance with this Agreement, the holder of the Shares to which the deemed Transfer Notice relates shall be deemed to be a Seller and:

- (i) the Transfer Notice shall be deemed to:
  - (A) be in respect of all (but not part only) of the Shares held by such Seller which Shares shall, for the purposes of the application of Clause 12.6 to the transfer procedure initiated by the deemed service of such Transfer Notice, be treated as the "**Sale Shares**"; and
  - (B) specify the number of Shares which comprise the Sale Shares;
- (ii) the Transfer Price shall be such price as shall be agreed between the Significant Shareholders or, in the event that the Significant Shareholders fail to agree the price within five (5) Business Days of the date on which such Transfer Notice is deemed to have been given, the Adjusted Fair Value of the Shares which are to be transferred; and
- (iii) if, in aggregate, the Continuing Parties fail to accept the offer made pursuant to Clause 12.6(a) for all the Sale Shares within the period referred to in Clause 12.6(a), the Transfer Notice shall be deemed to have been withdrawn in respect of any Sale Shares which have not been accepted by the Continuing Parties (but, for the avoidance

of doubt, the Transfer Notice shall not be deemed to be withdrawn in respect of any Sale Shares which have been accepted by the Continuing Parties in accordance with Clause 12.6(a)).

- (c) A Transfer Notice may not be withdrawn without the prior written consent of each Continuing Party or unless the circumstances described in Clause 12.5(b)(iii) apply.
- (d) A Transfer Notice shall constitute the Company as the agent of the Seller for the sale of the Sale Shares at the Transfer Price.

#### 12.6 Transfer Procedure

- (a) The Company shall as soon as practicable following receipt of a Transfer Notice or, where later, upon the determination of the Transfer Price give notice in writing to each Continuing Party informing it of:
  - (i) the amount of Sale Shares that is available to purchase;
  - (ii) the number of Shares which comprise the Sale Shares; and
  - (iii) the Transfer Price.

Such notice shall invite each Continuing Party to state, in writing within fifteen (15) days from the date of such notice (which date shall be specified therein), whether it is willing to purchase any and, if so, how many of the Sale Shares.

- (b) The Sale Shares shall be offered to each Continuing Party (as provided in Clause 12.6(a)) on terms that, in the event of competition, the Sale Shares shall be sold to the Continuing Parties accepting the offer in proportion (as nearly as may be) to their existing holdings of Shares (the "Proportionate Entitlement"). It shall be open to each such Continuing Party to specify if it is willing to purchase Sale Shares in excess of its Proportionate Entitlement ("Excess Shares") and, if the Continuing Party does so specify, he shall state the number of Excess Shares.
- (c) After the expiry of the offers to be made pursuant to Clause 12.6(a) (or sooner if all the Sale Shares offered shall have been accepted in the manner provided in Clause 12.6(a)), the Board shall allocate the Sale Shares in the following manner:
  - (i) if the total amount of the Sale Shares applied for is equal to or less than the available amount of Sale Shares, the Board shall allocate the number applied for in accordance with the applications; or
  - (ii) if the total number of Shares applied for is more than the available number of Sale Shares, each Continuing Party shall be allocated its Proportionate Entitlement (or such lesser number of Sale Shares for which it may have applied); applications for Excess Shares shall be allocated in accordance with such applications or, in the event of competition, (as nearly as may be) to each Continuing Party applying for Excess Shares in the proportion which the amount of Shares held by such Continuing Party bears to the total amount of Shares held by all such Continuing Parties applying for Excess Shares **provided that** such Continuing Party shall not be allocated more Excess Shares than it shall have stated itself willing to take,



and in either case the Company shall forthwith give notice of each such allocation (an "Allocation Notice") to the Seller and each of the persons to whom Sale Shares has been allocated (a "Shareholder Applicant") and shall specify in the Allocation Notice the place and time (being not earlier than five days and not later than 14 days after the date of the Allocation Notice) at which the sale of the Sale Shares shall be completed.

- (d) If, after the expiry of the offers to be made pursuant to Clause 12.6(a), the total amount of the Sale Shares applied for is less than the available amount of Sale Shares, the transfer procedure in this Clause 12.6 shall be deemed to be complete and the Seller shall be entitled to transfer any remaining Sale Shares pursuant to Clause 12.7 provided that the transfer and completion of the sale of the Sale Shares shall not be sanctioned or effected by the Board if such transfer obliges the Seller to procure the making of a Tag-Along Offer and no such offer has been made and (if accepted) completed;
- (e) Subject to Clauses 12.6(f) and (g) below, upon such allocations being made as aforesaid, the Seller shall be bound, on payment of the Transfer Price, to transfer the Sale Shares comprised in the Allocation Notice to the Shareholder Applicants named therein at the time and place therein specified. If the Seller makes default in so doing:
  - (i) the chairman for the time being of the Company (unless such person is a Director appointed by the Seller) or, failing him, one of the Directors (but, for the avoidance of doubt, not a Director appointed by the Seller), or, failing him, some other person duly nominated by a resolution of the Board for that purpose, shall forthwith be deemed to be the duly appointed attorney of the Seller with full power to execute, complete and deliver in the name and on behalf of the Seller all documents necessary to give effect to the transfer of the relevant Sale Shares to the Shareholder Applicants;
  - (ii) the Board and/or any Director may receive and give a good discharge for the purchase money on behalf of the Seller and (subject to the transfer being duly stamped) enter the name of each Shareholder Applicant in the register of members of the Company as the holder by transfer of the Shares comprised in the Sale Shares so purchased by it; and
  - (iii) the Board shall forthwith pay the purchase money into a separate bank account in the Company's name and shall hold such money on trust (but without interest) for the Seller until he shall deliver up his certificate or certificates for the relevant Shares comprising the Sale Shares (or an indemnity, in a form reasonably satisfactory to the Board, in respect of any lost certificate) to the Company when he shall thereupon be paid the purchase money (but without interest).

The appointment referred to in Clause 12.6(e)(i) shall be irrevocable and is given to secure the performance of the obligations of the relevant Shareholder under this Agreement and the Memorandum and Articles.

(f) Any proposed transfer of Sale Shares pursuant to this Clause 12.6 shall be conditional upon the person so acquiring Sale Shares (or part thereof) (the "**Relevant Shareholder**") obtaining all necessary consents, permissions or approvals of any regulatory or supervisory authorities or other persons

in any relevant jurisdiction which are or may be required to enable the Relevant Shareholder lawfully to acquire such Sale Shares (each, a "**Regulatory Consent**") as soon as possible. In the event that any Regulatory Consent is not obtained by a Relevant Member within 120 days of the date of the Allocation Notice (or such longer period as the Board may agree to), the proposed transfer of Sale Shares to that Relevant Shareholder shall not be effective, and the Seller shall give, or shall be deemed to have given, a new Transfer Notice to the Company indicating that it desires to transfer such Sale Shares and the provisions of this Clause 12 shall apply accordingly (provided that such Sale Shares shall not be offered to the Relevant Member).

- (g) If a Seller has given or has been deemed to have given a Transfer Notice then until such time as the offer(s) made pursuant to Clause 12.6(a) have expired in accordance with Clause 12.6(a) or the Transfer Notice is validly withdrawn:
  - (i) the quorum necessary for the transaction of any business of the Directors that relates to the administration of the transfer of any Shares of the Seller shall be any two individuals personally present; and
  - (ii) any Directors nominated by that Seller shall not be entitled to exercise any voting rights in relation to the administration of the transfer of any of its Shares by the Board.

#### 12.7 Seller's right to sell Sale Shares to Third Party Purchaser

If, following the completion of the procedure set out in Clause 12.6, all of the Sale Shares of a Seller has not been purchased, then such Seller may transfer the remaining Sale Shares on a bona fide arm's length sale to the Third Party Purchaser specified in the Transfer Notice (or if the Transfer Notice is deemed to have been given, a Third Party Purchaser) at a price not less than the purchase price specified in the Transfer Notice **provided that**:

- (a) the Shares which comprise such remaining Sale Shares are transferred simultaneously and
- (b) the provisions of Clause 12.3 are satisfied in respect of such Third Party Purchaser.

#### 12.8 Valuation of Shares

If the terms of this Agreement require that the Fair Value of any Shares be determined then the Cash Manager shall, as soon as practicable following the time it becomes apparent that such a valuation is required, apply to the Valuer to obtain a determination from it of the Fair Value of such Shares.

# 12.9 Tag-Along

If any Shareholder (the "**Relevant Transferor**") receives an offer from any person (or persons connected with each other, or persons acting in concert with each other) (the "**Buyer**") for the purchase of all or any Shares held by it and wishes to accept such offer, then the Relevant Transferor shall procure, before the transfer is made and lodged for registration, that the Buyer has made an offer (the "**Tag-Along Offer**") to each holder (other than any holder who is connected with or acting in concert with the Buyer) to purchase the same proportion of the

Shares held by such holder as the proportion being sold by the Relevant Transferor on the same terms and conditions (including as to price) as shall have been agreed between the Relevant Transferor(s) and the Buyer. This Clause 12.9 shall not apply in respect of any transfer made:

- (a) pursuant to Clause 12.4; or
- (b) to a person which is an existing Shareholder at the date of transfer, provided such transfer is made in accordance with the terms of this Agreement.

## 12.10 AerCap transfer prior to the first anniversary

Without prejudice to Clause 4.6, if, prior to the first anniversary of Completion, AerCap transfers any Shares:

- (a) to any party other than a member of the AerCap Group; and
- (b) at a price of less than US\$1.00 per Share,

AerCap shall pay to Waha on demand an amount in US\$ equal to "X", where X is determined as follows:

 $X = (US\$1.00 - SP) \times NS$ 

and where:

"SP" means the price payable per Share (in US\$); and

"NS" means the number of Shares transferred by AerCap.

# 13. TERMINATION EVENTS

- (a) If a Termination Event occurs in relation to a Shareholder (the "Affected Shareholder"), the Affected Shareholder shall be obliged to notify the other Shareholders (the "Non-Affected Shareholders") promptly in writing of such Termination Event. After a Termination Event, any Non-Affected Shareholder may within the period of 20 days from the later of the date of the Termination Event and the date of receipt of the notice from the Affected Shareholder serve a notice on the Affected Shareholder (such notice to be copied to the other Non-Affected Shareholders, if any) requiring that the Affected Shareholder and each of its Related Companies (together, the "Affected Shareholder Group") shall offer to transfer all of its Shares in accordance with Clause 12 (*Transfer of Shares*). Upon service of such a notice by any Non-Affected Shareholder, the Affected Shareholder Group shall immediately be deemed to have given Transfer Notices in accordance with Clause 12.5 (*Transfer Notice*) and the provisions of Clause 12 (*Transfer of Shares*) shall apply.
- (b) If a Transfer Notice is deemed to have been given by a member of the Affected Shareholder Group pursuant to this Clause 13, then such Transfer Notice shall be deemed to have been given:
  - (i) in the case of an event falling within paragraphs (a), (b) or (c) of the definition of "**Termination Event**", immediately prior to the occurrence of such event; and



(ii) in the case of a Material Default which is a Termination Event, the date on which the related Default Termination Notice expires.

# 14. **DEFAULT**

- 14.1 If a Shareholder (the "**Defaulting Shareholder**") commits any breach of this Agreement or any agreement between the Defaulting Shareholder and the Company, any other Significant Shareholder may serve notice upon the Defaulting Shareholder. The notice shall specify the breach and require the Defaulting Shareholder immediately to stop the breach and, to the extent that it is possible, to make good the results of the breach within 30 days. This does not affect the rights of each other Shareholder subsequently to claim damages or other compensation under applicable law for the breach or, where appropriate, to seek an immediate remedy of an injunction, specific performance or similar court order to enforce the Defaulting Shareholder's obligations.
- 14.2 If the Defaulting Shareholder commits a breach of this Agreement which is, in the reasonable opinion of any other Shareholder, material (a "**Material Default**"), the other Party shall be entitled to treat the Material Default as a Termination Event for the purposes of Clause 13 if:
  - (a) the Material Default has not been remedied within the 30 day period referred to in Clause 14.1 and the other party (after such 30 day period) has given not less than 10 days notice (a "**Default Termination Notice**") of its intention to treat the breach as a Termination Event and the material breach remains unremedied at the end of that further 10 day period; or
  - (b) in the event that litigation or arbitration proceedings are commenced after 30 days of receipt by the Defaulting Party of a Default Termination Notice to establish whether or not a Material Default has occurred and do so establish, the breach has not been remedied within 30 days of final resolution of that dispute.

For the avoidance of doubt, any failure by a Defaulting Shareholder to comply with its obligations under Clause 3.5 (*Required Equity Subscriptions*) or the procedures specified in Clause 12 (*Transfer of Shares*) or Clause 27 (*Deadlock*) shall, without limitation, constitute a Material Default.

# 15. TRANSFER TERMS

The transfer by a Seller of any Sale Shares to a Third Party Purchaser under Clause 12 shall be on the following terms:

- (a) such Sale Shares shall be sold free from all liens, charges and encumbrances and third party rights together with all rights of any nature attaching to them including all rights to any dividends, interest or other distributions declared, paid or made after the date of the relevant Transfer Notice;
- (b) such Third Party Purchaser shall assume with effect from the completion date of such transfer, any obligations of such Seller under (and shall procure the release of) any guarantees, indemnities, letters of comfort and/or counter-indemnities to third parties in relation to the business of the Group;
- (c) such Seller shall deliver to such Third Party Purchaser duly executed transfer(s) in favour of such Third Party Purchaser, or as it may direct,

together with, if appropriate, share certificate(s) for the Shares included in the Sale Shares and a certified copy of any authority under which such transfer(s) is/are executed;

- (d) against delivery of the transfer(s), such Third Party Purchaser shall pay the total consideration for such Sale Shares to such Seller by cash for value on the completion date;
- (e) such Seller shall ensure (insofar as it is able) that the relevant transfer or transfers (subject to their being duly stamped, stamp duty (if any) to be paid by such Third Party Purchaser) are registered in the name of such Third Party Purchaser;
- (f) such Seller shall do all such other things and execute all other documents (including any deed) as such Third Party Purchaser may reasonably request to give effect to the sale and purchase of such Sale Shares;
- (g) such Seller shall ensure that all the directors appointed by it to any Group Member resign and the resignation(s) take effect up such transfer without any liability on any Group Member for compensation for loss of office or otherwise; and
- (h) such Third Party Purchaser shall enter into a Deed of Adherence.

#### 16. ASSURANCES

So far as it is legally able, each Shareholder undertakes with the other Shareholder and the Company to exercise all voting rights and powers (direct and indirect) available to it in relation to any person and/or any Group Member to ensure that the provisions of this Agreement (and the other agreements referred to in this Agreement) are completely and punctually fulfilled, observed and performed and generally that full effect is given to the principles set out in this Agreement.

## 17. NON-ASSIGNMENT

No Shareholder shall, nor shall purport to, assign, transfer, charge or otherwise deal with all or any of its rights and/or obligations under this Agreement or grant, declare, create or dispose of any right or interest in it, or sub-contract the performance of any of its obligations under this Agreement in whole or in part (otherwise than pursuant to a transfer of Shares to a third party in accordance with the terms of this Agreement).

#### 18. WAIVER OF RIGHTS

No waiver by a Party of a failure by any other Party to perform any provision of this Agreement operates or is to be construed as a waiver in respect of any other failure whether of a like or different character.

#### 19. **AMENDMENTS**

A variation of this Agreement is valid only if it is in writing and signed by or on behalf of each Party to the Agreement at the time of such variation.

# 20. INVALIDITY

If any provision of this Agreement is or is held to be invalid or unenforceable, then so far as it is invalid or unenforceable it has no effect and is deemed not to be included in this Agreement. This shall not invalidate any of the remaining provisions of this Agreement. The Parties shall then use all reasonable endeavours to replace the invalid or unenforceable provision by a valid provision the effect of which is as close as possible to the intended effect of the invalid or unenforceable provision.

# 21. NO PARTNERSHIP OR AGENCY

- 21.1 Nothing in this Agreement (or any of the arrangements contemplated by it) is or shall be deemed to constitute a partnership between the Parties nor, except as may be expressly set out in it, constitute any Party the agent of any other Party for any purpose.
- 21.2 Unless the Parties agree otherwise in writing none of them shall:
  - (a) enter into contracts with third parties as agent for the Company or for the other Parties; or
  - (b) describe itself as such an agent or in any way hold itself out as being such an agent.

# 22. ANNOUNCEMENTS

- 22.1 No formal public announcement or press release in connection with the signature or subject matter of this Agreement shall (subject to Clause 22.2) be made or issued by or on behalf of any Party without the prior written approval of the other Parties (such approval not to he unreasonably withheld or delayed).
- 22.2 If a Party has an obligation to make or issue any announcement required by law or by any stock exchange or by any Governmental Entity, the relevant Party shall give the other Parties every reasonable opportunity to comment on any announcement or release before it is made or issued (provided that this shall not have the effect of preventing the Parties making the announcement or release from complying with its legal and/or stock exchange obligations.)

# 23. ENTIRE AGREEMENT

- 23.1 This Agreement and the agreements referred to herein set out the entire agreement and understanding between the parties with respect to the subject matter of it.
- 23.2 Neither Party has relied on or has been induced to enter into this Agreement in reliance on any representation, warranty or undertaking which is not expressly set out or referred to in this Agreement or in the Deed of Warranty.
- 23.3 A Party may claim in contract for breach of warranty under this Agreement but no Party shall have any claim or remedy in respect of misrepresentation (whether negligent or otherwise and whether made prior to, and/or in, this Agreement) or untrue statement made by the other Party.
- 23.4 Liability for, or any remedy in respect of, fraudulent misrepresentation is not excluded.

# 24. CONFLICT WITH MEMORANDUM AND ARTICLES

#### 24.1 Supremacy of this Agreement

If the provisions of this Agreement conflict with the Memorandum and Articles or other constitutional documents the provisions of this Agreement shall prevail as between the Parties. The Parties shall:

- (a) exercise all voting and other rights and powers available to them to give effect to the provisions of this Agreement, and
- (b) (if necessary) ensure any required amendment to the Memorandum and Articles or other constitutional document of the Company.

#### 24.2 Transfers of Shares

Without prejudice to the generality of Clause 24.1, the provisions of this Agreement shall prevail in relation to the transfer of Shares.

# 25. **DURATION**

- 25.1 This Agreement shall cease and determine:
  - (a) on the passing of an effective resolution to wind up the Company or the issue of a binding order for the winding up of the Company;
  - (b) in respect of a Shareholder, upon the Shareholder ceasing to be beneficial owner of any Shares provided that, each transferee of such Shares shall have entered into a Deed of Adherence unless it is already a party to this agreement as a "Shareholder";
  - (c) if all the Shares are held by a single Shareholder.
- 25.2 Any cessation and determination pursuant to Clause 25.1 shall:
  - (a) be without prejudice to the rights, obligations or liabilities of any party which shall have accrued or arisen prior to such cessation and determination; and
  - (b) not affect the continued applicability and enforceability of the terms of Clause 8 of this Agreement (*Confidentiality*).

#### 26. NOTICES

- 26.1 Any notice or other formal communication to be given under this Agreement shall be in writing and signed by or on behalf of the party giving it. It shall be:
  - (a) sent by fax to the number set out in Clause 26.2; or
  - (b) delivered by hand or sent by prepaid recorded delivery, special delivery or registered post to the relevant address in Clause 26.2,

and in each case (A) it shall be marked for the attention of the relevant Party set out in Clause 26.2 (or as otherwise notified from time to time under this Agreement) and (B) a copy of the notice shall be sent by email to the email address specified for the relevant Party set out in Clause 26.2 (or as otherwise



notified from time to time under this Agreement). Any notice given by hand delivery, fax or post shall be deemed to have been duly given:

- (i) if hand delivered sent by recorded delivery, special delivery or registered post, the day of delivery when delivered;
- (ii) if sent by fax, when the sender receives a completed transmission sheet or otherwise receives a mechanical or electronic confirmation of transmission

**provided that**, where the delivery or transmission occurs after 6 pm on a Business Day or at any time on a day which is not a Business Day, service shall be deemed to occur at 9 am on the next following Business Day. References to time in this Clause are to local time in the country of the addressee.

#### 26.2 Address of notices

The addresses, fax numbers and email address of the parties for the purpose of Clause 26.1 are:

(a) AerCap

Address: Stationsplein 965, 1117 CE Schiphol Airport, Amsterdam, The Netherlands

Fax No: +31 (20) 655 9100

For the attention of: Erwin den Dikken

with a copies to: edikken@aercap.com and Cole Reese, COO, AerCap Group: creese@aercap.com

(b) Waha

Address: Parkstraat 107, 2514 JH, The Hague, The Netherlands

Fax No: to be advised in writing by Waha

For the attention of: Chief Executive Officer

with a copy to: waha-av-notice@wahacapital.ae

#### (c) The Company

Address: AerCap House, Shannon, County Clare, Ireland

Fax No: +353 61 723850

For the attention of: Company Secretary

with a copies to: ehehir@aercap.com and edikken@aercap.com

# 26.3 Changes to addressees

A Party shall notify the other of a change to its name, relevant addressee, address, facsimile number, telephone number(s) or email address(es) for the

purposes of Clause 26.2 within five Business Days of any change taking place. Such notification shall only be effective on:

- (a) the date specified in the notification as the date on which the change is to take place; or
- (b) if no date is specified or the date specified is less than five Business Days after the date on which notice is given, the date falling five Business Days after notice of any such change has been given.

# 27. **DEADLOCK**

- 27.1 Where:
  - (a) a proposed transaction or course of action by any Group Member requires the consent of the Shareholders pursuant to this Agreement or otherwise and:
    - (i) a Shareholder refuses to provide the consent within ten Business Days of being first asked to do so; and
    - (ii) in the reasonable opinion of any Shareholder which is, or Shareholders which together are, the beneficial owner(s) of 50% or more in nominal value of the Shares or, if the Shares are beneficially owned by two Shareholders equally, either such Shareholder, the inability of the Group Member to proceed with the proposed transaction or course of action has the effect of preventing the Group from continuing to effectively carry on the Business; or
  - (b) the Directors fail to agree at two (2) consecutive Board meetings which have been duly convened, on any matter which, in the reasonable opinion of three (3) or more Directors, has the effect of preventing the Group from continuing to effectively carry on the Business,

any Shareholder may give the other Shareholders written notice (a "Deadlock Notice") to the effect that a deadlock exists.

- 27.2 Within seven (7) Business Days of the service of a Deadlock Notice, each of the Shareholders shall cause its appointees on the Board to prepare and circulate to the other Shareholders and the other Directors a written statement setting out its position or, as the case may be, the position of its appointees on the Board, on the matter in dispute and its reasons for adopting such position (each a "**Position Statement**"). Each Position Statement shall be considered by the Chief Executive Officer of each Shareholder then holding office who shall respectively use their reasonable endeavours to resolve such dispute. If they agree upon a resolution or disposition of the matter, they shall jointly execute a statement setting forth the terms of such resolution or disposition and the Shareholders shall exercise the voting rights and other powers of control available to them in relation to the Group to procure that such resolution or disposition is fully and promptly carried into effect.
- 27.3 If a resolution or disposition is not agreed in accordance with the provisions of Clause 27.2 within five (5) Business Days after delivery of the last of the Position Statements, or such longer period as the Shareholders may agree in writing, then any Shareholder may require that the matter(s) in dispute be the subject of a mediation in accordance with the Model Mediation Procedure and Agreement (the

"Mediation Rules") of the Centre for Effective Dispute Resolution ("CEDR Solve") by serving written notice to this effect on the other Shareholders (a "Mediation Notice").

- 27.4 A mediation under this Agreement (a "**Mediation**") shall be conducted in accordance with the procedure in the Mediation Rules amended to take account of any relevant provisions of this Agreement including, without limitation, the provisions of this Clause 27. If the Shareholders are unable to agree on any such amendment within seven (7) Business Days of the date of the Mediation Notice, such terms shall be decided by CEDR Solve after consultation with the Shareholders.
- 27.5 A Mediation shall commence not later than 15 Business Days after the date of the Mediation Notice.
- 27.6 No party may commence any court proceedings in relation to any matter which is the subject of a Deadlock Notice unless such matter has been the subject of a Mediation and such Mediation has terminated without the conclusion of a binding settlement agreement between the Shareholders resolving the dispute which is the subject of the Deadlock Notice.
- 27.7 Any Mediation Notice served under this Agreement shall be copied to CEDR Solve within two (2) Business Days of service.
- 27.8 Any Mediation shall take place in London and the language of the mediation will be English. The courts of England and Wales have exclusive jurisdiction to hear and decide any suit, action or proceedings, and to settle any disputes, which may arise out of or in connection with any Mediation and any settlement agreement entered into as a result of any Mediation and, for these purposes, each party irrevocably submits to the jurisdiction of the courts of England and Wales.
- 27.9 If a Mediation is terminated without resolution of the matter which was the subject of the relevant Deadlock Notice, for the period of 15 Business Days following such termination (the "**Dissolution Period**") the Shareholders shall attempt to reach agreement on the most appropriate mechanism to dissolve the joint venture between them in a manner satisfactory to the Shareholders, either by the transfer of all of the Shares of one or more of the Shareholders, the transfer of all of the issued Shares or the splitting of the Group's business, assets and liabilities between the Shareholders or on any other basis considered acceptable by the Shareholders.
- 27.10 If no agreement is reached pursuant to Clause 27.9, then the Cash Manager shall, in accordance with Clause 12.8 (Valuation of Shares) request that the Valuer determine the Fair Value of the Shares and the Cash Manager shall promptly following receipt of such valuation notify the Fair Value to the Shareholders. Each Shareholder shall be entitled, within seven (7) Business Days of notification of the Fair Value of the Shares (such seven (7) Business Day period being the "Offer Period"), to make one or more offers to purchase all of the Shares of the other Shareholders (an "Offer") at a specific price per Share (and not a price which is to be determined by a formula or otherwise by reference to any other price or offer) to be stated in a notice in writing to each other Shareholder (the "Offer Notice"). In the event that on the expiry of the Offer Period:
  - (a) any one or more Offers equals or exceeds the Fair Value, the highest Offer shall be deemed to be accepted by each of the Shareholders other than

the Shareholder who made that Offer and the Shareholders shall complete the sale and purchase of the Shares on the earlier of:

- (i) the day thirty (30) Business Days after the notification of the Fair Value of the Shares; and
- (ii) the date on which any consent permission or approval of any regulatory authority required by law for the sale and purchase have been obtained

**provided that** the provisions of Clause 12.5(d) shall apply to transfers of Shares pursuant to such deemed acceptance *mutatis mutandis*; and

- (b) no Offer exceeds the Fair Value, the Shareholders shall be free to accept any Offer which has been made at any time during the period of seven (7) Business Days after the date of the Offer Notice containing that Offer provided that all of the Shareholders (other than the Shareholder who made that Offer) accept such Offer during that period and in that event the Shareholders shall complete the sale and purchase of the Shares on the earlier of:
  - (i) the day 14 Business Days after the date of the Offer Notice; and
  - (ii) the date on which any consent permission or approval of any regulatory authority required by law for the sale and purchase have been obtained,
  - (iii) **provided that** the provisions of Clause 12.5(d) shall apply to transfers of Shares pursuant to such acceptances *mutatis mutandis*.
- 27.11 If no Shareholder makes an offer pursuant to Clause 27.10 any Shareholder may by notice in writing to the other Shareholders require that the Shareholders procure that their appointees on the Board shall, at the earliest practicable date:
  - (a) make or concur in the making of a statutory declaration in the terms mentioned in Section 256 of the Companies Act 1963 (if the state of the Company's affairs admits of the making of such a declaration); and
  - (b) where the state of the Company's affairs enables the making of the declaration referred to in paragraph (a) above convene an extraordinary general meeting of the Company to consider:
    - (i) the matter from which the deadlock arose; and
    - (ii) the passing of a special resolution to place the Company in members' voluntary liquidation,
    - (iii) such meeting or meetings to be held within 5 weeks after the making of any declaration made pursuant to Clause 27.11(a); and
  - (c) where the state of the Company's affairs does not enable the making of the declaration referred to paragraph
    (a) above, convene a meeting of the Company's creditors in accordance with Section 266 of the Companies Act 1963.

27.12 If, at an extraordinary general meeting referred to in Clause 27.11(b), no resolution is carried in relation to the matter from which the deadlock arose by reason of an equality of votes for and against any proposal for dealing with such matter, the Shareholders shall vote in favour of the special resolution for winding up the Company.

# 28. COUNTERPARTS

This Agreement may be executed in any number of counterparts and by the Parties to it on separate counterparts, each of which shall be an original but all of which together shall constitute one and the same instrument.

## 29. **EXPENSES**

Each of AerCap and Waha shall be solely responsible for the costs and expenses incurred by them in connection with the negotiation and settlement of this Agreement (or any of the transactions contemplated herein) and the Memorandum and Articles. The Company shall not be responsible for any costs and expenses incurred in connection with the negotiation and settlement of this Agreement (or any of the transactions contemplated herein) and the Memorandum and settlement of this Agreement (or any of the transactions contemplated herein) and the Memorandum and Articles.

# 30. **RIGHTS OF THIRD PARTIES ACT**

A person who is not a party to this Agreement has no right under the Contracts (*Rights of Third Parties*) Act 1999 (the "**Third Parties Act**") to enforce any term of this Agreement.

## 31. GOVERNING LAW; SUBMISSION TO JURISDICTION

- 31.1 This Agreement, and any non-contractual obligations arising out of or in connection with this Agreement, shall be governed by and construed in accordance with the laws of England and Wales.
- 31.2 The courts of England and Wales have exclusive jurisdiction to hear and decide any suit, action or proceedings, and to settle any disputes, whether relating to a contractual obligation or a non-contractual obligation, which may arise out of or in connection with this Agreement ("**Proceedings**") and, for these purposes, each party irrevocably submits to the non-exclusive jurisdiction of the courts of England and Wales.
- 31.3 Each party irrevocably waives any objection which it might at any time have to the courts of England and Wales being nominated as the forum to hear and decide any Proceedings and agrees not to claim that the courts of Ireland are not a convenient or appropriate forum.
- 31.4 Waha hereby irrevocably authorises and appoints T&H Secretarial Services Limited of Sceptre Court, 40 Tower Hill, London EC3N 4DX (or such other address as may from time to time be notified to the parties) as its authorised agent to accept service of all legal process in England and Wales on its behalf and service on such appointee shall be deemed to be service on Waha as the case may be. Waha agrees that any failure by its process agent to notify it of the legal process shall not invalidate the proceedings concerned. Nothing contained in this Clause 31 affects the right to serve process in another manner permitted by law.
- 31.5 AerCap hereby irrevocably authorises and appoints McCann FitzGerald of St Michael's House, 1 George Yard, Lombard Street, London EC3V 9DF (or such

other address as may from time to time be notified to the parties) as its authorised agent to accept service of all legal process in England and Wales on its behalf and service on such appointee shall be deemed to be service on AerCap as the case may be. AerCap agrees that any failure by its process agent to notify it of the legal process shall not invalidate the proceedings concerned. Nothing contained in this Clause 31 affects the right to serve process in another manner permitted by law.

AS WITNESS this Agreement has been signed by the duly authorised representatives of the parties the day and year first before written.

# **SCHEDULE 1**

#### FORM OF DEED OF ADHERENCE

#### THIS DEED OF ADHERENCE is made the day of

## **BETWEEN:**

- (1) [•] whose contact details appear in the Schedule (the "New Party").
- (2) [All the Parties to the Principal Agreement including any person who has entered into a Deed of Adherence to the Principal Agreement but excluding any Shareholder (other than the Transferor) who has ceased to hold Shares.]

# **RECITALS:**

- (A) Under the terms of a joint venture agreement dated [•] 2009 (the "Principal Agreement") and entered into between
  (1) AerCap AerVenture Holding B.V. (2) [Waha] and (3) AerVenture Limited and [and to which [•] (the "Transferor") is [an original party] [a party by virtue of a Deed of Adherence dated [•]] the Transferor has sold and transferred to the New Party [*insert number and type of Shares*] (the "Transfer Shares") subject to the New Party entering into this Deed of Adherence].
- (B) The New Party wishes to accept the Transfer Shares subject to such condition and to enter into this Deed of Adherence pursuant to Clause 12 (*Transfer of Shares*) of the Principal Agreement.

# IT IS AGREED AS FOLLOWS:

- 1. Expressions defined in the Principal Agreement shall (unless the context otherwise requires) have the same meaning when used in this Deed.
- 2. The New Party hereby undertakes to and covenants with all the Parties to the Principal Agreement (including any person who has entered into a Deed of Adherence pursuant to the Principal Agreement) to comply with the provisions of and to perform all the obligations in the Principal Agreement (save as expressed provided therein) so far as they become due to be observed and performed on or after the date hereof as if the New Party had been an original Party to the Principal Agreement in place of the Transferor [in respect of the Transfer Shares].
- 3. The New Party shall become a Shareholder and the Transferor shall cease to be a Shareholder [in respect of the Transfer Shares] and on and after the date hereof the New Party shall have the benefit of the provisions of the Principal Agreement (save as expressly provided therein) as if the New Party had been an original Party thereto in place of the Transferor [in respect of the Transfer Shares] and the Principal Agreement shall be construed and apply accordingly.
- 4. For the avoidance of doubt, the New Party shall not be entitled to any amount which has fallen due for payment to the Transferor before the date hereof and shall not be liable in respect of any breach or non-performance of the obligations of the Transferor pursuant to the Principal Agreement before the date hereof and the Transferor shall remain entitled to each such amount and shall not be released from any liability in respect of any such breach or non-performance.
- 5. Except as expressly varied by this Deed, the Principal Agreement shall continue in full force and effect, and the Principal Agreement shall be interpreted accordingly.

6. The provisions of Clauses 1 (*Interpretation*), 17 (*Non-Assignment*), 19 (*Amendments*), 20 (*Invalidity*), 22 (*Announcements*), 27 (*Deadlock*), 28 (*Counterparts*) and 31 (*Governing Law; Submission to Jurisdiction*) of the Principal Agreement shall apply to this Deed *mutatis mutandis* as if those provisions had been set out expressly in this Deed, which shall take effect from the date set out above.

IN WITNESS whereof the parties have executed and delivered this document as their deed the day and year first above written.

#### Schedule to Deed of Adherence

#### **Details of New Party**

Name: Registered no. (if a company): Country of incorporation (if a company): Address: Facsimile no.: Other notice details:

[Execution clause]

# **SCHEDULE 2**

# LOAN NOTE INSTRUMENT

# Dated [•]

# AerVenture Limited (as Issuer)

# INSTRUMENT constituting Convertible Redeemable Loan Notes

# McCann FitzGerald

Solicitors Riverside One Sir John Rogerson's Quay Dublin 2 BXG/1968583

# CONTENTS

CLAUSE		PAGE
1.	Interpretation	58
2.	Amount and description of Notes	61
3.	Redemption	61
4.	Interest and Late Payments	62
5.	Certificates and Terms and Conditions	63
6.	Conversion	63
7.	Acceleration Events	64
8.	Covenants	64
9.	Miscellaneous	65
10.	Notices	65
11.	Alteration of this Instrument	66
12.	Governing Law	66
Schedule 1		67
Form of Notes Certificate		67
Schedule 2		69
Register and Transfer		69



**AERVENTURE LIMITED** (registered no. 410443), whose registered office is at AerCap House, Shannon, Co Clare, Ireland (the "Company").

# RECITALS

By resolution of the board of directors of the Company passed on [ $\bullet$ ], the Company has resolved to create and issue loan notes having a nominal value of US\$36,500,000 on the terms of this Instrument.

# NOW THIS INSTRUMENT WITNESSES as follows:

# 1. Interpretation

1.1 Words and phrases defined in the Agreement shall have the same meaning in this Instrument and, in addition, the following expressions shall, unless the context otherwise requires, have the meanings set out opposite them:

"Acceleration Event" means the occurrence of any of the following events:

- (a) any Group Member defaults in the payment of any sums due or payable pursuant to this Instrument and the default continues unremedied for two (2) Business Days; or
- (b) any Group Member defaults in the performance of any other obligation, covenant, term or condition contained herein and (if such default is capable of remedy) the default continues unremedied for ten (10) Business Days; or
- (c) any Group Member becomes unable or admits inability to pay its debts as they fall due (or is deemed to be unable to pay its debts within the meaning of Section 214 of the Companies Act 1963 or any statutory modification or re enactment thereof) or if any Group Member commences negotiations to re-schedule the whole or any part of its indebtedness which it would or might otherwise be unable to pay when due; or
- (d) a petition is presented (and not withdrawn within two (2) Business Days) or an order is made or an effective resolution is passed or legal proceedings are started for the winding up or dissolution of any Group Member other than for the purposes of reconstruction or amalgamation on terms previously approved in writing by the Noteholder; or
- (e) any Group Member convenes a meeting of, or proposes to enter into, any arrangement or composition for the benefit of its creditors; or
- (f) any judgment or order is made against any Group Member which, if appealable is not stayed pending appeal, and remains unsatisfied for fourteen (14) days from its date or a distress, execution, sequestration or other legal process is levied, issued or enforced against any of the property of any Group Member and is not discharged within fourteen (14) days; or
- (g) an encumbrancer takes possession or indicates an intention to take possession or a receiver or other similar officer is appointed in respect of



the whole or any material part of the undertaking or assets of any Group Member; or

- (h) an application is made or a petition is presented to the High Court or Circuit Court of Ireland or other steps are taken for the appointment of an examiner to any Group Member or if an examiner is otherwise appointed to any Group Member or if any Group Member is placed under the protection of the Court; or
- (i) any Group Member stops or threatens to stop payment of its debts or ceases or threatens to cease to carry on its business or any part thereof material to any Group Member; or
- (j) any governmental or other licence, approval, authorisation, consent or exemption required to enable any Group Member to carry on the Business or to perform any of its obligations under this Instrument lapses, ceases to apply or is withdrawn or is modified; or
- (k) it becomes unlawful for any Group Member to perform any of its obligations hereunder;

"Acts" means the Companies Acts, 1963 to 2006;

"Agreement" means the Agreement dated 21 June 2009 between (1) AerCap AerVenture Holding B.V., (2) Waha Capital PJSC and (3) AerVenture Limited as the same has been amended and acceded to from time to time;

"Certificate" means a certificate for Notes in or substantially in the form set out in Schedule 1;

"Conversion Date" means, in respect of any Notes, the date on which such Notes are converted into Shares in accordance with this Instrument;

"Conversion Notice" has the meaning given in Clause 6.2;

"Conversion Rate" means one Share for each US\$1 of principal (but not accrued interest) represented by a Note, provided that where a Share Reorganisation takes place, such rate shall be adjusted, with effect from the date the Share Reorganisation takes place, in the manner which the auditors of the Company determine is necessary so that the percentage of the issued share capital of the Company into which the then outstanding Notes are capable of being converted after the Share Reorganisation is the same as the percentage of the issued share capital of the Company into which the then outstanding Notes were capable of being converted into before the Share Reorganisation;

"dollar" and "US\$" means the lawful currency for the time being of the United States of America;

"Effective Date" means the date on which the Notes are issued;

"Final Repayment Date" means 30 June 2015;

"Instrument" means this instrument as from time to time modified in accordance with the provisions to this Instrument including any supplemental instrument executed in accordance with the provisions contained in this Instrument;

"Interest Payment Date" means the dates falling ten (10) Business Days after 31 March, 30 June, 30 September and 31 December in each year;

"Interest Rate" means 20 per cent. per annum;

"Notes" means the notes constituted by this Instrument or any reference to a "Note" is a reference to any one of the issued Notes;

"Noteholders" means the holder or several holders of Notes for the time being entered on the Register;

"Ordinary Shares" means the ordinary shares of US\$1 in the capital of the Company;

"**Redemption Date**" means, in respect of any Notes, the date on which such Notes are redeemed in accordance with this Instrument;

"Register" means the register of holders of the Notes maintained pursuant to paragraph 1 of Schedule 2; and

"Share Reorganisation" means an event pursuant to which the share capital of the Company is the subject of any subdivision, consolidation or any analogous variation.

- 1.2 In this Instrument, unless the context requires otherwise:
  - (a) a reference to a person (including a party to this Instrument) includes a reference to that person's legal personal representatives, successors and permitted assigns;
  - (b) a reference to a document is a reference to that document as from time to time supplemented or varied;
  - (c) any reference in this Instrument and/or in the Schedules to any statute or statutory provision shall be deemed to include any statute or statutory provision which amends, extends, consolidates, re-enacts or replaces same, or which has been amended, extended, consolidated, re-enacted or replaced (whether before or after the date of this Instrument) by same and shall include any orders, regulations, instruments or other subordinate legislation made under the relevant statute;
  - (d) words importing the singular shall include the plural number and vice versa and words importing a gender shall include each gender;
  - (e) words and phrases the definitions of which are contained or referred to in the Companies Acts shall be construed as having the meanings thereby attributed to them;
  - (f) any reference to any Clause, sub-Clause, paragraph, or Schedule shall be a reference to the Clause, sub-Clause, paragraph, or Schedule of this Instrument in which the reference occurs unless it is indicated that reference to some other provision is intended;
  - (g) the provisions of the Schedules to this Instrument shall form an integral part of this Instrument and shall have as full effect as if they were incorporated in the body of this Instrument and the expressions "this

Instrument" and "the Instrument" shall be deemed to include the Schedules to this Instrument;

- (h) any reference to a "person" shall be construed as a reference to any individual, firm, company, corporation, undertaking, government, state or agency of a state, or any association or partnership (whether or not having separate legal personality);
- (i) the headings contained in this Instrument and the Schedules are inserted for convenience of reference only and shall not in any way form part of nor affect nor be taken into account in the construction or interpretation of any provisions of this Instrument or the said Schedules; and
- (j) all references in this Agreement to costs, charges and expenses include any value added tax or similar tax charged or chargeable in respect thereof.

#### 2. Amount and description of Notes

- 2.1 The principal amount of the Notes is US\$36,500,000. The Notes shall be issued and delivered in denominations and integral multiples of US\$1 in nominal amount, subject to and with the benefit of the provisions of this Instrument. All the obligations and covenants contained in this Instrument shall be binding on the Company and the Noteholder and all persons claiming through either of them. The Notes shall be issued in accordance with the Agreement.
- 2.2 The Notes, as and when issued, shall constitute a direct and unconditional obligation of the Company in accordance with the terms of this Instrument and will rank pari passu equally and rateably without discrimination or preference as unsecured obligations of the Company and with all other unsecured indebtedness of the Company, except to the extent provided by law.
- 2.3 On receipt of a request by the Noteholder, the Company shall as soon as practicable deliver to the Noteholder a certificate from the Company containing such information regarding the principal outstanding on the Notes as the Noteholder may from time to time require. The Noteholder shall be entitled to rely on such certificates for all purposes.
- 2.4 On conversion and/or redemption of all the Notes, this Instrument shall terminate and be of no further force or effect.

#### 3. **Redemption**

- 3.1 The Company covenants that, unless the Notes have been converted into Shares in accordance with clause 6 of this Instrument, it will on the Final Redemption Date, or on such earlier date as the amounts outstanding under the Notes become due and payable in accordance with clause 7, pay or procure to be paid to each Noteholder all outstanding amounts in respect of the Notes held by that Noteholder on that date together with all interest accrued (calculated in accordance with clause 4) up to but excluding the Final Redemption Date.
- 3.2 At any time the Company may redeem some or all of the Notes at par, together with all interest accrued thereon (calculated in accordance with clause 4) up to but excluding the date of redemption, by giving each of the Noteholders not less than 10 Business Days' written notice (a "**Pre-Payment Notice**") to this effect. Such notice shall specify:

- (a) the nominal amount of Notes held by that Noteholder which are to be redeemed; and
- (b) the proposed date of redemption (the "**Pre-Payment Date**"),

and shall request that the Noteholder deliver the Certificate for such Notes to the Company in accordance with clause 5. On the Pre-Payment Date, the Company shall redeem the relevant Notes save to the extent that such Notes have been converted into Shares pursuant to Clause 6 prior to the Pre-Payment Date.

- 3.3 The Company may only elect to redeem Notes pursuant to clause 3.2 if it elects to redeem the same percentage of the Notes held by each Noteholder (rounded to the nearest US\$1 of principal).
- 3.4 All Notes repaid or purchased by the Company shall be cancelled and shall not be available for reissue.
- 3.5 On receipt of a request by a Noteholder, the Company shall as soon as practicable deliver to the Noteholder a certificate from the Company containing such information regarding the principal and/or interest outstanding on the Notes as the Noteholder may from time to time require. The Noteholder shall be entitled to rely on such certificates for all purposes.

# 4. Interest and Late Payments

- 4.1 Interest on the Notes will accrue at the Interest Rate on a daily basis, be calculated on the basis of the actual number of days elapsed and a 360 day year and is payable, in respect of each Note:
  - (a) on each Interest Payment Date; and
  - (b) on each date on which redemption of such Note is due.
- 4.2 If any amount due to the Noteholder (whether interest or any amount payable following the redemption of Notes) is not paid for any reason within 14 days of the date on which payment is due in accordance with this Instrument, the overdue amount shall bear interest (in addition to the Interest Rate) at the rate of two per cent. (2%) per annum, accruing and compounding on a daily basis and such additional interest will be paid in accordance with Clause 4.1.
- 4.3 Upon conversion of a Note into Shares in accordance with this Instrument:
  - (a) all interest on the Note shall cease to accrue; and
  - (b) the Company shall pay to the Noteholder (the "**Converting Noteholder**") the amount of interest accrued on the Notes which are being converted up until the relevant Conversion Date save that, where the Company certifies in writing to the Converting Noteholder prior to the Conversion Date that the Company does not have sufficient cash to pay such interest, the Converting Noteholder may elect in writing either:
    - (i) to convert the amount of accrued interest into Shares on the Conversion Date and the provisions of Clause 6 shall apply accordingly; or
    - (ii) to require that the Company issue to it additional Notes with a nominal amount equal to the amount of accrued interest. Where



the Converting Noteholder so elects, the Company shall take all action necessary to issue such additional Notes (including, where necessary, by amending this Instrument so as to increase the principal amount of the Notes).

## 5. Certificates and Terms and Conditions

- 5.1 The certificates for the Notes shall be in the form or substantially in the form set out in Schedule 1 and shall be issued under the seal of the Company affixed with the authority of the directors and in the manner provided in the Articles.
- 5.2 The Company shall comply with the terms and conditions of the Certificates and of this Instrument and the Notes shall be held subject to and with the benefit of those terms and conditions all of which shall be binding on the Company and the Noteholder and all persons claiming through or under them respectively.
- 5.3 The Noteholder shall be entitled to receive a Certificate stating the nominal amount of the Notes held by it and their subscription price but so that joint holders shall be entitled to one Certificate only in respect of the Notes held jointly by them which Certificate shall be delivered to the joint holder whose name stands first in the Register in respect of the joint holding or to such other person as the joint holders may, in writing, direct. Delivery of a certificate to one of such persons shall be sufficient delivery to all. The Company shall not be bound to register more than four persons as the joint holders of any Note.
- 5.4 If any Certificate is defaced or worn out, then, upon production of such Certificate to the directors of the Company they may cancel the Certificate and issue a Certificate in lieu thereof, and if any such Certificate is lost or destroyed, then, upon proof of loss or destruction being given to the satisfaction of the directors of the Company, and, in the case of a lost Certificate or in default of proof of destruction of a Certificate, on such terms as to evidence and indemnity as the directors of the Company may deem adequate, a new Certificate in lieu thereof may be given to the person entitled to the lost or destroyed Certificate. An entry as to the issue of the new Certificate and indemnity (if any) shall be made in the Register.
- 5.5 Notes shall only be redeemed against surrender of the relevant Certificate(s) for cancellation. If the Noteholder fails to deliver the Certificate for his Notes on redemption, the Company may retain the redemption moneys until delivery of the Certificate (or of an indemnity in respect of the Certificate in a form reasonably satisfactory to the Company).

#### 6. Conversion

- 6.1 Subject to the remainder of this clause 6, at any time prior to the Final Repayment Date, a Noteholder may require the Company to allot fully paid Shares in exchange for and in satisfaction of some or all of the Notes held by it at the Conversion Rate. This right may be exercised by the Noteholder serving notice in writing on the Company (a "Conversion Notice"), which shall specify the nominal amount of the Notes which are to be converted.
- 6.2 A Noteholder shall not be entitled to the allotment of a fraction of a Share and any amount of a Note which, but for this clause 6.2, would be converted into a fraction of a Share shall continue to be outstanding unless it would remain as the only outstanding amount of a Note held by the Noteholder submitting the Conversion Notice, in which case such outstanding amount of a Note shall be

cancelled and the Noteholder shall have no claim against the Company in respect of the amount so cancelled.

- 6.3 A Noteholder shall only be entitled to exercise the right contemplated by clause 6.1 where the nominal amount of the Notes to be converted by that Noteholder is at least US\$100,000 or the entire holding of Notes of the Noteholder (whichever is the lesser amount).
- 6.4 To be effective, a Conversion Notice shall be deposited at the registered office of the Company and shall be executed by the Noteholder and be accompanied by the relevant Certificate and such other documents (including any power of attorney) as the Company may reasonably require to enable the Company to establish that the Conversion Notice has been duly executed by the Noteholder.
- 6.5 The Company shall, within five Business Days of each Conversion Date, allot to the Noteholder the amount of Shares credited as fully paid to which he shall be entitled by virtue of the exercise of such rights and shall as soon as practicable thereafter enter or procure the entry of the name of the Noteholder in the register of members of the Company in respect of such Shares. The allotment and issue of Shares pursuant to clause 6 shall be in full satisfaction and discharge of the principal monies equivalent to the nominal amount of the Notes so converted.
- 6.6 All Shares issued in respect of Notes converted pursuant to clause 6 shall be credited as fully paid and will rank for any dividends or distributions declared after the Conversion Date, whether by reference to accounts drawn up to a date in any financial year or other period ended prior to, or on or after the relevant Conversion Date and will otherwise rank pari passu with the Shares in issue at that date.
- 6.7 Any issue of Ordinary Shares pursuant to this Clause 6 shall be conditional upon the Noteholder obtaining all necessary consents, permissions or approvals of any regulatory or supervisory authorities in any relevant jurisdiction which are or may be required to enable that Noteholder to lawfully acquire such Ordinary Shares.

#### 7. Acceleration Events

- 7.1 The occurrence of any Acceleration Event shall at the option of any of the Noteholders (exercised by written notice to the Company) cause the Notes to be, and they shall immediately become, due and repayable at their principal amount together with all accrued interest and all other amounts due and payable in respect of the Notes pursuant to this Instrument.
- 7.2 The Noteholders may, at any time after the Notes shall have become due and payable in accordance with clause 7.1, and without notice:
  - (a) take such proceedings and/or other action against the Company to enforce their obligations under this Instrument; and
  - (b) exercise such powers and discretions as are given to them by this Instrument and/or the Agreement.

# 8. Covenants

The Company covenants that, for so long as any of the Notes remain outstanding, the Company shall, save as otherwise directed in writing by the Noteholder:



- (a) at its own cost execute, acknowledge, deliver, file and register all such additional instruments, certificates, documents and assurances and perform such other acts or things as the Noteholder may reasonably request to give effect to the purposes of this Instrument or the transactions contemplated by this Instrument or to ensure the performance of the obligations imposed on it or to protect the rights of the Noteholder under this Instrument and the Notes;
- (b) ensure that the authorised and issued share capital of the Company shall be maintained at a level sufficient so that the Company is at all times able to comply with its obligations under this Instrument with respect to the issue of Shares upon conversion of the Notes; and
- (c) give notice in accordance with the terms of this Instrument to Noteholders of the occurrence or likely occurrence of any Acceleration Event immediately upon becoming aware of it.

#### 9. Miscellaneous

- 9.1 The Notes shall be subject to the provisions of Schedule 2.
- 9.2 The Company will pay all stamp duties, capital duties and other similar duties on or in connection with the execution, maintenance or enforcement of this Instrument.

#### 10. Notices

- 10.1 Notices or other communications given pursuant to this Instrument shall be in writing and shall be sufficiently given:
  - (a) if delivered by hand or sent by post to the address and for the attention of the person set forth in this Clause of the party to which the notice or communication is being given or to such other address and for the attention of such other person as such party shall communicate to the party giving the notice or communication; or
  - (b) if sent by facsimile to the correct facsimile number of the party to which it is being sent.
- 10.2 Any notice, or communication, given or sent by post under this Clause shall be sent by ordinary post.
- 10.3 Every notice or communication given in accordance with this Clause shall be deemed to have been received as follows:

Means of Dispatch	Deemed Received
Delivery by hand:	the day of delivery;
Post:	one Business Day after posting; and
Facsimile	when the sender receives a completed transmission sheet or otherwise receives a mechanical confirmation of transmission

Provided that if, in accordance with the above provisions, any such notice or other communication would otherwise be deemed to be given or made outside working hours (being 9 a.m. to 5 p.m. on a Business Day) such notice or other communication shall be deemed to be given or made at the start of working hours on the next Business Day.

10.4 The relevant addressee, address and facsimile number for the purposes of this Instrument are:

Name of Party Address/Fax no

The Company AerCap House Shannon Co Clare, Ireland Facsimile no: +353 61 723 850 Attention: The Company Secretary

The Noteholder

As specified in the Register

# 11. Alteration of this Instrument

The provisions of this Instrument and the conditions on which the Notes are held shall not, without the prior approval in writing of the Company and the written approval of the Noteholder, be altered, abrogated or added to.

#### 12. Governing Law

- 12.1 This Instrument and the Notes shall be governed by and construed in accordance with the laws of England.
- 12.2 The courts of England have non-exclusive jurisdiction to hear and decide any suit, action or proceedings, and to settle any disputes, which may arise out of or in connection with this Instrument ("**Proceedings**") and, for these purposes, each party irrevocably submits to the jurisdiction of the courts of England and waives any objection which it might at any time have to the courts of England being nominated as the forum to hear and decide any Proceedings and agrees not to claim that the courts of England are not a convenient or appropriate forum.

IN WITNESS WHEREOF this Instrument has been executed the day and year first before written.

# Schedule 1

# Form of Notes Certificate

Certificate No [ • ]

Nominal amount of Notes US\$•.

AerVenture Limited Registered Number 410443

Issue of Notes approved by a Resolution of the Board of Directors of the Company passed on [ ] 200[ ].

# THIS IS TO CERTIFY that [•]

is the registered holder(s) of US\$[ ] nominal of the above mentioned Notes which are constituted by an Instrument entered into by the Company on and dated [] 200[] (the "**Instrument**") and are issued subject to the provisions (including the conditions) therein contained.

Interest at a rate of 20% per annum is payable on the Notes represented by this Certificate in accordance with the terms of the Instrument.

Without prejudice to the generality of the foregoing, the amount outstanding under the Notes shall be determined in accordance with the Instrument and payments of such principal outstanding shall be made subject to the terms of, and when contemplated by, the Instrument.

Given under the Common Seal of the Company on • .

**PRESENT** when the common seal of **AERVENTURE LIMITED** was hereunto affixed in the presence of:

# NOTES:

Where the context so admits, words and expressions defined in the Instrument shall bear the same respective meanings in the conditions endorsed hereon.

The Notes are redeemable, convertible and transferable in accordance with the terms and conditions contained in the Instrument a copy of which is available from the Company. No transfer of the Notes will be accepted for registration unless accompanied by this certificate and lodged at the registered office of The Company.

A copy of the Instrument is available for inspection at the registered office of the Company at all reasonable times during office hours and will be sent as soon as possible on request to any holder of the Notes or any part thereof.

#### Schedule 2

#### **Register and Transfer**

#### 1. The Register

- 1.1 A register of the Notes will be kept by the Company at its registered office. There shall be entered in the Register:
  - (a) the names and addresses of the holder for the time being of the Notes;
  - (b) the amount of the Notes held by the registered holder;
  - (c) the date upon which the name of the registered holder is entered in the Register in respect of the Notes standing in his name; and
  - (d) the serial number of each Certificate issued.
- 1.2 Any change of the name or address of the holder of Notes shall be notified to the Company promptly and the Register shall be altered accordingly.
- 1.3 The holder of Notes shall be at liberty at all reasonable times during office hours to inspect the Register.

#### 2. **Register to be conclusive**

Except as required by law, the Company will recognise the registered holder of the Notes as the absolute owner thereof and shall not be bound to take notice or see to the execution of any trust (whether express implied or constructive) to which any Notes may be subject.

#### 3. Set-off

The holder of the Notes will be recognised by the Company as entitled to the Notes free from any equity, set-off or cross-claim on the part of the Company against the original or any intermediate holder of the Notes.

#### 4. Transfer of Notes

- 4.1 The provisions of clause 12 of the Agreement shall apply to transfers of Notes as if, and on the basis that:-
  - (a) such Notes constitute "Shares"; and
  - (b) the Noteholder proposing to transfer the Notes constitutes a "Shareholder",

for the purposes of the Agreement, and no transfer of any Notes shall be made or registered unless such transfer complies with the Agreement on this basis.

4.2 A Noteholder shall only be entitled to transfer Notes in multiples of US\$1,000 unless the entire holding of Notes of the Noteholder is less than US\$1,000, in which case the Noteholder may transfer that entire holding.

- 4.3 Every transfer of the Notes shall be made by an instrument of transfer in the usual or common form or in any other form which may be approved by the Directors. The instrument of transfer of the Notes shall be executed by or on behalf of the transferor but need not be executed by or on behalf of the transfere. The transferor shall be deemed to remain the holder of the Notes until the name of the transferee is entered in the Register in respect of the Notes being transferred.
- 4.4 The Directors may decline to recognise any instrument of transfer of the Notes otherwise permitted by this Instrument unless the instrument is deposited at the registered office of the Company accompanied by the Certificate for the Notes, and such other evidence as the Directors may reasonably require to show the right of the transferor to make the transfer. The Directors may waive production of any Certificate upon production to them of satisfactory evidence of the loss or destruction of the Certificate together with such indemnity as they may reasonably require.
- 4.5 No fee shall be charged for any registration of a transfer of the Notes or for the registration of any other documents which in the opinion of the Directors require registration.
- 4.6 The registration of a transfer shall be conclusive evidence of the approval by the Directors of such a transfer.
- 4.7 No Notes shall be issued or transferred to a person who is not already a party to the Agreement unless that person has executed a deed of adherence in the form or substantially in the form as set out in the Agreement.

#### 5. Payments

- 5.1 Any moneys payable on or in respect of any Notes shall be paid by bank draft or telegraphic transfer to such account as shall be nominated by the Noteholder.
- 5.2 If several persons are entered in the Register as joint holders of the Notes then, without prejudice to paragraph 5.1 above, the receipt or payment to the order of any one of them for any moneys payable on or in respect of the Notes shall be as effective a discharge to the Company as if the person to whose order the payment is made were the sole registered holder of the Notes in question.
- 5.3 All payments of moneys payable on or in respect of the Notes will be made gross, free and clear of all right of counterclaim or set off and without deductions or withholdings unless such deductions or withholdings are required by law, in which event the Company shall pay such additional amount as shall be required to ensure that the net amount received by the Noteholder will equal the full amount which would have been received by it had no such deduction or withholding been required to be made.

#### 6. No fee for registration

No fee will be charged for the registration of any transfer, power of attorney or other document relating to or affecting the title to the Notes.

Director

Director/Secretary

# **SCHEDULE 3**

## **15% LOAN NOTE INSTRUMENT**

# Dated [•]

#### AerVenture Limited (as Issuer)

#### INSTRUMENT constituting Redeemable Loan Notes

# McCann FitzGerald

Solicitors Riverside One Sir John Rogerson's Quay Dublin 2 BXG/1976973

# CONTENTS

CLAU	PAGE	
1.	Interpretation	75
2.	Amount and description of Notes	78
3.	Redemption	78
4.	Interest and Late Payments	79
5.	Certificates and Terms and Conditions	79
6.	Acceleration Events	80
7.	Covenants	80
8.	Miscellaneous	80
9.	Notices	80
10.	Alteration of this Instrument	81
11.	Governing Law	81
Schedule 1		
Form of Notes Certificate		
Schedule 2		
Register and Transfer		
	74	

**AERVENTURE LIMITED** (registered no. 410443), whose registered office is at AerCap House, Shannon, Co Clare, Ireland (the "Company").

#### RECITALS

By resolution of the board of directors of the Company passed on  $[\bullet]$ , the Company has resolved to create and issue loan notes having a nominal value of US\$[•] on the terms of this Instrument.

#### NOW THIS INSTRUMENT WITNESSES as follows:

#### 1. Interpretation

1.1 Words and phrases defined in the Agreement shall have the same meaning in this Instrument and, in addition, the following expressions shall, unless the context otherwise requires, have the meanings set out opposite them:

"Acceleration Event" means the occurrence of any of the following events:

- (a) any Group Member defaults in the payment of any sums due or payable pursuant to this Instrument and the default continues unremedied for two (2) Business Days; or
- (b) any Group Member defaults in the performance of any other obligation, covenant, term or condition contained herein and (if such default is capable of remedy) the default continues unremedied for ten (10) Business Days; or
- (c) any Group Member becomes unable or admits inability to pay its debts as they fall due (or is deemed to be unable to pay its debts within the meaning of Section 214 of the Companies Act 1963 or any statutory modification or re enactment thereof) or if any Group Member commences negotiations to re-schedule the whole or any part of its indebtedness which it would or might otherwise be unable to pay when due; or
- (d) a petition is presented (and not withdrawn within two (2) Business Days) or an order is made or an effective resolution is passed or legal proceedings are started for the winding up or dissolution of any Group Member other than for the purposes of reconstruction or amalgamation on terms previously approved in writing by the Noteholder; or
- (e) any Group Member convenes a meeting of, or proposes to enter into, any arrangement or composition for the benefit of its creditors; or
- (f) any judgment or order is made against any Group Member which, if appealable is not stayed pending appeal, and remains unsatisfied for fourteen (14) days from its date or a distress, execution, sequestration or other legal process is levied, issued or enforced against any of the property of any Group Member and is not discharged within fourteen (14) days; or
- (g) an encumbrancer takes possession or indicates an intention to take possession or a receiver or other similar officer is appointed in respect of



the whole or any material part of the undertaking or assets of any Group Member; or

- (h) an application is made or a petition is presented to the High Court or Circuit Court of Ireland or other steps are taken for the appointment of an examiner to any Group Member or if an examiner is otherwise appointed to any Group Member or if any Group Member is placed under the protection of the Court; or
- (i) any Group Member stops or threatens to stop payment of its debts or ceases or threatens to cease to carry on its business or any part thereof material to any Group Member; or
- (j) any governmental or other licence, approval, authorisation, consent or exemption required to enable any Group Member to carry on the Business or to perform any of its obligations under this Instrument lapses, ceases to apply or is withdrawn or is modified; or
- (k) it becomes unlawful for any Group Member to perform any of its obligations hereunder;

"Acts" means the Companies Acts, 1963 to 2006;

"Agreement" means the Agreement dated 21 June 2009 between (1) AerCap AerVenture Holding B.V., (2) Waha Capital PJSC and (3) AerVenture Limited as the same has been amended and acceded to from time to time;

"Certificate" means a certificate for Notes in or substantially in the form set out in Schedule 1;

"dollar" and "US\$" means the lawful currency for the time being of the United States of America;

"Effective Date" means the date on which the Notes are issued;

"Final Repayment Date" means 30 June 2015;

"Instrument" means this instrument as from time to time modified in accordance with the provisions to this Instrument including any supplemental instrument executed in accordance with the provisions contained in this Instrument;

"Interest Payment Date" means the dates falling ten (10) Business Days after 31 March, 30 June, 30 September and 31 December in each year;

"Interest Rate" means 15 per cent. per annum;

"Notes" means the notes constituted by this Instrument or any reference to a "Note" is a reference to any one of the issued Notes;

"Noteholders" means the holder or several holders of Notes for the time being entered on the Register;

"Ordinary Shares" means the ordinary shares of US\$1 in the capital of the Company;

"**Redemption Date**" means, in respect of any Notes, the date on which such Notes are redeemed in accordance with this Instrument;

"Register" means the register of holders of the Notes maintained pursuant to paragraph 1 of Schedule 2; and

"Share Reorganisation" means an event pursuant to which the share capital of the Company is the subject of any subdivision, consolidation or any analogous variation.

- 1.2 In this Instrument, unless the context requires otherwise:
  - (a) a reference to a person (including a party to this Instrument) includes a reference to that person's legal personal representatives, successors and permitted assigns;
  - (b) a reference to a document is a reference to that document as from time to time supplemented or varied;
  - (c) any reference in this Instrument and/or in the Schedules to any statute or statutory provision shall be deemed to include any statute or statutory provision which amends, extends, consolidates, re-enacts or replaces same, or which has been amended, extended, consolidated, re-enacted or replaced (whether before or after the date of this Instrument) by same and shall include any orders, regulations, instruments or other subordinate legislation made under the relevant statute;
  - (d) words importing the singular shall include the plural number and vice versa and words importing a gender shall include each gender;
  - (e) words and phrases the definitions of which are contained or referred to in the Companies Acts shall be construed as having the meanings thereby attributed to them;
  - (f) any reference to any Clause, sub-Clause, paragraph, or Schedule shall be a reference to the Clause, sub-Clause, paragraph, or Schedule of this Instrument in which the reference occurs unless it is indicated that reference to some other provision is intended;
  - (g) the provisions of the Schedules to this Instrument shall form an integral part of this Instrument and shall have as full effect as if they were incorporated in the body of this Instrument and the expressions "this Instrument" and "the Instrument" shall be deemed to include the Schedules to this Instrument;
  - (h) any reference to a "person" shall be construed as a reference to any individual, firm, company, corporation, undertaking, government, state or agency of a state, or any association or partnership (whether or not having separate legal personality);
  - (i) the headings contained in this Instrument and the Schedules are inserted for convenience of reference only and shall not in any way form part of nor affect nor be taken into account in the construction or interpretation of any provisions of this Instrument or the said Schedules; and

(j) all references in this Agreement to costs, charges and expenses include any value added tax or similar tax charged or chargeable in respect thereof.

#### 2. Amount and description of Notes

- 2.1 The principal amount of the Notes is US\$[•]. The Notes shall be issued and delivered in denominations and integral multiples of US\$1 in nominal amount, subject to and with the benefit of the provisions of this Instrument. All the obligations and covenants contained in this Instrument shall be binding on the Company and the Noteholder and all persons claiming through either of them. The Notes shall be issued in accordance with the Agreement.
- 2.2 The Notes, as and when issued, shall constitute a direct and unconditional obligation of the Company in accordance with the terms of this Instrument and will rank pari passu equally and rateably without discrimination or preference as unsecured obligations of the Company and with all other unsecured indebtedness of the Company, except to the extent provided by law.
- 2.3 On receipt of a request by the Noteholder, the Company shall as soon as practicable deliver to the Noteholder a certificate from the Company containing such information regarding the principal outstanding on the Notes as the Noteholder may from time to time require. The Noteholder shall be entitled to rely on such certificates for all purposes.
- 2.4 On redemption of all the Notes, this Instrument shall terminate and be of no further force or effect.

#### 3. Redemption

- 3.1 The Company covenants that, it will on the Final Redemption Date, or on such earlier date as the amounts outstanding under the Notes become due and payable in accordance with Clause 6, pay or procure to be paid to each Noteholder all outstanding amounts in respect of the Notes held by that Noteholder on that date together with all interest accrued (calculated in accordance with Clause 4) up to but excluding the Final Redemption Date.
- 3.2 At any time the Company may redeem some or all of the Notes at par, together with all interest accrued thereon (calculated in accordance with Clause 4) up to but excluding the date of redemption, by giving each of the Noteholders not less than 10 Business Days' written notice (a "**Pre-Payment Notice**") to this effect. Such notice shall specify:
  - (a) the nominal amount of Notes held by that Noteholder which are to be redeemed; and
  - (b) the proposed date of redemption (the "**Pre-Payment Date**"),

and shall request that the Noteholder deliver the Certificate for such Notes to the Company in accordance with Clause 5. On the Pre-Payment Date, the Company shall redeem the relevant Notes.

- 3.3 The Company may only elect to redeem Notes pursuant to Clause 3.2 if it elects to redeem the same percentage of the Notes held by each Noteholder (rounded to the nearest US\$1 of principal).
  - 78

- 3.4 All Notes repaid or purchased by the Company shall be cancelled and shall not be available for reissue.
- 3.5 On receipt of a request by a Noteholder, the Company shall as soon as practicable deliver to the Noteholder a certificate from the Company containing such information regarding the principal and/or interest outstanding on the Notes as the Noteholder may from time to time require. The Noteholder shall be entitled to rely on such certificates for all purposes.

#### 4. Interest and Late Payments

- 4.1 Interest on the Notes will accrue at the Interest Rate on a daily basis, be calculated on the basis of the actual number of days elapsed and a 360 day year and is payable, in respect of each Note:
  - (a) on each Interest Payment Date; and
  - (b) on each date on which redemption of such Note is due.
- 4.2 If any amount due to the Noteholder (whether interest or any amount payable following the redemption of Notes) is not paid for any reason within 14 days of the date on which payment is due in accordance with this Instrument, the overdue amount shall bear interest (in addition to the Interest Rate) at the rate of two per cent. (2%) per annum, accruing and compounding on a daily basis and such additional interest will be paid in accordance with Clause 4.1.

#### 5. Certificates and Terms and Conditions

- 5.1 The certificates for the Notes shall be in the form or substantially in the form set out in Schedule 1 and shall be issued under the seal of the Company affixed with the authority of the directors and in the manner provided in the Articles.
- 5.2 The Company shall comply with the terms and conditions of the Certificates and of this Instrument and the Notes shall be held subject to and with the benefit of those terms and conditions all of which shall be binding on the Company and the Noteholder and all persons claiming through or under them respectively.
- 5.3 The Noteholder shall be entitled to receive a Certificate stating the nominal amount of the Notes held by it and their subscription price but so that joint holders shall be entitled to one Certificate only in respect of the Notes held jointly by them which Certificate shall be delivered to the joint holder whose name stands first in the Register in respect of the joint holding or to such other person as the joint holders may, in writing, direct. Delivery of a certificate to one of such persons shall be sufficient delivery to all. The Company shall not be bound to register more than four persons as the joint holders of any Note.
- 5.4 If any Certificate is defaced or worn out, then, upon production of such Certificate to the directors of the Company they may cancel the Certificate and issue a Certificate in lieu thereof, and if any such Certificate is lost or destroyed, then, upon proof of loss or destruction being given to the satisfaction of the directors of the Company, and, in the case of a lost Certificate or in default of proof of destruction of a Certificate, on such terms as to evidence and indemnity as the directors of the Company may deem adequate, a new Certificate in lieu thereof may be given to the person entitled to the lost or destroyed Certificate. An entry as to the issue of the new Certificate and indemnity (if any) shall be made in the Register.



5.5 Notes shall only be redeemed against surrender of the relevant Certificate(s) for cancellation. If the Noteholder fails to deliver the Certificate for his Notes on redemption, the Company may retain the redemption moneys until delivery of the Certificate (or of an indemnity in respect of the Certificate in a form reasonably satisfactory to the Company).

#### 6. Acceleration Events

- 6.1 The occurrence of any Acceleration Event shall at the option of any of the Noteholders (exercised by written notice to the Company) cause the Notes to be, and they shall immediately become, due and repayable at their principal amount together with all accrued interest and all other amounts due and payable in respect of the Notes pursuant to this Instrument.
- 6.2 The Noteholders may, at any time after the Notes shall have become due and payable in accordance with Clause 6.1, and without notice:
  - (a) take such proceedings and/or other action against the Company to enforce their obligations under this Instrument; and
  - (b) exercise such powers and discretions as are given to them by this Instrument and/or the Agreement.

#### 7. Covenants

The Company covenants that, for so long as any of the Notes remain outstanding, the Company shall, save as otherwise directed in writing by the Noteholder:

- (a) at its own cost execute, acknowledge, deliver, file and register all such additional instruments, certificates, documents and assurances and perform such other acts or things as the Noteholder may reasonably request to give effect to the purposes of this Instrument or the transactions contemplated by this Instrument or to ensure the performance of the obligations imposed on it or to protect the rights of the Noteholder under this Instrument and the Notes; and
- (b) give notice in accordance with the terms of this Instrument to Noteholders of the occurrence or likely occurrence of any Acceleration Event immediately upon becoming aware of it.

#### 8. Miscellaneous

- 8.1 The Notes shall be subject to the provisions of Schedule 2.
- 8.2 The Company will pay all stamp duties, capital duties and other similar duties on or in connection with the execution, maintenance or enforcement of this Instrument.

#### 9. Notices

- 9.1 Notices or other communications given pursuant to this Instrument shall be in writing and shall be sufficiently given:
  - (a) if delivered by hand or sent by post to the address and for the attention of the person set forth in this Clause of the party to which the notice or communication is being given or to such other address and for the

attention of such other person as such party shall communicate to the party giving the notice or communication; or

- (b) if sent by facsimile to the correct facsimile number of the party to which it is being sent.
- 9.2 Any notice, or communication, given or sent by post under this Clause shall be sent by ordinary post.
- 9.3 Every notice or communication given in accordance with this Clause shall be deemed to have been received as follows:

Means of Dispatch	Deemed Received
Delivery by hand:	the day of delivery;
Post:	one Business Day after posting; and
Facsimile	when the sender receives a completed transmission sheet or otherwise receives a mechanical confirmation of transmission

Provided that if, in accordance with the above provisions, any such notice or other communication would otherwise be deemed to be given or made outside working hours (being 9 a.m. to 5 p.m. on a Business Day) such notice or other communication shall be deemed to be given or made at the start of working hours on the next Business Day.

9.4 The relevant addressee, address and facsimile number for the purposes of this Instrument are:

Name of Party Address/Fax no

The Company	AerCap House
	Shannon
	Co Clare, Ireland
	Facsimile no: +353 61 723 850
	Attention: The Company Secretary

The Noteholder As specified in the Register

#### 10. Alteration of this Instrument

The provisions of this Instrument and the conditions on which the Notes are held shall not, without the prior approval in writing of the Company and the written approval of the Noteholder, be altered, abrogated or added to.

#### 11. Governing Law

- 11.1 This Instrument and the Notes shall be governed by and construed in accordance with the laws of England.
- 11.2 The courts of England have non-exclusive jurisdiction to hear and decide any suit, action or proceedings, and to settle any disputes, which may arise out of or in connection with this Instrument ("**Proceedings**") and, for these purposes, each party irrevocably submits to the jurisdiction of the courts of England and waives

any objection which it might at any time have to the courts of England being nominated as the forum to hear and decide any Proceedings and agrees not to claim that the courts of England are not a convenient or appropriate forum.

IN WITNESS WHEREOF this Instrument has been executed the day and year first before written.

#### Schedule 1

#### Form of Notes Certificate

Certificate No [ • ]

Nominal amount of Notes US\$.

AerVenture Limited Registered Number 410443

Issue of Notes approved by a Resolution of the Board of Directors of the Company passed on [ ] 200[ ].

THIS IS TO CERTIFY that [•]

is the registered holder(s) of US\$[] nominal of the above mentioned Notes which are constituted by an Instrument entered into by the Company on and dated [] 200[] (the "**Instrument**") and are issued subject to the provisions (including the conditions) therein contained.

Interest at a rate of 15% per annum is payable on the Notes represented by this Certificate in accordance with the terms of the Instrument.

Without prejudice to the generality of the foregoing, the amount outstanding under the Notes shall be determined in accordance with the Instrument and payments of such principal outstanding shall be made subject to the terms of, and when contemplated by, the Instrument.

Given under the Common Seal of the Company on  ${\mbox{\ \bullet}}$  .

**PRESENT** when the common seal of **AERVENTURE LIMITED** was hereunto affixed in the presence of:

### NOTES:

Where the context so admits, words and expressions defined in the Instrument shall bear the same respective meanings in the conditions endorsed hereon.

The Notes are redeemable and transferable in accordance with the terms and conditions contained in the Instrument a copy of which is available from the Company. No transfer of the Notes will be accepted for registration unless accompanied by this certificate and lodged at the registered office of The Company.

A copy of the Instrument is available for inspection at the registered office of the Company at all reasonable times during office hours and will be sent as soon as possible on request to any holder of the Notes or any part thereof.

#### Schedule 2

#### **Register and Transfer**

#### 1. The Register

- 1.1 A register of the Notes will be kept by the Company at its registered office. There shall be entered in the Register:
  - (a) the names and addresses of the holder for the time being of the Notes;
  - (b) the amount of the Notes held by the registered holder;
  - (c) the date upon which the name of the registered holder is entered in the Register in respect of the Notes standing in his name; and
  - (d) the serial number of each Certificate issued.
- 1.2 Any change of the name or address of the holder of Notes shall be notified to the Company promptly and the Register shall be altered accordingly.
- 1.3 The holder of Notes shall be at liberty at all reasonable times during office hours to inspect the Register.

#### 2. **Register to be conclusive**

Except as required by law, the Company will recognise the registered holder of the Notes as the absolute owner thereof and shall not be bound to take notice or see to the execution of any trust (whether express implied or constructive) to which any Notes may be subject.

#### 3. Set-off

The holder of the Notes will be recognised by the Company as entitled to the Notes free from any equity, set-off or cross-claim on the part of the Company against the original or any intermediate holder of the Notes.

#### 4. Transfer of Notes

- 4.1 The provisions of clause 12 of the Agreement shall apply to transfers of Notes as if, and on the basis that:-
  - (a) such Notes constitute "**Shares**"; and
  - (b) the Noteholder proposing to transfer the Notes constitutes a "Shareholder",

for the purposes of the Agreement, and no transfer of any Notes shall be made or registered unless such transfer complies with the Agreement on this basis.

4.2 A Noteholder shall only be entitled to transfer Notes in multiples of US\$1,000 unless the entire holding of Notes of the Noteholder is less than US\$1,000, in which case the Noteholder may transfer that entire holding.

- 4.3 Every transfer of the Notes shall be made by an instrument of transfer in the usual or common form or in any other form which may be approved by the Directors. The instrument of transfer of the Notes shall be executed by or on behalf of the transferor but need not be executed by or on behalf of the transfere. The transferor shall be deemed to remain the holder of the Notes until the name of the transferee is entered in the Register in respect of the Notes being transferred.
- 4.4 The Directors may decline to recognise any instrument of transfer of the Notes otherwise permitted by this Instrument unless the instrument is deposited at the registered office of the Company accompanied by the Certificate for the Notes, and such other evidence as the Directors may reasonably require to show the right of the transfer to make the transfer. The Directors may waive production of any Certificate upon production to them of satisfactory evidence of the loss or destruction of the Certificate together with such indemnity as they may reasonably require.
- 4.5 No fee shall be charged for any registration of a transfer of the Notes or for the registration of any other documents which in the opinion of the Directors require registration.
- 4.6 The registration of a transfer shall be conclusive evidence of the approval by the Directors of such a transfer.
- 4.7 No Notes shall be issued or transferred to a person who is not already a party to the Agreement unless that person has executed a deed of adherence in the form or substantially in the form as set out in the Agreement.

#### 5. Payments

- 5.1 Any moneys payable on or in respect of any Notes shall be paid by bank draft or telegraphic transfer to such account as shall be nominated by the Noteholder.
- 5.2 If several persons are entered in the Register as joint holders of the Notes then, without prejudice to paragraph 5.1 above, the receipt or payment to the order of any one of them for any moneys payable on or in respect of the Notes shall be as effective a discharge to the Company as if the person to whose order the payment is made were the sole registered holder of the Notes in question.
- 5.3 All payments of moneys payable on or in respect of the Notes will be made gross, free and clear of all right of counterclaim or set off and without deductions or withholdings unless such deductions or withholdings are required by law, in which event the Company shall pay such additional amount as shall be required to ensure that the net amount received by the Noteholder will equal the full amount which would have been received by it had no such deduction or withholding been required to be made.

#### 6. No fee for registration

No fee will be charged for the registration of any transfer, power of attorney or other document relating to or affecting the title to the Notes.

# PRESENT when the Common Seal of **AERVENTURE LIMITED** was affixed hereto:

Director

Director/Secretary

# SIGNATURE PAGE

<b>SIGNED</b> by for and on behalf of	) )	
AERCAP AERVENTURE HOLDING B.V.	)	
SIGNED by	)	
for and on behalf of	)	
WAHA AV PARTICIPATIONS B.V.	)	Director
SIGNED, SEALED and DELIVERED	)	
by	)	
the duly authorised attorney	)	
of AERVENTURE LIMITED	)	
in the presence of:	ý	
· · · ·	1	
		88

Created by Morningstar Document Research <u>documentresearch.morningstar.com</u>